

Road taxation and spending in the EU



Road taxation and spending in the EU

This report is prepared by: Arno Schroten Maarten 't Hoen

Delft, CE Delft, June 2016

Publication code: 16.4G40.26

Taxes / Road transport taxes / Investments / Costs / Infrastructure / Comparing research FT: Revenue

Client: Fédération Internationale de l'Automobile.

CE publications are available from <a>www.cedelft.eu

Further information on this study can be obtained from the contact person, Arno Schroten.

© copyright, CE Delft, Delft

CE Delft

Committed to the Environment

Through its independent research and consultancy work CE Delft is helping build a sustainable world. In the fields of energy, transport and resources our expertise is leading-edge. With our wealth of know-how on technologies, policies and economic issues we support government agencies, NGOs and industries in pursuit of structural change. For 35 years now, the skills and enthusiasm of CE Delft's staff have been devoted to achieving this mission.



Content

| | Summary | 5 |
|------------|--|----------|
| | Glossary | 7 |
| | Country abbreviations | 10 |
| 1 | Introduction | 11 |
| 1.1 | Background | 11 |
| 1.2 | Objective and research questions | 11 |
| 1.3 | Methodological approach and uncertainties | 12 |
| 1.5 | Outline of the report | 16 |
| 2 | Revenue from taxes and charges | 17 |
| 2.1 | Introduction | 17 |
| 2.2 | Methodology to estimate the total tax/charge revenue | 17 |
| 2.3 2 1 | Taxation and charging of road transport in Europe | 19 22 |
| 2.5 | Average revenue from road transport taxes and charges | 27 |
| 3 | Infrastructure expenditures | 32 |
| 3.1 | Introduction | 32 |
| 3.2 | Defining road infrastructure expenditures | 32 |
| 3.3 3⊿ | Investments in road infrastructure Operational and maintenance expenditures | 54 ⊿1 |
| 3.5 | Total infrastructure expenditures | 48 |
| 4 | Infrastructure costs | 51 |
| 4.1 | Introduction | 51 |
| 4.2 | Methodology to estimate the infrastructure costs | 51 |
| 4.3 | Total infrastructure costs | 57 |
| 4.4 | Average minastructure costs | 00 |
| 5 | Comparison of revenue and costs | 64 |
| 5.1 5.2 | Introduction | 64 |
| 5.Z | Infrastructure cost coverage ratio road transport | 04 65 |
| 5.4 | Infrastructure cost coverage ratios per vehicle type | 66 |



| | 6 | Conclusions | 71 |
|-------|------|---|-----|
| | 6.1 | Tax/charge revenue from road transport is higher than infrastructure | 71 |
| | 62 | Significant contribution road transport to total tax revenue | 71 |
| | 6.3 | Costs of FU road infrastructure are estimated at € 178 billion in 2013 | 71 |
| | 6.4 | Road infrastructure expenditures have been severely affected by economic crisis | 72 |
| | 6.5 | Uncertainties | 72 |
| | | | |
| | 7 | References | 74 |
| Annex | A | Data sources and data reliability | 76 |
| | A.1 | Introduction | 76 |
| | A.2 | Approach to data reliability assessment | 76 |
| | A.3 | Taxes and charges | 77 |
| | A.4 | Investments | 80 |
| | A.5 | Operational and maintenance expenditures | 81 |
| | A.6 | Vehicle, passenger and tonne kilometres | 82 |
| | A.7 | Road network length | 84 |
| | A.8 | Other data | 84 |
| | A.9 | Recommendations to improve data availability and quality | 85 |
| Annex | В | Profiles case study countries | 86 |
| | B.1 | Introduction | 86 |
| | B.2 | Austria | 86 |
| | B.3 | Czech Republic | 88 |
| | B.4 | Denmark | 89 |
| | B.5 | Germany | 91 |
| | B.6 | France | 92 |
| | B.7 | Italy | 94 |
| | B.8 | The Netherlands | 96 |
| | B.9 | Poland | 97 |
| | B.10 | Spain | 99 |
| | B.11 | Sweden | 101 |
| | B.12 | UK | 103 |
| Annex | с | Tax revenue not adjusted for PPP | 105 |
| Annex | D | Average tax/charge revenue | 106 |
| Annex | E | Infrastructure expenditures not adjusted for PPP | 109 |
| Annex | F | Development of expenditures in the EU Member States | 112 |
| | F.1 | Introduction | 112 |
| | F.2 | Investment indices | 113 |
| | F.3 | O&M expenditure indices | 116 |
| | F.4 | Investments as share of GDP | 119 |
| | F.5 | O&M expenditures as share of GDP | 122 |





Summary

National and local governments spend a significant amount of their budgets in building and maintaining the road infrastructure network. The exact amount, however, is not common knowledge nor is the revenue that is being collected from road transport. With the European Commission intending to revise the Eurovignette Directive¹, CE Delft, on behalf of FIA Region I, has examined how much is being invested by the EU27 Member States² into road networks versus the revenue that is being collected from transport taxes and charges. The examination includes the revenue coming from cars, motorcycles, vans, busses and heavy goods vehicles (HGVs).

Costs versus revenue

Revenue from taxes and charges on road transport exceeds the infrastructure costs in the EU.

Figure 1

2013

The revenue from specific road transport taxes and charges (\in 286 billion) exceeds the infrastructure cost (\in 178 billion) in 2013 in the EU. In only a few countries (Croatia, Hungary and Romania) this was not the case, which is mainly because these countries have spent very large amounts in upgrading their road infrastructure network over the last two decades.

Infrastructure cost and revenue from specific transport taxes/charges per mode of transport in



As is shown in Figure 1, the tax/charge revenues from light vehicles (i.e. cars, motorcycles, vans) considerably exceed the cost they inflict to road infrastructure. For buses, on the other hand, tax/charge revenue is significantly lower than the infrastructure cost throughout the EU. Due to their higher mass, these vehicles inflict relatively more damage to roads than light vehicles, while at the same time these vehicles are less heavily taxed. At the EU level, revenue and cost are almost equal for HGVs, although large differences exist between countries. In some countries (e.g. Germany,



Directive providing a framework for HGV road charging schemes that must be followed by Member States who wish to apply those schemes. The revision of this Directive may include an

extension of the Directive's scope to other vehicles, such as passenger cars.
 ² Because of a lack of reliable data on road infrastructure spending, Cyprus is not covered in this assessment.

Sweden, UK) the tax/charge revenue significantly exceed infrastructure cost, while in countries (e.g. The Netherlands, Romania, Hungary) revenue are only 50% of the cost.

Revenues from taxes and charges

The revenue that was collected from specific transport taxes (registration tax (incl. VAT), ownership tax, fuel excise duties (incl. VAT), and road infrastructure charges) in 2013 in the EU27 came to a total of \leq 286 billion. Additionally, the revenue from VAT on vehicles and transport fuels is about \leq 79 billion. When the revenue was broken down by vehicle type, it was seen that cars contribute 71% of the total, due to the large volume of cars driving on the roads and the relatively high overall tax/charge levels for passenger cars (compared to other modes of transportation).

When looked at as a proportion of the national tax budget, income from specific road transport taxes/charges came to 5-10% of total tax revenue. Significant differences between countries do exist: in Slovenia, road transport contributed as much as 14% of the total national tax income, while in Sweden it was only 4%.

Infrastructure costs

For this study, road infrastructure cost include investments in new infrastructure and renewal of existing roads; expenditure on maintenance of existing roads; and operational expenditure enabling use of the road network (e.g. lighting). From the total infrastructure cost of \in 178 billion, about half (54%) can be allocated to passenger cars. The highest cost per kilometre are found for busses and HGVs, because these heavier vehicles inflict more damage to road infrastructure.

Development infrastructure spending over time

This study also examined long term trends in road infrastructure spending by governments from 1995-2013 and found that investments has been decreasing, especially since the beginning of the economic crisis. Particularly in Southern (e.g. Portugal, Spain, Italy) and Eastern (e.g. Poland, Hungary, Lithuania) European countries, budgets for investments in new infrastructure and maintenance of existing roads have been cut. There are, on the other hand, also a few countries where budgets have been increased to stimulate economic growth (e.g. France).

The crisis has, among other things, put a stop to the increase in infrastructure spending in early 2000 in Central and Eastern European countries. The trend towards increase in that period was a response to the rising need for improvement of road networks in order to facilitate economic development in those countries. Access to large-scale EU funds for road infrastructure investments has contributed to this trend as well.

Robustness of results

This study's results contain some uncertainties, which has to be kept in mind when interpreting them. Nevertheless, the data presented reflects at least the right order of magnitude for the costs and revenue that are considered. However, direct comparisons between countries should be made carefully, since data availability and quality vary significantly between countries. Trend data (e.g. development of infrastructure spending in the period 1995-2013) for individual countries and groups of countries (e.g. Western European countries) is relatively consistent over time, and the identification of trends for these countries is considered to be reliable.

EU road transport infrastructure costs are approximately € 178 billion in 2013

Road infrastructure expenditures seriously affected by economic crisis



Glossary

| Term | Explanation |
|-------------------------------------|---|
| €2013 | Euros expressed in price level 2013. |
| Bus | Passenger road motor vehicle designed to carry more than 24 persons (including the driver), and with provision to carry seated as well as standing passengers (ITF et al., 2009). In this study mini-busses (passenger road motor vehicle designed to carry 10-23 seated or standing persons) and coaches (see: Coach) are considered as busses as well. |
| CEEC | Central and Eastern European Countries, including: Bulgaria, Czech Republic, Estonia, Croatia, Hungary, Latvia, Lithuania, Poland, Romania, Slovakia, Slovenia; all Member States of the European Union, as distinguished from Western European Countries. |
| Coach | Passenger road motor vehicle designed to seat 24 or more persons (including the driver) and constructed exclusively for the carriage of seated passengers (ITF et al., 2009). |
| Charges | Compulsory requited payments, where requited means that the payer does receive anything directly in return. |
| Consumer Price Index (CPI) | Consumer Price Index; indicator measuring the weighted average of prices in a predetermined basket of goods. Changes in this indicator are used to correct monetarised data for inflation. |
| Enhancement costs/expenditures | All costs/expenditures of new infrastructure or expansion of existing infrastructure with respect to functionality and/or lifetime. |
| EU27 | All 28 EU Member States except Cyprus. |
| Excise duty | Excise duties are indirect taxes (see: Tax) on the sale or use of specific products. For example, fuel excise duties are taxes on the sale of (motor) fuels. |
| Expenditures (on infrastructure) | The actual amounts of money extracted annually from the public (or private) accounts to finance infrastructure. Infrastructure expenditures do not include financing costs. |
| External costs | Unintended costs imposed on third parties for which no compensation is received. Important types of external costs of transport are: air pollution, climate change, noise, accidents and congestion. |
| Fixed costs | Costs that do not vary with transport volume while the functionality of the infrastructure remains unchanged, or costs that enhance the functionality of the infrastructure. |
| Foreign vehicle | A vehicle registered in a country other than the reporting country and bearing registration plates of that foreign country. |
| Gross Domestic Product (GDP) | Aggregate measure of production equal to the sum of the gross value added of all resident, institutional units engaged in production. |
| Heavy Goods Vehicle (HGV) | Goods road vehicle with a gross vehicle weight above 3,500 kg, designed, exclusively or primarily, to carry goods. |
| Infrastructure costs | The direct expenses on infrastructure plus the financing costs or - regarded from a different point of view - the opportunity costs for not spending the resources for more profitable purposes. |
| Infrastructure cost | Ratio reflecting the share of infrastructure costs covered by |
| coverage ratio | tax/charge revenue. |
| Investment (expenditure) | Expenditures on the enhancement (see: Enhancement costs/expenditures) and the renewal (see: Renewal costs/expenditures) of the infrastructure network. |



| Term | Explanation |
|--------------------------|--|
| Load | Weight of the goods transported by a vehicle. |
| Maintenance | Costs/expenditures referring to the costs/expenditures of/for |
| costs/expenditures | 'ordinary' maintenance. These are relatively minor repairs with an |
| | economic lifetime of less than 1 to 2 years. |
| Motorcycle (MC) | Two-, three- or four-wheeled road motor vehicle not exceeding |
| | 400 kg (900 lb) of unladen weight. All such vehicles with a cylinder |
| | capacity of 50 cc or over are included. |
| National vehicle | A vehicle registered in the reporting country and bearing |
| | registration plates of that country or having been separately |
| | registered. |
| Operation costs/ | These costs/expenditures refer to the costs/expenditures of the |
| expenditures | organisation of efficient use of the infrastructure. |
| Operation and | The sum of operation (see: Operation costs/expenditures) and |
| Maintenance (O&M) | maintenance (see: Maintenance expenditures/costs). |
| costs/expenditures | |
| Ownership tax | Periodical tax levied on the ownership of a vehicle. Often referred |
| | to as circulation tax. |
| Passenger car | Road motor vehicle, other than a moped or a motorcycle, intended |
| | for the carriage of passengers and designed to seat no more than |
| | nine persons (including the driver). |
| Passenger car | Indicator measuring the impact that a single vehicle has on traffic |
| equivalent (PCE) | variables (e.g. speed, density) compared to a single car. |
| Passenger kilometre | Unit of measurement representing the transport of one passenger |
| (pkm) | by road over one kilometre. |
| Paved road | Road surfaced with crushed stone (macadam) with hydrocarbon |
| | binder or bituminized agents, with concrete or with cobblestone. |
| Perpetual Inventory | Method to estimate infrastructure costs based on time series data |
| Method (PIM) | on infrastructure expenditures. To estimate the enhancement (see: |
| | Enhancement costs/expenditures) and renewal (see: Renewal |
| | costs/expenditures) costs, this method first calculates the annual |
| | depreciation costs by distributing the initial investment over the |
| | lifetime of the infrastructure. In addition to the depreciation costs, |
| | interest costs are estimated by using an appropriate interest rate. |
| | The sum of depreciation and interest costs equals enhancement |
| | and/or renewal costs. O&M costs (see: O&M costs/expenditures) |
| | are not capitalised in the PIM, but running costs are taken into |
| | account instead. |
| Purchasing Power Parity | Indicator reflecting the purchasing power of countries. This |
| (PPP) | indicator is used to correct monetarised figures for differences in |
| | purchasing power of a euro across countries. |
| Registration tax | Taxes levied on the (first) registration of a vehicle in a country. |
| | Registration fees are not included, as these are direct payments for |
| | actual activities carried out (i.e. registering vehicles). |
| Renewal | All costs/expenditures associated with the renewal of (parts of) the |
| costs/expenditures | infrastructure. The renewed (parts of) the infrastructure will at |
| | least have a lifetime of more than 1-2 years. Renewal |
| | costs/expenditures do include extraordinary maintenance with a |
| | lifespan of more than 1-2 years. |
| Road | Line of communication (travelled way) open to public traffic, |
| | primarily for the use of road motor vehicles, using a stabilised base |
| | other than rails or air strips. |
| Road network | All roads in a given area. |
| Road network length | The length of all roads in a given area. |
| Road traffic on national | Any movement of road vehicles within a national territory |
| territory | irrespective of the country in which these vehicles are registered. |



| Term | Explanation |
|-----------------------------------|--|
| Southern European Countries | Group of countries, including: Greece, Italy, Portugal, and Spain. |
| Steady state level expenditure | Expenditures related to the minimum package of maintenance (and operational) measures required to ensure the long-term physical and functional integrity of existing infrastructure under current conditions. |
| Subsidy | Fiscal supports with direct relevance to public budgets and with no direct service in return. |
| Tax | Compulsory unrequited payments, where unrequited means that the payer does not receive anything directly in return. |
| Tonne kilometre (vkm) | Unit of measurement of goods transport which represents the transport of one tonne over one kilometre. |
| Unpaved road | Road with a stabilised base not surfaced with crushed stone, hydrocarbon binder or bituminised agents, concrete or cobblestone. |
| Value added tax (VAT) | Indirect tax (see: Tax) on the domestic consumption of goods and services. VAT is imposed on the added value at each stage of production. Producers are VAT-registered and they are entitled to deduct from the VAT amount the VAT paid on his or her purchases. For the final consumer, not being VAT-registered, VAT is a tax on the consumption of a good or service. |
| Van | Four-wheeled Goods road motor vehicle with a gross vehicle weight of not more than 3,500 kg. |
| Variable costs | Costs that vary with transport volumes while the functionality of the infrastructure remains unchanged. |
| Vehicle kilometre (vkm) | Unit of measurement representing the movement of a vehicle over one kilometre. |
| WEC | Western European Countries, including: Austria, Belgium, Denmark, Germany, Finland, France, Greece, Ireland, Italy, Luxembourg, Malta, The Netherlands, Portugal, Spain, Sweden, UK; all Member States of the European Union, as distinguished from Central and Eastern European Countries. Since for Cyprus only part of the data was available, it has not been included in this group of countries in this study. |



Country abbreviations

| Abbreviation | Country | | | |
|--------------|-----------------|--|--|--|
| AT | Austria | | | |
| BE | Belgium | | | |
| BG | Bulgaria | | | |
| CZ | Czech Republic | | | |
| DK | Denmark | | | |
| DE | Germany | | | |
| EE | Estonia | | | |
| FI | Finland | | | |
| FR | France | | | |
| GR | Greece | | | |
| HR | Croatia | | | |
| HU | Hungary | | | |
| IE | Ireland | | | |
| IT | Italy | | | |
| LV | Latvia | | | |
| LT | Lithuania | | | |
| LU | Luxembourg | | | |
| MT | Malta | | | |
| NL | The Netherlands | | | |
| PO | Poland | | | |
| РТ | Portugal | | | |
| RO | Romania | | | |
| SK | Slovakia | | | |
| SI | Slovenia | | | |
| ES | Spain | | | |
| SE | Sweden | | | |
| UK | United Kingdom | | | |



1 Introduction

1.1 Background

Transport is part of any society and has a clear impact on citizens' lifestyle, location of activities and possibilities for consumption. Developments in transport, such as the increase in average speed with which items and services can be delivered, have led to fundamental changes in the way in which societies are organised. These developments are often regarded as an important driver of economic growth.

An efficient transport system requires a functional and well-maintained infrastructure. Significant parts of public budgets are dedicated to infrastructure investment, both to build new roads and to maintain and operate existing ones. On the other hand, governments levy several taxes and charges on the possession and use of vehicles, resulting in considerable amounts of revenue. As EU-wide information on these costs and revenue is limited, the current balance between infrastructure costs and revenue from transport taxes and charges is unknown.

This study aims to fill this knowledge gap, by studying the infrastructure costs and tax/charge revenue from road transport in the EU. Comparing costs and revenue will allow us to assess whether the road sector is a net contributor to the national budgets.

1.2 Objective and research questions

The objective of this study is:

- to provide an overview of the revenue of road taxes and charges, and the road infrastructure expenditures and costs in the EU in 2013;
- to make (graphical) comparisons of the tax/charges revenue and infrastructure costs of road transport for the EU countries, differentiating between different vehicle types.

In order to achieve this objective, the following research questions are addressed in this study:

- 1. What is the amount of road tax/charge revenue in 2013 in the EU Member States?
 - a What road taxes/charges are applied in the various EU Member States?
 - b What is the total revenue from these taxes/charges in 2013 for every EU Member State?
 - c What share do the various vehicle categories have in the total tax/charge revenue?



- 2. What is the amount of the expenditure on road infrastructure in 2013 in the EU Member States?
 - a Which expenditures on road infrastructure should be considered?
 - b What has been the size of these different categories of road expenditures in 2013 and over time (1995-2013)?
 - c Are there significant differences between EU Member States with respect to these expenditures and how can these differences be explained?
- 3. What are the road infrastructure costs in 2013 in the EU Member States?
 - a How are infrastructure costs estimated based on expenditure data?b What is the total amount of road infrastructure costs in 2013 in the EU? How can main differences between Member States be explained?
 - c Which share do the various vehicle categories have in the total infrastructure costs?
- 4. How are the total tax revenue compared to the total infrastructure costs in 2013 in the EU Member States?
 - a From what perspectives can tax revenue and infrastructure costs be compared?
 - b What are the infrastructure cost coverage ratios in 2013 in the various EU countries?

1.3 Scope of the study

In this study, the following basic principles are applied:

- In this report, we present the total infrastructure costs and total revenue from taxes and charges³ (see Sections 4.2 and 2.2 for more information on the scope applied for these two concepts respectively). Additionally, we present average figures, expressed in € per passenger kilometre for passenger modes, € per tonne kilometre for Heavy Goods Vehicles (HGVs), and € per vehicle kilometre for vans.
- Geographical scope: the study covers the EU28⁴. The road infrastructure expenditure data for Cyprus, however, did not allow to include it in the overviews. Therefore, for Cyprus only data on tax/charge revenue are presented in Chapter 2.
- All data is presented for the year 2013 for tax/charge revenue and infrastructure costs, as this is the most recent year for which all required data is available. Infrastructure expenditure data covers the period 1995-2012.
- All financial data is shown in euro price levels of 2013. Data from sources where price levels from other years were used, were translated to price level 2013 by Consumer Price Indices (CPI) for the specific countries. All financial figures were adjusted for differences in price level to allow comparison between countries. The consequences of these corrections are discussed in the textbox below.
- This study distinguishes between the following vehicle categories:
 - passenger car;
 - motorcycle (MC);

⁴ Although Croatia joined the EU in the middle of 2013 (1 July 2013), we have included it in this study.





³ Taxes are compulsory unrequited payments, where unrequited means that the payer does not receive anything directly in return. Charges, on the other hand, are requited payments in that they include the delivery of a service in exchange for a payment. For example, infrastructure charges are payments made for vehicles to use specific parts of the road infrastructure.

- bus (also includes coaches and mini-busses)⁵;
- van;
- heavy goods vehicles (HGV).
- We present both infrastructure expenditure and cost data. Expenditure data shows the impact of transport infrastructure on public accounts.
 Costs are preferred for the comparison between tax/charge revenue and infrastructure costs (see also Section 3.2).
- Transport subsidies are not covered by this study, with the exception of EU subsidies/loans for infrastructure investments and maintenance (e.g. Cohesion funding). These subsidies are (implicitly) part of the infrastructure expenditures (and hence costs) discussed in Chapter 3 (and 4).
- External costs of road transport are not covered in this study.
- All data is gathered from national and international data sources, including data from statistical offices, public accounts and other data sources on a national level. No primary data gathering on the level of lower governments such as provinces or communities has taken place.

Interpretation of results adjusted for price level variations between countries

European countries differ with respect to price levels; e.g. prices in Romania are significantly lower than in Germany. This implies that the purchasing power of one euro is larger in Romania than it is in Germany. In other words, you can buy more for one euro in Romania than you can in Germany. These differences in purchasing power also affect the comparison of infrastructure spending (and tax/charge revenue) between European countries. An investment of € 1 million in roads in Romania is much larger (in terms of domestic purchasing power) than the same investment in Germany. To make a fair comparison between countries, financial figures have to be corrected for the differences in purchasing power between countries. This can be done by applying Purchasing Power Parities (PPPs), which are indicators that reflect the differences in purchasing power between countries (by benchmarking them to the EU28 average price level). Applying such corrections would imply that the size of financial figures for countries with relatively low price levels (like Romania) will increase, while the size of financial figures for countries with relatively high price levels (like Germany) will decrease.

In this study all financial figures have been corrected with PPP-indicators from Eurostat. This implies that all PPP corrected financial figures (i.e. Infrastructure costs/expenditures, tax/charge revenue) for individual countries are shown for the EU28 average price level. Summing up these figures for all European countries provide total figures for the EU, expressed in the EU28 average price level.

PPP-adjusted figures cannot be compared directly with figures in national accounts, national statistics or national studies. This should be kept in mind when comparing the results of this study to results/figures from other sources.

1.4 Methodological approach and uncertainties

This study consists of three phases:

 Data gathering; in this phase we collected data on revenue from road transport taxes and charges and road infrastructure expenditure. More general data (vehicle kilometres (vkm), passenger kilometres (pkm), tonne kilometres (tkm), road network length, Gross Domestic Product



⁵ We were not able to distinguish between busses and coaches in our analyses, as this distinction is not made in many transport statistics (e.g. vehicle kilometer data is often only available for busses and coaches together).

(GDP), exchange rates, inflation rates, etc.) was collected as well. Several methodologies have been applied to gather all these data:

- Case studies: for eleven countries (Austria, Czech Republic, Denmark, Germany, France, Italy, The Netherlands, Poland, Spain, Sweden and the UK) an in-depth search for infrastructure expenditure data has been carried out by national experts⁶. These case studies provided detailed data that was used to further detail the infrastructure expenditure data for other countries. For more information on the case studies, see Annex B.
- Assessment of EU-wide statistical databases (mainly Eurostat, ITF and OECD) and reports (e.g. EU pocketbook).
- Assessment of national statistical databases (mainly from national statistical agencies) and reports.
- National representatives of Finance Ministries, Transport Ministries, road authorities, road charging scheme operators, and statistical agencies were requested to provide specific data (mainly to fill in data gaps).
- 2. Data processing; in this phase we first checked the reliability of the data (e.g. by crosschecking them with data from other sources) and missing data was estimated (see Annex A for more details). Next, infrastructure costs were estimated based on the infrastructure expenditure data gathered (see Section 4.2 for a detailed description of the methodology used to estimate the infrastructure costs). Finally, infrastructure cost coverage ratios were estimated by combining the data on tax/charge revenue and the estimated infrastructure costs.
- 3. Data analysis and interpretation; The data on tax/charge revenue, infrastructure expenditures/costs and infrastructure cost coverage ratios was analysed by means of graphical overviews and comparisons. The results were interpreted to form conclusions.

Data reliability

The results provided by this study are based on a lot of data, from different sources and of varying quality. Therefore, it is important to carefully consider the reliability of the data used in this report and the impact any uncertainties have on the results presented. In Annex A the reliability of the data used in this report is discussed in detail. A brief overview of the main results of that assessment are given in this section. Some recommendations on further research to improve the reliability of the data are given in Annex A as well.

Reliable data is available for most countries on total revenue from road transport taxes and charges (see Figure 2A), based on reliable sources or estimates (for more details, see Annex A). The exact allocation per mode of transport is only known in a few countries (i.e. Austria, Belgium, Denmark, Germany, Latvia, The Netherlands). Additional analysis was therefore needed to estimate the total revenue per mode of transport (methodology applied is described in Section 2.2), which results in a modest level of uncertainty (see Figure 2B).



⁶ The following national experts have been deployed for this task: Herry (Austria), Herbert Seelmann (Czech Republic), Nilsson Production (Denmark, Sweden), Setec (France), TRT (Italy), Agnieszka Markowska (Poland), University of Madrid (Spain), CE Delft (Germany, The Netherlands, UK).

Figure 2 Assessment of reliability of data on tax/charge revenue



The data on infrastructure expenditures on investments and Operation and Maintenance (O&M expenditures) is less reliable as is shown in Figure 3 (see Section 3.2 for definitions of investments and O&M expenditures). There may be serious concerns regarding the reliability of the expenditure data used for some Central, Eastern and Southern countries. Due to a lack of data for some countries such as Bulgaria, Estonia and Greece, data for those countries has been estimated, which results in significant levels of uncertainty (see Annex A for more details). Furthermore, there are differences in data quality (e.g. level of detail), which affects data reliability. Finally, there may be differences in accounting principles used in the various countries (due to a lack of common definitions and practices to measure infrastructure expenditures), which may hamper the data comparability.



Figure 3 Assessment of reliability of infrastructure expenditure data



As many of the results are presented in terms of \notin /vkm, \notin /pkm, \notin /tkm or \notin /km road network, the reliability of these variables is interesting to consider as well. The geographical scope of most vkm/pkm/tkm statistics are related to the country in which the vehicle is registered. However, for this study we are interested in the traffic performances on national territory (by both national and foreign vehicles), as this matches with the scope of infrastructure expenditures and tax/charge revenue. The availability of this type of data is rather poor and we therefore have to combine several data sources to complete a consistent set of vkm/pkm/tkm data. These data sources vary in terms of reliability, as is shown in Figure 4A. These uncertainties are reflected in the average tax/charge revenue and average infrastructure cost estimates, as they are expressed in \notin /vkm, \notin /pkm or \notin /tkm. For road network length, on the other hand, relatively reliable data is available for most of the European countries (see Figure 4B). Therefore, results expressed in \notin /vkm, \notin /pkm or \notin /tkm.

Figure 4 Assessment of reliability of transport performance indicators and road network length



1.5 Outline of the report

In this report, we present data on the revenue from taxes and charges of road transport in Chapter 2. In Chapter 3, we discuss the expenditures on road infrastructure in the period 1995-2013 in the EU. The estimated infrastructure costs are presented in Chapter 4, and are then compared with the revenue from taxes and charges in Chapter 5. Finally, the main conclusions are presented in Chapter 6.



2 Revenue from taxes and charges

2.1 Introduction

Several types of taxes and charges related to the ownership and use of road vehicles and road infrastructures are levied in the EU. In this chapter, we provide an overview of these taxes and charges and their total revenue in 2013. This will provide an answer to the first research question of this study (see the textbox below).

Research question 1

What is the amount of road tax/charge revenue in 2013 in the EU Member States?

This question consists of three sub questions:

- 1. What road taxes/charges are applied in the various EU Member States?
- 2. What is the total revenue from these taxes/charges in 2013 for every EU Member State?
- 3. What share do the various vehicle categories have in the total tax/charge revenue?

In the remainder of this chapter, we first discuss the methodology used to estimate the total tax revenue (per vehicle category) in Section 2.2. Next, we give an overview of the different types of taxes and charges that are levied in the various EU Member States (Section 2.3). The total revenue from road taxes and charges is considered in Section 2.4, while the average revenue (in terms of \notin /vkm, \notin /pkm and \notin /tkm) is discussed in Section 2.5.

2.2 Methodology to estimate the total tax/charge revenue

Scope

This study covers the following taxes and charges:

- registration taxes/purchase taxes⁷;
- ownership taxes (circulation taxes);
- road tolls and vignettes;
- fuel excise duty;
- VAT on registration taxes;
- VAT on vehicle purchases;
- VAT on fuel excise duties;
- VAT on fuel purchases.

Company car taxation and insurance taxation are not covered in this study, as they are not specific transport taxes (e.g. company car taxation is a type of income taxation as it taxes the benefit in kind that is attributed to company cars). It is also questionable to what extent VAT on vehicle and fuel purchased can be considered transport taxes. Eurostat (2001) excludes VAT from the concept of environmental/transport taxes, because it was considered to have



¹ Registration fees are not considered in this study, as these are direct payments for actual activities carried out (i.e. registering vehicles). As the costs of these activities are not considered in these studies, the payments for them should not be included in the analysis as well.

no influence on relative prices (in the transport market) in the same way that other transport taxes do. However, as stated by Steinbach et al. (2009), there is one exception where VAT should be included. In cases where VAT is charged on a tax/duty that is considered a transport tax, VAT should be included in the concept of transport taxes as well. This implies that the VAT charged on fuel excise duties and on registration taxes is considered transport taxes, while VAT on the production costs of fuel and vehicles is not. For this reason, we consider these different types of VAT separately in this chapter.

Finally, the revenue from parking charges is not considered in this study, as the data availability on this issue is very fragmented and poor. We have consulted national statistical agencies as well as the European Parking Association (EPA), but only information on parking revenue in the Netherlands and the UK were found. These results are presented in Section 2.4, in order to provide insight in the potential contribution this revenue has in total road transport tax and charge revenue.

Methodology

A top-down approach was used to estimate the revenue from registration tax, circulation tax and road toll per vehicle type. Total revenue figures were collected from statistical databases or public accounts (see Annex A for the data sources used) and subsequently allocated to the various vehicle types based on relevant allocation parameters (see Table 1). A similar approach was used for the VAT revenue from registration taxes and fuel excise duties. On the other hand, a bottom-up approach was applied to fuel excise duty and VAT revenue. The total revenue per vehicle type was estimated through a multiplication of the total amount of fuel consumed (by new registers vehicles) by the average tax rate. This was carried out separately for each vehicle type.

| Taxes/charges | Estimation approach |
|---------------------|--|
| Registration taxes | Top-down approach: the total revenue was allocated to the various vehicles by use of 2013 sales volumes (from ACEA, EEA or Eurostat) weighted by the average tax rates per vehicle category. |
| Ownership taxes | Top-down approach: the total revenue was allocated to the various vehicles by use of the size of the various fleets in 2013 (based on Eurostat data) weighted by the average tax rates per vehicle category. |
| Tolls and vignettes | Both for road tolls and vignettes, a top-down approach was applied. For road tolls, the total revenue (mainly from ASECAP) was allocated based on vehicle kilometres on tolled roads, weighted by average toll rates per vehicle category. For vignettes, the total revenue is allocated based on the size on the various fleets in 2013, weighted by the average rates. |
| Fuel excise duty | Bottom-up approach: based on total fuel volumes sold in the various EU Member States and excise duty rates, the total revenue is estimated. The allocation to the various vehicle types was based on the shares the various vehicle categories have in total petrol/diesel/LPG consumption. Fuel excise refund schemes for HGVs in some EU Member States (e.g. France) are taken into account. |

Table 1 Estimation approach total tax/ charge revenue



| Taxes/charges | Estimation approach |
|---------------------------|--|
| VAT on registration taxes | Top-down approach: based on the total revenue from registration taxes per vehicle type and VAT rates in the various Member States, the revenue from VAT on registration taxes per vehicle type was estimated. |
| VAT on vehicles | Bottom-up approach: based on average sales prices before taxes (from ICCT, ACEA)) and VAT rates in the various Member States, the VAT on new vehicles was estimated. This was only done for passenger cars and motorcycles. For busses, vans and HGVs it was assumed that VAT on vehicle purchases can be recovered. VAT on second hand cars was not considered due to a lack of data. |
| VAT on fuel excise duties | Top-down approach: based on the total revenue from fuel excise duties per vehicle type and VAT rates in the various Member States, the revenue from VAT on fuel excise duties per vehicle type was estimated. |
| VAT on fuel | Bottom-up approach: first the VAT per litre was estimated, based on average commodity prices and VAT rates for the various Member States. Next, total revenue was estimated by use of data on total fuel sales in the various Member States (from Eurostat). For busses, vans and HGVs it was assumed that VAT on fuel purchases can be recovered. |

2.3 Taxation and charging of road transport in Europe

There are large differences in the way road transport is taxed/charged in EU countries. In this section, we present a brief overview of the situation in 2013.⁸ The ACEA Tax Guide gives a more detailed overview per country (ACEA, 2013).

2.3.1 Vehicle taxes

Taxation of the registration and/or ownership of road vehicles is applied in all EU Member States, but the scope of these taxes differ widely between countries (see Figure 5). Registration tax (or charge or excise duty) is applied on vehicles (re)entering the fleet in twenty countries, but only three countries (Italy, Romania and Slovakia) apply it to all vehicle types. While owners of passenger cars have to pay registration tax in twenty countries, owners of HGVs only have to pay it in five countries (Italy, Romania, Slovakia, France and Greece).



⁸ Notice that in the years after 2013 there have been changes in the taxes and charges that are applied for road transport in some of the EU countries, As 2013 is the base year for our analysis, these changes have not been taken into account in this study.

Figure 5 Vehicle types on which vehicle taxes are levied in 2013



All Member States levy a periodic ownership or circulation tax on road transport vehicles (see Figure 5B). The majority of the countries apply this tax to all vehicle types. Contrary to the registration taxes, HGVs are subject to this form of taxation in nearly all countries; they are only exempted in Croatia. For passenger cars an ownership tax is applied in most countries as well; only three countries (Estonia, Lithuania and Poland) exempt passenger cars from ownership tax (ACEA, 2013).

Member States also apply significantly different tax base/structures to registration and circulation taxes (ACEA, 2013). Vehicle value, CO_2 emissions (expressed in g/km) and engine size/power are often used as parameters to define the registration tax rate. Ownership taxes are often based on engine size/power and CO_2 emissions for passenger car or on weight very often combined with axle configuration and suspension for HGV.

2.3.2 Tolls and vignettes

Infrastructure charges are applied in most of the EU Member States. Two main types of infrastructure charges can be distinguished: distance-based systems (tolls) and time-based systems (vignettes). As is shown in Figure 6, nine EU countries apply distance based road charging schemes covering all vehicle types, mainly on (part of) the national road network (ACEA, 2013; CE Delft et al., 2012). Additionally, four countries apply a road charging scheme for heavy duty vehicles: Germany only applies it to HGVs, while the Czech Republic, Slovakia and Slovenia include busses as well. Some countries without nationwide road tolling apply local schemes for the use of specific infrastructure. Examples include the Oresund Bridge between Denmark and Sweden, the M6 in the UK and various tunnels in Belgium and the Netherlands. Where possible, revenue from these local schemes was included as well (see Annex A for more details).







Time-based vignettes are applied in thirteen countries for at least one type of road vehicle. Belgium, Denmark, Luxembourg, the Netherlands and Sweden have a common vignette scheme for HGVs. Vignette schemes are applied in seven Central and Eastern European countries and Austria as well, always covering passenger cars and other vehicle types on a country-by-country basis.

2.3.3 Fuel excise duties

All EU Member States levy excise duties on transport fuels. An overview of the tax levels for the main transport fuels (petrol and diesel) is given in Figure 7. The highest fuel taxes are levied in the Netherlands (for gasoline), Italy and the UK (both gasoline and diesel). The excise duty on gasoline is higher than on diesel in all countries, with the exception of the UK where equal taxes on gasoline and diesel are applied. In some EU countries (i.e. Belgium, France, Hungary, Ireland, Italy, Slovenia and Spain), a refund scheme for part of the fuel excise duty exists for HGVs. These schemes are taken into account in estimating the total tax and charge revenue in the next section.



Figure 7 Fuel excise duty levels in the EU in 2013



Source: European Commission (2013).

2.3.4 VAT

Value added taxes⁹ (VAT) are levied on the purchase of transport vehicles (including registration taxes) and transport fuels (including fuel excise duties) in all EU countries. The VAT rates applied in 2013 are presented in Figure 8. The highest VAT rates are found for Hungary, followed by Denmark, Croatia and Sweden. The lowest VAT rate is applied in Luxembourg.

VAT has to be paid by the final consumer, implying that companies can reclaim their VAT payments on intermediate products. As transport is an intermediate product for companies, they can reclaim all VAT payments on vehicle purchases and transport fuels. For that reason, this study assumes that there are no VAT revenue from busses, vans and HGVs (as these vehicles are mainly used by companies).



⁹ Indirect taxes on the consumption of goods or services. VAT is imposed on the added value at each stage of production. Producers are VAT-registered and they are entitled to deduct from the VAT amount the VAT paid on his or her purchases. For the final consumer, not being VAT-registered, VAT is a tax on the consumption of a good or service.

Figure 8 VAT rates applied for vehicle purchases and transport fuels in the EU in 2013



Source: ACEA (2013).

2.4 Total revenue from road transport taxes and charges

An overview of the total tax/charge revenue in 2013 in the EU27 is given in Table 2, while the tax/charge revenue in Cyprus are presented in the following textbox. The data sources used and details on the figures' reliability are presented in Annex A. All figures are PPP adjusted to allow comparison between countries (see also Section 1.3). The unadjusted figures can be found in Annex C.

| Member State | Registration tax | Ownership tax | Tolls and vignettes | Fuel excise duty | VAT on registration taxes | VAT on fuel excise duty | VAT on vehicle purchase | VAT on fuel | Total (incl. VAT on taxes) | Total (incl. all VAT) |
|-------------------|---------------------|------------------|---------------------|---------------------|---------------------------------|----------------------------|-------------------------------|-------------|----------------------------------|--------------------------|
| Austria | 0.4 | 1.6 | 1.5 | 3.4 | 0.1 | 0.6 | 1.3 | 0.8 | 7.6 | 9.7 |
| Belgium | 0.4 | 1.5 | 0.1 | 3.8 | 0.1 | 0.6 | 1.1 | 0.8 | 6.5 | 8.4 |
| Bulgaria | - | 0.2 | 0.2 | 1.9 | - | 0.1 | 0.2 | 0.7 | 2.4 | 3.3 |
| Czech Republic | - | 0.3 | 0.7 | 4.3 | - | 0.4 | 0.9 | 1.0 | 5.6 | 7.5 |
| Denmark | 1.9 | 1.4 | 0.4 | 1.6 | 0.4 | 0.4 | 0.4 | 0.3 | 6.1 | 6.8 |
| Germany | - | 8.1 | 4.2 | 32.4 | - | 5.1 | 12.5 | 5.9 | 49.7 | 68.0 |
| Estonia | - | 0.01 | - | 0.5 | - | 0.0 | 0.1 | 0.1 | 0.5 | 0.8 |
| Finland | 0.8 | 0.7 | - | 2.0 | 0.2 | 0.4 | 0.2 | 0.2 | 4.1 | 4.5 |
| France | 1.8 | 0.2 | 9.8 | 20.1 | 0.4 | 3.0 | 6.2 | 3.5 | 35.3 | 45.0 |
| Greece | 0.1 | 1.4 | 0.6 | 4.0 | 0.02 | 0.5 | 0.2 | 0.8 | 6.7 | 7.7 |
| Croatia | 0.03 | 0.05 | 0.5 | 1.2 | 0.01 | 0.1 | 0.2 | 0.5 | 1.9 | 2.7 |
| Hungary | 0.1 | 0.2 | 0.3 | 2.7 | 0.01 | 0.3 | 0.5 | 1.0 | 3.7 | 5.2 |
| Ireland | 0.4 | 1.0 | 0.2 | 2.1 | 0.1 | 0.3 | 0.2 | 0.3 | 4.1 | 4.7 |

Table 2 Total revenue from road taxes/charges in the EU in 2013 (billion €2013, PPP adjusted)



| Member State | Registration tax | Ownership tax | Tolls and vignettes | Fuel excise duty | VAT on registration taxes | VAT on fuel excise duty | VAT on vehicle purchase | VAT on fuel | Total (incl. VAT on taxes) | Total (incl. all VAT) |
|-----------------|---------------------|------------------|---------------------|---------------------|---------------------------------|----------------------------|-------------------------------|-------------|----------------------------------|--------------------------|
| Italy | 1.3 | 5.7 | 6.6 | 23.3 | 0.3 | 4.0 | 5.5 | 4.5 | 41.1 | 51.2 |
| Latvia | 0.01 | 0.1 | - | 0.5 | 0.00 | 0.04 | 0.1 | 0.1 | 0.7 | 0.9 |
| Lithuania | - | 0.1 | 0.05 | 1.0 | - | 0.1 | 0.1 | 0.3 | 1.2 | 1.6 |
| Luxembourg | - | 0.1 | 0.01 | 0.8 | - | 0.1 | 0.2 | 0.1 | 0.9 | 1.2 |
| Malta | 0.04 | 0.1 | - | 0.1 | 0.01 | 0.0 | 0.01 | 0.03 | 0.2 | 0.2 |
| Netherlands | 1.06 | 4.6 | 0.1 | 6.9 | 0.3 | 0.8 | 0.4 | 0.6 | 13.7 | 14.8 |
| Poland | 0.59 | 0.4 | 0.9 | 11.2 | 0.1 | 0.9 | 2.2 | 3.6 | 14.0 | 19.9 |
| Portugal | 0.46 | 0.6 | 1.0 | 3.3 | 0.1 | 0.4 | 0.7 | 1.0 | 6.0 | 7.6 |
| Romania | 0.33 | 0.4 | 0.5 | 4.0 | 0.03 | 0.2 | 0.5 | 1.0 | 5.5 | 6.9 |
| Slovakia | 0.01 | 0.2 | 0.3 | 1.6 | 0.00 | 0.1 | 0.4 | 0.3 | 2.3 | 2.9 |
| Slovenia | 0.04 | 0.2 | 0.4 | 1.2 | 0.01 | 0.1 | 0.2 | 0.3 | 2.0 | 2.5 |
| Spain | 0.38 | 3.1 | 1.7 | 13.0 | 0.1 | 2.0 | 2.6 | 4.5 | 20.3 | 27.4 |
| Sweden | - | 1.0 | 0.2 | 3.9 | - | 0.9 | 1.2 | 0.5 | 6.0 | 7.7 |
| UK | - | 4.7 ^a | 0.3 | 28.8 | - | 4.3 | 4.5 | 3.4 | 38.1 | 46.0 |
| EU27 | 10.2 | 37.8 | 30.6 | 179.7 | 2.1 | 25.9 | 42.6 | 36.0 | 286.4 | 365.0 |

^a No data for Northern Ireland is available. Therefore, data for Great Britain is used as a proxy for the UK.

Revenue from taxes and charges in Cyprus

The total 2013 revenue from road transport taxes and charges in Cyprus amounts to \notin 0.7 billion (PPP adjusted). About 56% of the revenue is coming from fuel excise duties, while total VAT revenue and vehicle taxes contribute 24% and 20% to total revenue respectively. No infrastructure charges (tolls and/or vignettes) are applied in Cyprus.

Table 3 Total revenue from road taxes/charges in Cyprus (billion €2013, PPP adjusted)

| Registration tax | Ownership tax | Tolls and vignettes | Fuel excise duty | VAT on registration tax | VAT on fuel excise duty | VAT on vehicle purchase | VAT on fuel | Total (excl. VAT on vehicle purchase and fuel) | Total (incl. VAT) |
|---------------------|------------------|---------------------|---------------------|-------------------------------|----------------------------|-------------------------------|-------------|--|----------------------|
| 0.03 | 0.1 | 0 | 0.4 | 0.01 | 0.04 | 0.03 | 0.1 | 0.6 | 0.7 |

Table 4 shows that of the passenger modes of transportation, passenger cars have the highest average revenue, followed by motorcycles and busses (\notin 41, \notin 32 and \notin 8 per 1,000 passenger kilometres, respectively). The average revenue from HGVs is \notin 86 per 1,000 tonne kilometres. Finally, vans have average revenue of \notin 48 per 1,000 vehicle kilometres.

Table 4 Average tax revenue in Cyprus in 2013 (PPP adjusted)

| Vehicle type | Average infrastructure costs | | | | | |
|-----------------------------------|------------------------------|--|--|--|--|--|
| Passenger transport (€/1,000 pkm) | | | | | | |
| Passenger car | 41 | | | | | |
| Motorcycle | 32 | | | | | |
| Bus | 8 | | | | | |
| Freight transport (€/1,000 tkm) | | | | | | |
| HGV | 86 | | | | | |
| Vans (€/1,000 vkm) | | | | | | |
| Vans | 48 | | | | | |

For more information on the data used to estimate these results for Cyprus, see Annex A.

The total revenue from road transport taxes and charges in Europe in 2013¹⁰ is \in 365 billion (PPP adjusted), of which \in 79 billion are revenue from VAT on vehicle purchases and fuel. The main share of the transport related tax revenue (excl. VAT on vehicle purchases and fuel) is from passenger cars (about 71%), as is shown in Figure 9A. HGVs contribute about 15%, while vans, busses and motorcycles contribute 9%, 2% and 3% respectively. The large share of passenger cars in total tax revenue can be explained by the large share these vehicles have in total fleets and vehicle kilometres.

Figure 9 Total revenue breakdown per mode of transport and per tax/charge



^a Only transport related tax/charge revenue are considered, which means that revenue from VAT on vehicle purchases (excl. taxes) and fuel (excl. taxes) are not included.

The total revenue of transport taxes and charges consists mainly of fuel excise duties (49%) and VAT payments (30%). Fuel Excise Duty preponderance is linked to the fact that it is levied in every country for all vehicle categories. VAT on car purchases and fuel (both including taxes) is also charged in every country (but not on all vehicles), resulting in a relatively large share in total revenue. The scope of vehicle taxes and infrastructure charges is more limited (see Section 2.3) and their share in total revenue is therefore smaller (13% and 9%, respectively). Figure 10 shows that the contribution of the various taxes/charges to total revenue varies significantly between countries. In Estonia, for example, more than 99% of tax revenue comes from fuel taxes and VAT, while in Denmark they represent less than 45% of total revenue. In the latter country the revenue from vehicle taxes (i.e. registration and ownership tax) are high, as is the case for countries like The Netherlands, Malta and Ireland. In France, Croatia, Portugal, Slovenia and Austria infrastructure taxes contribute significantly to total road tax/charge revenue (10-20%).

¹⁰ As mentioned in Section 2.2, the revenue of (local) parking charges is not gathered for all EU countries. To give an impression of the share that this revenue can have in total revenue from taxes and charges, we have gathered data on parking charge revenue (from public parking places, both on- and off-street) in the Netherlands and the UK (from the national statistical agencies). In the Netherlands, this revenue was equal to € 566 million (PPP adjusted) in 2013, which is about 4% of the total road transport tax/charge revenue. The 2013 parking revenue in the UK is € 1,183 million (PPP adjusted), which is about 2.5% of the total revenue.

Figure 10 Share of different types of taxes in total road tax/charge revenue in the various EU Member States in 2013



In the majority of European countries, revenue from specific road transport taxes (i.e. registration and ownership taxes and fuel excise duties, VAT on registration taxes and fuel excise duties) contributes 5% to 10% of total tax revenue. In some countries (Bulgaria, Greece, Lithuania, Slovakia and Slovenia) this share is slightly higher (11% to 14.5%), while in France and Sweden the share is slightly lower (about 4%). If revenue from VAT on vehicle purchases and fuel is included as well, the contribution to total tax revenue range from 5% in Sweden to 20% in Slovenia.

Figure 11 Total road transport tax revenue as share of total tax revenue in 2013



Note:

- The total tax revenue is based on Eurostat data and includes revenue from national, regional and local taxes. Due to differences in scope, the total tax revenue presented by Eurostat may differ from the total revenue presented in national accounts.
- As infrastructure charges are not part of the total tax revenue (charge revenue is not considered tax revenue in public accounting), they are not included in the total road transport tax revenue as well.



Finally, the total revenue from specific road transport taxes/charges (including VAT on registration taxes and fuel excise duties) as share of GDP in 2013 is shown for the various EU countries in Figure 12. In most countries, the revenue collected from these taxes and charges is 2% to 3% of GDP. A main exception is Slovenia, where the revenue from road transport taxes and charges is more than 4% of GDP. Revenue from VAT on vehicle purchases and fuel is equal to 0.2% to 1.1% of GDP in the various EU countries.



Figure 12 Total revenue from road transport taxes and charges as % of GDP in 2013

2.5 Average revenue from road transport taxes and charges

In this section we present the average tax/charge revenue for the various vehicle categories, expressed in $\ell/1,000$ passenger kilometres for the passenger transport modes, $\ell/1,000$ tonne kilometres for HGVs and $\ell/1,000$ vehicle kilometres for vans¹¹. Three (complementary) explanations may account for the divergences between countries:

- differences in the actual tax/charge levels applied in the various countries
- differences in average mileage, fuel efficiency, occupancy rate (passenger modes of transportation) and average load (HGVs) between countries; e.g. if vans in one country are on average more fuel efficient than in another country, then the fuel tax revenue per vehicle kilometre will be lower in the former country.
- differences in price level (purchasing power) between countries; as we correct the revenue for differences in purchasing power in the various EU countries, the revenue in countries with a relatively low average price level (e.g. Romania) is corrected upwards, and vice versa (see Section 1.3 for a more detailed explanation).



¹¹ Graphs showing the average revenue in €/1,000 vehicle kilometres for passenger cars, motorcycles, busses and HGVs are presented in Annex D.

2.5.1 Passenger transport

Figure 13 presents the average tax/charge revenue for passenger cars in 2013. This revenue is highest in Luxembourg, which is mainly due to the relatively high fuel excise duty revenue and VAT on fuel per passenger kilometre. This is the result of the large level of foreign vehicles fuelling up in Luxembourg to take advantage of the relatively low fuel prices (see Figure 7). As these vehicles mainly use the fuel purchased in Luxembourg for passenger kilometres made in other countries, the fuel excise duty revenue per passenger kilometres in Luxembourg is relatively high. Tank tourism may have an impact on the fuel excise duty revenue of other countries as well (e.g. Austria, Germany and Ireland), but to a lesser extent than in Luxembourg.

Fuel excise duties are the main source of revenue for most EU countries with the exceptions of Denmark, Malta, The Netherlands and, to a lesser extent, Austria and Ireland. Most of these countries have rather high vehicle tax levels (per vehicle). In Malta, the relatively low annual mileages also contribute to high revenue of vehicle taxes per passenger kilometre.





For motorcycles, the highest average revenue of taxes/charges is found for Denmark and Austria. For Denmark, this is mainly due to the high level of registration taxes for motorcycles, while in Austria both registration and ownership taxes on motorcycles are relatively high. Relatively low average tax/charge revenue for motorcycles is found for Bulgaria, Estonia, Slovakia and Sweden. In these countries, no vehicle taxes and infrastructure charges are levied on motorcycles, resulting in relatively low average revenue.



Figure 14 Average revenue from taxes and charges for motorcycles in 2013



The average revenue from taxes and charges for busses (per passenger kilometre) is considerably lower than for passenger cars and motorcycles. In most countries, no or relatively low vehicle taxes are levied on busses, resulting in low total revenue.



Figure 15 Average revenue from taxes and charges for busses in 2013

2.5.2 Freight transport

The average tax revenue for HGVs (expressed in \notin /1,000 tkm) is shown in Figure 16. As for the other modes of transportation, fuel excise duty revenue account for the main part of the tax/charge revenue. However, in countries

with HGV charging schemes (Austria, Czech Republic, Germany, Slovakia, Slovenia) and countries with more general road charging schemes (e.g. France, Italy, Croatia) the contribution of tolls to total revenue is significant as well. In general, the average revenue in these countries is higher than in countries without a large-scale HGV road charging scheme. The very high average revenue found for Malta can be explained by the low number of tonne kilometres (due to low average trip lengths and low average loads), resulting in high vehicle taxes per tonne kilometre.





2.5.3 Vans

As for the passenger transport modes, the average tax and charge revenue from vans is highest in Denmark, which is mainly due to the high vehicle taxes in this country. In Greece the average tax/charge revenue from vans is high as well, which is due to the relatively high fuel excise duty on petrol (and the large share of petrol vans in Greece). Additionally, significant vehicle taxes (mainly ownership taxes) are levied on vans in Greece. The lowest average revenue is found in Belgium, Estonia and Lithuania. In these countries, vans are (almost) exclusively taxed by the fuel excise duty.



30

Figure 17 Average revenue from taxes and charges for vans in 2013





3 Infrastructure expenditures

3.1 Introduction

National, regional and local governments (and some private road owners¹²) spend significant amounts of money on expanding, renewing and maintaining the road infrastructure network in Europe. However, the size and composition of these expenditures differ widely over time and between Member States. In this chapter, we present and discuss the expenditures on road infrastructure in Europe for the period 1995-2013, taking into account these temporal and geographical differences.

This chapter provides answers on the second research question (and underlying sub questions) (see textbox below). In order to give these answers, we first briefly define road infrastructure expenditures (Section 3.2). Next, we present and discuss the evidence on investments in road infrastructure. The same is done for operational and maintenance (O&M) expenditures in Section 3.4.

Research question 2

What is the amount of the expenditure on road infrastructure in 2013 in the EU Member States?

This question consists of three sub questions:

- 1. Which expenditures on road infrastructure should be considered?
- 2. What has been the size of these different categories of road expenditures in 2013 and over time (1995-2013)?
- 3. Are there significant differences between EU Member States with respect to these expenditures and how can these differences be explained?

3.2 Defining road infrastructure expenditures

In this study, we define infrastructure as the physical and organisational network, which allows movements between different locations (HLG, 1999). These are roads, but also the organisation of the traffic (e.g. traffic management systems). Parking places are considered to be part of the road infrastructure as well. However, since the data availability on expenditures on (public) parking places in Europe is rather poor, we could not take them into account¹³.

The actual amounts of money extracted annually from public (or private) accounts to finance infrastructure are called infrastructure expenditures. VAT payments are not included in these figures. An overview of infrastructure

¹² E.g. operators of concessionary roads. As we look at the total expenditures on road infrastructure, spending by private agents should be considered as well.

¹³ Consultation of the European Parking Association made clear that expenditure data on parking places is not available at the EU level. The 11 country analyses carried out in this study (see Section 1.4) didn't provide any useful input on these expenditures either, except for the Netherlands. Therefore, we only present an estimation of the costs of Dutch parking places in the next chapter.

expenditures therefore provides an understanding of the direct impact of infrastructure on these budgets. As infrastructure expenditures are not directly comparable to tax/charge revenue data, infrastructure costs are estimated in this study as well (Chapter 4). This is further explained in the textbox below.

Expenditures vs. costs

Accounting for the total resources consumed by the construction, maintenance and operation of long life road infrastructures can either be done by simply summing up expenses or by using (real economic) costs. In contrast to expenditures, costs do take the financing costs or - regarded from a different point of view - the opportunity costs for not spending the resources for more profitable purposes, into account. Financing (or opportunity) costs are expressed by the interest on capital. As financing of road infrastructure is an issue for public bodies (as well as for private investors), full economic cost data are preferred to summing up expenditures in all cases.

Furthermore, infrastructure expenditures may vary widely over time, e.g. due to long planning and construction phases of big projects, which means that they cannot accurately reflect the actual costs caused by road transport to the infrastructure. For example, the investments in a road built in 2000 are zero in 2013, but as this road is used by vehicles in 2013 as well, part of these investments should be allocated to the vehicles in 2013. This can be done by applying a cost accounting approach (see Chapter 4).

In this study, we distinguish between investments and operational and maintenance (O&M) expenditures (ITF, 2013b):

- Investments: expenditures on the enhancement and renewal of the road infrastructure network. Renewal refers to major renovations increasing the performance of existing infrastructure assets or extending their previously expected service lives. These expenditures can be undertaken at any time and are not directly dictated by the condition of the asset.
- O&M expenditures: expenditures associated to 'ordinary' maintenance, i.e. maintenance that cannot be avoided, as the assets that are to continue to be used, are part of the O&M expenditures. These activities do not change the performance of the infrastructure asset, but simply maintain it in good working order or restore it to its previous condition in the event of breakdown. Operation expenditures are made to enable an efficient use of the infrastructure (e.g. lighting).

An (non-exhaustive) overview of the main elements of the various expenditure categories are given in the following textbox¹⁴.

Overview of main elements of road infrastructure expenditures Enhancement expenditures

- expenditures on preparation of investments in new roads or expansion of existing roads (e.g. feasibility studies);
- expenditures on building new roads (including expenditures on bridges, noise barriers, etc.);

expenditures on expanding existing roads.



¹⁴ Preferably all these elements are included in the infrastructure expenditure figures presented for the various countries in Sections 3.3 and 3.4. However, as for most countries, only data in aggregated form was available; we were not able to check to what extent all these elements were included.

Renewal expenditures

Large scale maintenance (economic lifetime > 2 years) of road surfaces, bridges, fly-overs and other infrastructure assets (e.g. noise barriers). These expenditures improve the performance of the asset and extend its economic lifetime.

Maintenance expenditures

- small repairs to the road surface, bridges/fly-overs and other constructions (economic life-time < 2 years), without improving the actual performance of the asset;
- maintenance of street lightning;
- maintenance of traffic signs and traffic lights;
- maintenance of road sides.

Operational expenditures

- traffic police;
- traffic management (systems);
- overhead costs infrastructure managers (housing, vehicles, energy, etc.);
- charging road tolls;
- cleaning icy/snowy roads;
- Lighting.

Assessing maintenance expenditures (both 'ordinary' maintenance and renewal), could be based on actual expenditures or on the so-called 'standard cost approach'. The latter refers to the expenditures related to the minimum package of maintenance (and operational) measures required to ensure the long-term physical and functional integrity of existing infrastructure under current conditions ('steady state level'). This approach corrects infrastructure spending, which is systematically below (or above) what is needed for maintenance. It hence provides a better starting point for the calculation of the infrastructure costs. Our analyses focus on the actual infrastructure expenditures, since no steady state level figures are available for most member states.

3.3 Investments in road infrastructure

3.3.1 Total investments

Table 5 gives an overview of the total investments in road infrastructure in the EU. Both the average annual investments in the period 1995-2013 and the investments in 2013 are presented (see Annex A for the data sources and a discussion on the reliability of the data). All figures are PPP adjusted to allow comparison between countries (see also Section 1.3). The unadjusted figures can be found in Annex E.

Table 5 Total investments in road infrastructure (mln €2013, PPP adjusted)

| Member State | Long-term (1995- 2013) average annual investments | Investments in 2013 | Ratio 2013 investments/long term average annual investments |
|----------------|---|---------------------|--|
| Austria | 1,978 | 1,648 | 83% |
| Belgium | 1,568 | 1,461 | 93% |
| Bulgaria | 509 | 709 | 139% |
| Czech Republic | 2,275 | 1,278 | 56% |
| Denmark | 829 | 746 | 90% |
| Germany | 10,198 | 9,340 | 92% |



| Member State | Long-term (1995- 2013) average annual investments | Investments in 2013 | Ratio 2013 investments/long term average annual investments |
|-----------------|---|---------------------|--|
| Estonia | 174 | 164 | 9 4% |
| Finland | 705 | 865 | 123% |
| France | 10,908 | 9,451 | 87% |
| Greece | 2,237 | 1,295 | 58% |
| Croatia | 1,356 | 850 | 63% |
| Hungary | 1,273 | 737 | 58% |
| Ireland | 1,287 | 710 | 55% |
| Italy | 8,640 | 3,215 | 37% |
| Latvia | 218 | 202 | 9 3% |
| Lithuania | 424 | 391 | 92 % |
| Luxembourg | 164 | 181 | 110% |
| Malta | 39 | 62 | 159% |
| The Netherlands | 5,068 | 5,124 | 101% |
| Poland | 4,414 | 4,068 | 92 % |
| Portugal | 2,900 | 384 | 13% |
| Romania | 5,765 | 7,534 | 131% |
| Slovakia | 664 | 485 | 73% |
| Slovenia | 656 | 154 | 23% |
| Spain | 9,302 | 3,951 | 42% |
| Sweden | 1,134 | 1,441 | 127% |
| United Kingdom | 6,304 | 6,400 | 102% |
| EU27 | 80,989 | 62,846 | 78% |

Note: As no reliable data on Cypriot road infrastructure investments was available, Cyprus is not shown in this table.

In most Member States, 2013 investments were below the long-term annual investments. For the EU27 as a whole, the 2013 investments were about 78% of the long-term annual investments. This decrease in total annual investments in road infrastructure is also seen in Figure 18. From 2007/2008 (the start of the economic crisis), road infrastructure investments strongly decreased in Western European countries (WEC)¹⁵, while they decreased later in Central and Eastern European countries (CEEC)¹⁶ (2011-2013). Figure 18 also shows the significant increase in road infrastructure investments between 2003-2011in Central and Eastern European Countries, reflecting efforts to meet rising needs for road network capacity. Additionally, extensive funding from European programmes (mainly Cohesion and Structural funds) came available for these countries, which were used to finance these efforts.



¹⁵ WEC includes: Austria, Belgium, Denmark, Germany, Finland, France, Greece, Ireland, Italy, Luxembourg, Malta, the Netherlands, Portugal, Spain, Sweden and the UK.

¹⁶ CEEC includes: Bulgaria, Czech Republic, Estonia, Croatia, Hungary, Latvia, Lithuania, Poland, Romania, Slovakia and Slovenia.
Figure 18 Investment index for road infrastructure in the period 1995-2013



Note: The indices are based on investments in ξ_{2013} .

When we zoom in on investment levels in the individual Member States over the period 1995-2013 (see Figure 19A and Annex E for more details), we see that the countries which show the highest growth rates in investments are mainly located in Central and Eastern Europe. Particularly in the period before 2007 (see Figure 19C) the investment levels in these countries rose significantly. Investment also increased significantly in Malta between 1995-2013, more particularly between 2005 and 2013 (see Annex E). In 2004 Malta joined the EU, which gave them access to EU funding for investments in road infrastructure. The largest decline in investment levels are found for Southern European countries like Portugal, Spain and Italy. Large-scale road infrastructure investment programmes were implemented in the nineties in these countries (EEA, 2002) resulting in a high base level of investments. But at the same time Portugal, Spain and Italy are among the European countries most severely affected by the economic crisis, resulting in significant cuts in road infrastructure investment budgets since 2008/2009 (see also Figure 19B).

Portugal, Spain and Italy were not the only countries to cut infrastructure investment in recent years: Figure 19B shows that investment levels in most European countries actually decreased over the period 2008-2013. In many of these countries, the economic crisis is probably (one of) the main explanation(s) for this trend. Bulgaria and Malta are exceptions, which is probably the result of major road investment programmes mainly financed through EU funds (which were not drastically cut during the crisis). In some new EU member countries (e.g. Romania, Lithuania), EU funding has contributed to relatively high investment levels over the period 2008-2013 as well (see Annex E). However, for these countries, the peak in investment levels was already in 2008/2009 and a decreasing trend in investment levels (partly explained by the economic crisis) is therefore found for these countries (Steer Davies Gleave, 2014). Finally, in countries with a mature road network



(e.g. Denmark, Germany, France, The Netherlands, UK) investment levels are as expected - rather stable over time. Over the period 2008-2013 investment levels only slightly decreased, indicating that the economic crisis had only a limited impact on investment budgets in these countries.

Figure 19 Change in investments over time









3.3.2 Investments as share of GDP

Although the investment needs for road infrastructure depend on a number of factors, such as the quality and age of the existing infrastructure and the geography of the country, showing the investments as percentage of Gross Domestic Product (GDP) may provide a useful benchmark for comparing investment levels between countries¹⁷. Figure 20 shows that till 2009, the investments as share of GDP remains guite stable, followed by a decline due to the economic crisis. Significant different patterns are found for Western European and Central and Eastern European countries. In Western European countries the investments as percentage of GDP slowly decline over the period 1995-2013. In this period, the investment share of GDP fall from 0.7% in 1995 to 0.4% in 2013. This decline is a continuation of the downward trend reported by ITF (2013a) for the period before 1995; the investment share of GDP in Western Europe declined from 1.5% in 1975 to 1.2% in 1980 to slightly below 1% in 1995. Recently (2009-2013), the decline has accelerated, which may be due to the economic crisis. Central and Eastern European countries show a completely different pattern. As the investment share of GDP for this group of countries is rather stable over the period 1995-2003, a sharp increase is identified for the period 2004-2011 (in line with the findings in Section 3.3.1).





In Figure 21, the long-term share of investments as percentage of GDP, is shown for individual EU countries (for more detailed data, see Annex E). For Western European countries, the share lies between 0.4 and 0.8%. The relative investment levels have only been higher in Portugal (i.e. 0.95%), which can be explained by large investment programmes in the nineties (mainly financed by EU funds) and particularly the first decade of this century (with a significant role for public private partnerships)¹⁸. As a result, the



¹⁷ Because of the dependency of investment needs for road infrastructure on a large number of factors it is not possible to define an absolute benchmark for road investments (e.g. 1% of GDP).

¹⁸ However, it is also mentioned in the literature that several investment projects are unnecessary and management and corruption in public-private partnerships are a matter of

quality of Portuguese road infrastructure was ranked as fourth worldwide by the World Economic Forum in 2013 (WEF, 2014). The investment levels in Portuguese road infrastructure (and hence the share of these investments in GDP) have declined significantly over the last years, and hence the long-term investment level of Portugal is converging closer to the WEC average.



Figure 21 Investments as share of GDP (average shares for the period 1995-2013)

The share of investments in road infrastructure in GDP lies significantly higher in CEEC countries than in WEC countries: between 0.7 and 1.3%. In two countries, the road infrastructure investment share of GDP is even higher: 2.3% in Romania and 1.7% in Croatia. However, the explanations for these relatively high investment levels (compared to other CEEC countries) differ significantly between both countries.

In Romania, inefficiencies in the planning and construction phase are considered the main reason for the relatively high investment levels¹⁹. A lack of (government) expertise in construction management (partly caused by a lack of appropriately skilled labour), a lack of competition between construction companies, and a relatively high level of corruption results in relatively high investment levels (IMF, 2015); (Ove Arup & Partners , 2010); (WBG, 2011). The relatively high investments in Croatia, on the other hand, are mainly caused by very ambitious investment programmes in the second half of the nineties and the first decade of this century, aimed to reconstruct



great concern (Pereira & Pereira, 2015), which may also have contributed to relatively high investment levels.

¹⁹ These type of inefficiencies are not only found in Romania, but also in other European countries. However, according to the literature, these inefficiencies are larger in Romania than in most other European countries.

war damaged infrastructure, to modernise existing and to build new infrastructure²⁰ (WBG, 2004); (WBG, 2015).

Relation between income levels and investment levels

The difference between Western European Countries and Central and Eastern European countries suggests a relationship between road infrastructure investments and the level of income. This relationship is influenced by the correlation between Member States income levels and access to EU funding. Figure 22 shows the investments as a percentage of GDP against GDP per capita using panel data for 27 EU countries for the period 1995-2013. Higher levels of investments (> 1% of GDP) are in general only found in countries with an average GDP/capita lower than € 20,000. A possible explanation may be that emerging economies need an increase in road capacity, while production becomes relatively less transport intensive if economies become more developed²¹ (McKinnon, 2006), weakening the link between GDP growth and infrastructure investments (ITF, 2013a). In fact, at GDP/capita levels above \notin 20,000, there seems to be no clear correlation between investment levels and income levels. Assuming that GDP/capita is an important driver of transport demand and hence a good proxy for actual needs for investments in road infrastructure, this may suggest that investment levels are guided by other factors (e.g. historical budget levels or budget allocation procedures) than actual investment needs as well (see ITF, 2013a).

Figure 22 Relationship between income levels and investments in road infrastructure



Note: This graph is based on panel data for the EU27 for the period 1995-2013. Every dot represents the combination of investments as share of GDP and GDP per capita for a specific country (e.g. Belgium) in a specific year (e.g. 2002). In total, there are 513 dots (27 countries x 19 years).



An illustration of the ambition level of these programmes is the large increase in the length of motorways in Croatia over the last twenty years: from 36 km in 1995 to 1,296 kilometre in 2013. This implies that the motorway density has increased to almost 23 km/1,000 km², which is just above the average motorway density in WECs (21.9 km/1,000 km²) and far above the density in CEECs (6.9 km/1,000 km²).

²¹ In other words, the ratio of transport movements to GDP declines.

3.4 Operational and maintenance expenditures

3.4.1 Total O&M expenditures

The total O&M expenditures on road infrastructure in the EU is given in Table 6. Both the average annual O&M expenditures for the period 1995-2013, as well as the expenditures for 2013 are presented (see Annex A for the data sources and a discussion on the reliability of the data). Total O&M expenditures unadjusted for PPP can be found in Annex E.

Table 6 Total O&M expenditures on road infrastructure (mln €2013, PPP adjusted)

| Member State | Long-term (1995-2013) average annual O&M expenditures | O&M expenditures in 2013 | Ratio 2013 O&M expenditures/long term average annual O&M expenditures |
|-----------------|--|--------------------------------|---|
| Austria | 2,807 | 2,113 | 75% |
| Belgium | 1,029 | 1,041 | 101% |
| Bulgaria | 236 | 285 | 117% |
| Czech Republic | 1,060 | 990 | 93% |
| Denmark | 887 | 656 | 74% |
| Germany | 5,436 | 6,879 | 127% |
| Estonia | 136 | 148 | 109% |
| Finland | 1,187 | 825 | 69% |
| France | 6,310 | 6,559 | 104% |
| Greece | 452 | 342 | 76% |
| Croatia | 589 | 602 | 102% |
| Hungary | 1,650 | 1,471 | 89 % |
| Ireland | 256 | 297 | 116% |
| Italy | 11,486 | 5,432 | 47% |
| Latvia | 152 | 193 | 127% |
| Lithuania | 320 | 282 | 88% |
| Luxembourg | 55 | 62 | 111% |
| Malta | 11 | 6 | 60% |
| The Netherlands | 1,716 | 1,448 | 84% |
| Poland | 839 | 761 | 91% |
| Romania | 761 | 576 | 76% |
| Portugal | 2,677 | 2,606 | 97 % |
| Slovakia | 303 | 393 | 129% |
| Slovenia | 272 | 260 | 95% |
| Spain | 3,945 | 2,967 | 75% |
| Sweden | 1,050 | 1,377 | 131% |
| United Kingdom | 5,075 | 3,473 | 68% |
| EU27 | 50,699 | 42,040 | 83% |

Note: As no reliable data on Cypriot road infrastructure investments was available, Cyprus is not shown in this table.

The 2013 O&M expenditures were slightly below long-term annual expenditures in the EU27 (about 7%). This is due to lower O&M expenditure levels in Western European countries. O&M expenditures in these countries are slightly increasing in the period 1995-2006 (with a brief decline in the beginning of this century), but are steadily decreasing in the years following (see Figure 23). In Central and Eastern European countries, O&M expenditures have been rising sharply from 1999 till 2007. In the following years,



expenditures are declining again (e.g. due to the economic crisis), although O&M expenditures in these countries are still significantly higher in 2013 than in 1995. In Annex D, more detailed figures (per country) on the development of O&M expenditures over time are given.



Figure 23 O&M expenditure index for road infrastructure for the period 1995-2013

The sharp increase in O&M expenditures between 1995-2013 (and particularly between 1995 and 2007) is found in every Central and Eastern European country, with the exception of Lithuania (see Figure 24A and Figure 24C). In many Western European countries, O&M expenditures have remained rather stable over time, although there are some countries (Denmark, Italy, Finland, UK and Austria) for which a considerable decrease has been found.

The economic crisis seems to have a significant impact on the O&M budgets in many Eastern and Southern European countries. As is shown in Figure 24B, the O&M expenditures in these countries show a large decrease between 2008 and 2013. According to Steer Davies Gleave (2014), these budget reductions have disproportionately affected local authorities, potentially increasing the gap in road quality between national and local roads. In contrast, in some Western European countries (e.g. France, Sweden, Germany) O&M expenditures have increased over the period 2008-2013. This is, at least partly, explained by governmental programmes aimed to stimulate economic growth by additional investments in road maintenance (e.g. in France several road infrastructure investment plans were implemented to boost economic growth (Steer Davies Gleave, 2014). Finally, the large increase in O&M expenditures for Ireland is probably methodological, as there seems to be a break in 2009 in the statistical data on O&M expenditures (see Annex A.5).



Note: The indices are based on investments in €2013.

Figure 24 Change in O&M expenditures over time



C) Between 1995 and 2007



Concerns are raised in many countries on underfunding road maintenance, leading to a decrease in quality of existing roads and indirectly to negative impacts on the economic competitiveness (ITF, 2013a). This concern could be investigated by comparing actual maintenance expenditures with steady state levels of expenditures, which keep the infrastructure on a long-term good quality level. Data is missing to carry out this analysis for all EU countries, but



some data is available for Ireland and The Netherlands (see following textbox). For Ireland, a significant risk on underfunding of road maintenance in the coming years is identified by Ireland's Department of Transport. For the Netherlands the data is less conclusive; 2013 expenditures on maintenance and renewal of national roads were far below steady state levels (probably due to the economic crisis), but this was preceded by a period with relatively high expenditure levels.

Steady state level expenditures in The Netherlands and Ireland

The steady state level maintenance and renewal expenditures (i.e. the level of expenditures guaranteeing sufficient long-term quality levels of roads) for national roads in the Netherlands has been estimated based on DVS data (DVS, 2007).

In Figure 25 these steady state level expenditures are compared with actual expenditures over the period 1995-2013. In the period 1995-2005, the actual expenditures were (mostly) below the steady state level expenditures, which could imply an underinvestment in road maintenance. However, in the period 2005-2010 expenditures are significantly above steady state levels, which could probably be explained by recovering overdue maintenance of national roads. In the period 2011-2013 maintenance and renewal expenditures have fallen below steady state levels again (due to the economic crisis).





For Ireland, the Department of Transport, Tourism and Sport (2014) estimates the steady state level of maintenance and renewal expenditures for the period up to 2020 on \in 1.6 billion per annum. The current budget provides \in 1.33 billion per annum (including the budget for investments as well), which is over \in 260 million short of the funding required to maintain the existing system in an adequate condition.

Another (less direct) indicator of underfunding of road maintenance is the share of O&M expenditures in total road expenditures. In the EU27, this share has been relatively constant over time: over the last twenty years it has varied between 34% and 41% (see Figure 26). However, the variation in Central and Eastern European countries is much larger than in Western European countries.



In Central and Eastern European countries there seems to have been a long-term increase over the period 1995-2008. However, in 2008 the share of O&M expenditures in total road transport expenditures sharply decreased (particularly in countries like Hungary, Poland and Lithuania). The cut in O&M expenditures has been larger compared to investment expenditures, probably because O&M expenditures are financed from national budgets for a larger extent (as EU funding contributes significantly to the investment budgets in these countries). Furthermore, investment expenditures are planned in long-term projects and can be less easily adjusted, compared to the more short-term planned O&M expenditures.





3.4.2 O&M expenditures as share of GDP

In line with the decreasing share of road infrastructure investments in GDP, the share of O&M expenditures in GDP shows a slowly decreasing trend in the EU27 over the period 1995-2013, mainly explained by the developments in Western European countries (see Figure 27). For Central and Eastern European countries, an increasing trend in the O&M expenditure share is found for the period 1998-2007, showing the rising need for good quality roads in these countries. However, due to the economic crisis, O&M budgets have been heavily cut (Steer Davies Gleave, 2014), resulting in decreasing shares of these expenditures in GDP. More detailed figures (per country) can be found in Annex F.5.



Figure 27 Development in share of O&M expenditures in GDP over time



The higher share of O&M expenditures in GDP in Central and Eastern European countries compared to Western European countries is also shown by Figure 28. As for investments, the share of O&M expenditures in GDP in Romania is slightly higher than in other Central and Eastern European countries. The inefficiencies in the road infrastructure sector, as discussed in Section 3.3.2, is an important explanation for this finding. For most Western European countries, the share of O&M expenditures in GDP is smaller than or equal to 0.5%, with the exception of Austria, Finland and Italy. Possible explanations for the higher shares in these countries will be discussed in the next section.



Figure 28 O&M expenditures as share of GDP (average shares for the period 1995-2013)



3.4.3 O&M expenditures per kilometre road network length

Figure 29 presents the O&M expenditures per kilometre road network²² for each EU Member State. These expenditures are highest in Austria, for which there may be several (complementary) reasons. First, the network complexity may be relatively high in Austria due to a rugged landscape and a relatively large number of bridges and tunnels. In Austria, tunnels and bridges make up about 17% of the total length of the main roads, while they only make 1-2% of the networks in countries such as France, Denmark or Ireland (CEDR, 2010). As it costs on average 10 times more to maintain 1 km of bridge (and even more for 1 km of tunnel) than to maintain 1 km of plain road (CEDR, 2010), the high complexity of the Austrian network will be a main reason for the relatively high O&M expenditures. Secondly, the expenditures on winter maintenance of Austrian roads will be above EU average.

Third, the quality of Austrian roads is relatively high (according to WEF (2014), Austria is ranked 6th worldwide with respect to the quality of its road network). And fourth, the traffic density of Austrian (main) roads is relatively high, resulting in relatively high use-dependent O&M expenditures.

As expected, O&M expenditures per kilometre road network length are relatively high in Croatia and Romania (due to high total O&M expenditures, see Section 3.4.1). High traffic densities and relatively high road quality may be an important reason for the high O&M expenditures on roads in the Netherlands and the UK (and to a lesser extent Denmark).

O&M expenditures on Italian roads are relatively high as well. An important explanation for this finding may be the relatively high level of network complexity in Italy. According to CEDR (2010) about 8% of the length of the main road network in Italy consists of tunnels and bridges.

Finally, the relatively low level of O&M expenditures in countries like Finland, Estonia, Latvia, Lithuania and (to a lesser extent) Sweden is linked to the large share of unpaved roads with very low traffic densities. In these countries the majority of the O&M expenditures are spent on a small share of the total road network, resulting in relatively low average O&M expenditures per kilometre road.



²² In this study, road network length refer to road length and not lane length. Dual-carriageways are considered one road (i.e. the length of both directions are not summed up).





3.5 Total infrastructure expenditures

3.5.1 Total road infrastructure expenditures

The total road infrastructure expenditures are the sum of the investments and O&M expenditures. Table 7 presents these total expenditures (PPP adjusted). Both the average annual expenditures in the period 1995-2013 and the expenditures in 2013 are presented. Total expenditures unadjusted for PPP can be found in Annex E.

| Table 7 | Total expenditures on road | d infrastructure (mln : | € ₂₀₁₃ , PPP adjusted) |
|---------|----------------------------|-------------------------|-----------------------------------|
| | | | |

| Member State | Long-term (1995-2013) average annual expenditures | Expenditures in 2013 | Ratio 2013 expenditures/long term average annual expenditures |
|----------------|---|-------------------------|--|
| Austria | 4,785 | 3,761 | 79 % |
| Belgium | 2,597 | 2,503 | 96 % |
| Bulgaria | 753 | 994 | 132% |
| Czech Republic | 3,335 | 2,268 | 68% |
| Denmark | 1,716 | 1402 | 82% |
| Germany | 15,634 | 16,219 | 104% |
| Estonia | 310 | 312 | 101% |
| Finland | 1,892 | 1,689 | 89 % |
| France | 17,218 | 16,010 | 93% |
| Greece | 2,688 | 1,637 | 61% |
| Croatia | 1,945 | 1,452 | 75% |
| Hungary | 2,923 | 2,208 | 76% |
| Ireland | 1,543 | 1,007 | 65% |
| Italy | 20,127 | 8,647 | 43% |
| Latvia | 370 | 395 | 107% |



| Member State | Long-term (1995-2013) average annual expenditures | Expenditures in 2013 | Ratio 2013 expenditures/long term average annual expenditures |
|-----------------|---|-------------------------|--|
| Lithuania | 744 | 673 | 90% |
| Luxembourg | 220 | 243 | 111% |
| Malta | 50 | 69 | 138% |
| The Netherlands | 6,783 | 6,572 | 97 % |
| Poland | 5,253 | 4,829 | 92% |
| Portugal | 3,661 | 959 | 26% |
| Romania | 8,442 | 10,140 | 120% |
| Slovakia | 968 | 877 | 91% |
| Slovenia | 928 | 413 | 45% |
| Spain | 13,247 | 6,917 | 52% |
| Sweden | 2,184 | 2,818 | 129% |
| United Kingdom | 11,379 | 9,872 | 87% |
| EU27 | 131,696 | 104,886 | 80% |

The total infrastructure expenditures in 2013 in Europe were about \leq 105 billion, which is below the long-term average annual expenditure level of about \leq 130 billion. This decrease in total annual expenditures in road infrastructure expenditures is also seen in Figure 30. This is in line with the separate trends we identified for investments and O&M expenditures. We refer to Section 3.3 and 3.4 for the main explanations for these trends.

Figure 30 Total road infrastructure expenditure index for the period 1995-2013



3.5.2 Total infrastructure expenditures as share of GDP

The share of total road infrastructure expenditures in GDP shows a decreasing trend in the EU27 over the period 1995-2013, mainly explained by the developments in Western European countries (see Figure 31). In Central and Eastern European countries, an increasing trend is found for the period 1995-2009 (up to 2.6% in 2009). In the most recent years, road infrastructure

expenditures as share of GDP is decreasing in these countries as well, mainly explained by the economic crisis.



Figure 31 Development in share of total road infrastructure expenditures in GDP over time

As shown by Figure 31 and Figure 32, the share of total road infrastructure expenditures in GDP is higher in Central and Eastern European countries compared to Western European countries. The explanations for this finding were discussed in detail in Sections 3.3 and 3.4.



Figure 32 Total road infrastructure expenditures as share of GDP (average shares for the period 1995-2013)



4 Infrastructure costs

4.1 Introduction

The construction, maintenance and operation of road infrastructure requires significant economic resources, both in terms of capital and labour. In this chapter, we account for the use of these resources by estimating the infrastructure costs of road transport in the EU. The main uncertainties associated to infrastructure expenditures (see Section 1.4 and Annex A) are reflected in infrastructure costs as well. Additionally, uncertainties with respect to the allocation of total infrastructure costs to the various vehicle types should be considered as well.

In this chapter we answer research question 3 and underlying sub questions (see textbox below). To provide the answers to these questions, we first discuss the methodology to estimate infrastructure costs (Section 4.2). Next, the estimated total and average costs are presented for all EU Member States in Sections 4.3 and 4.4.

Research question 3

What are the road infrastructure costs in 2013 in the EU Member States?

This question consists of three sub questions:

- 1. How are infrastructure costs estimated based on expenditure data?
- 2. What is the total amount of road infrastructure costs in 2013 in the EU? How can main differences between Member States be explained?
- 3. Which share do the various vehicle categories have in the total infrastructure costs?

4.2 Methodology to estimate the infrastructure costs

4.2.1 Defining infrastructure costs

Infrastructure costs can be defined as the direct expenses, plus the financing costs or - regarded from a different point of view - the opportunity costs for not spending the resources for more profitable purposes (Fraunhofer-ISI ; CE Delft, 2008). Financing or opportunity costs are expressed by the interest on capital, where the interest rates vary with the legal status of the investor.

In this study we consider four types of infrastructure costs²³:

- Enhancement costs: All costs of new infrastructure or expansion of existing infrastructure with respect to functionality and/or lifetime.
- Renewal costs: All costs associated to the renewal of (parts of) the infrastructure. The renewed (parts of) the infrastructure will at least have a lifetime of more than 1-2 years.
- Maintenance costs: These costs refer to the costs of 'ordinary' maintenance. These are relatively minor repairs with an economic lifetime of less than 1 to 2 years.

²³ In Chapter 3 enhancement and renewal expenditures are combined as investments and maintenance and operational expenditures as O&M expenditures.

- *Operational costs:* These costs refer to the costs of the organisation of an efficient use of the infrastructure.

Infrastructure costs can be classified by the way they are influenced by the infrastructure usage, i.e. transport volumes. According to this classification, we define the following types of costs (Ecorys; CE Delft, 2006):

- Variable costs: Costs that vary with transport volumes while the functionality of the infrastructure remains unchanged. Part of the maintenance and renewal costs belongs to this cost category.
- Fixed costs: Costs that do not vary with transport volumes while the functionality of the infrastructure remains unchanged, or costs that enhance the functionality of the infrastructure. Construction costs and operational costs are examples of fixed infrastructure costs. Also some of the maintenance and renewal costs are (partly) fixed costs²⁴.

In this study we will use the distinction between variable and fixed infrastructure costs for the allocation of the costs to different vehicle types (see Section 4.2.4).

4.2.2 Estimating infrastructure costs: a general overview

To estimate infrastructure costs, a top-down approach is applied (see Figure 33). First, the total costs of the road network in a country are estimated, based on data on the annual expenditures on road infrastructure (see Chapter 3). These annual expenditures are capitalised to estimate the total infrastructure costs (see Section 4.2.3). In the next step, the total infrastructure costs are allocated to the various vehicle modes by applying relevant cost drivers (based on the so-called equivalency factor method). This is explained in more detail in Section 4.2.4.

Figure 33 General approach to estimate infrastructure costs



²⁴ E.g. street lighting, traffic signs, etc. (CE Delft, 2008).



4.2.3 Computation of total infrastructure costs

Different methodologies are used to estimate enhancement and renewal costs on the one hand, and operation and maintenance costs on the other hand.

Enhancement and renewal costs

The estimation of both the enhancement and renewal costs is based on the Perpetual Inventory Method (PIM)²⁵. The PIM calculates the annual depreciation cost by distributing the initial investments over the lifetime of the infrastructure. In addition to the depreciation cost, interest/financing costs are estimated by using an appropriate interest rate. The sum of depreciation and financing costs equal enhancement and/or renewal costs. The calculation of depreciation and financing costs is illustrated by a simplified example in the textbox below.

Example: calculating investment costs

Assume that in the period 2001-2010 annually \leq 1,000 is invested in a road and that each investment has an expected lifetime of 10 years. Furthermore, we assume an interest rate of 5%.

To find the total depreciation costs for this road in 2010, we have to sum up the depreciation costs of each individual investment that has not yet been fully depreciated. If we apply a simple linear depreciation approach (constant depreciation rate over the lifetime of the investment), the annual depreciation cost for each individual investment is equal to \notin 100, and hence the total depreciation costs in 2010 are equal to \notin 1,000.

To calculate the financing costs, we first have to calculate the economic value of the investments in 2010. The investment done in 2001 has been depreciated for 90%, so its economic value in 2010 is only \in 100. By the same reasoning it can be calculated that the economic value in 2010 of the investment done in 2002 is \in 200, and so on. The total economic value of the road in 2010 can be found by summing the economic values of the individual investments: \in 5,500. By multiplying this total economic value with the interest rate (5%), the annual financing costs in 2010 are calculated. These are equal to \in 275.

The total investment costs in 2010 are equal to \notin 1,000 + \notin 250 = \notin 1,250. Compared to the average annual expenditures (\notin 1,000), these costs are considerably higher.

The PIM is based on the following assumptions:

 Depreciation approach: Different depreciation approaches could be used for the estimation of the enhancement and renewal costs. In this study, we use an annuity approach, which assumes constant annual costs (depreciation + financing costs). Another often used approach is a linear approach, which assumes constant depreciation costs (and hence diminishing total costs). CE Delft (2008) shows that the differences in the results of both approaches are rather limited.



²⁵ An alternative method to estimate the construction and renewal costs is by assessing future financing needs of the present network (Synthetic method). In this approach, for every type of infrastructure asset, a replacement value, reflecting its dimensioning, load, location and the latest technical standards and specifications, is estimated. Considering the age, past and projected traffic loads, and the physical condition of the asset, depreciation and interest costs are calculated, similar to the PIM approach. As mentioned by Fraunhofer-ISI & CE Delft (2008) and ITF (2013b), the Synthetic Method is more a decision support tool than the PIM approach, as it indicates the amount of money to be raised in order to maintain the quality of the network at a certain level. However, the PIM approach is more closely related to the common philosophy of public accounting and therefore more appropriate to use in this study.

- Depreciation period: The period over which the investment costs are depreciated depends on the assumed life expectancy of the infrastructure. In theory, different values of life expectancy should be assumed for the various infrastructure assets (e.g. 10 to 15 years for equipment, 90 to 100 years for earthworks). However, due to a lack of detailed distinction between various infrastructure assets in statistical databases and public accounts, applying different values of life expectancy is not possible. Therefore, we assume an average depreciation period of 35 years, which can be considered as the EU average life expectancy of infrastructure assets²⁶ (Fraunhofer-ISI; CE Delft, 2008).
- Interest rate: we use an interest rate of 4% for all EU Member States.
- Inflation correction: as mentioned in Section 1.3, we express all figures in this report in euro price level 2013. All historic infrastructure expenditures are corrected for inflation. Additionally, figures are adjusted for PPP.

Information on the share of enhancement and renewal expenditures in total investments was available for a few countries (see Table 8). Based on these data, we were able to estimate the average shares of enhancement and renewal expenditures in total investments (75% and 25%, respectively). These were used as default values for the countries for which this data was not available. The distinction between enhancement and renewal costs is needed for the allocation of the investment costs to the various modes (see Section 4.2.4).

Table 8Long-term average shares of enhancement and renewal expenditures in total investments
(averaged over all roads)

| Country | Enhancement expenditures | Renewal expenditures |
|-------------------------|--------------------------|----------------------|
| Austria | 84% | 16% |
| Netherlands | 68% | 32% |
| Poland | 70% | 30% |
| Selected default values | 75% | 25% |

Operation and maintenance costs

All the expenditure elements with a lifetime below one or two years (operation, management, minor repairs) are not capitalised according to the PIM approach. Instead, these running costs are taken directly into account when computing total infrastructure costs. This implies that no capitalisation approach is needed for the operation and maintenance costs of transport infrastructure; these can be directly based on the expenditures in 2013.

As mentioned in Section 3.2, maintenance costs would preferably be based on expenditures according to the standard cost approach. Since these expenditures correct for possible under-investments (or over-investments) for infrastructure maintenance, they better reflect the real costs caused by infrastructure users. However, data on expenditures according to the standard cost approach is often not available. In this study we only use maintenance expenditure based on the standard costs approach for the Netherlands. For all other countries we use the actual maintenance expenditures to estimate the maintenance costs.

For road transport, Fraunhofer-ISI & CE Delft (2008) show - based on a review of several European studies - that the average depreciation period of infrastructure assets is approximately 35 years.

For many countries only the total O&M costs are available. To allocate these costs to the various vehicle types, a distinction between operation and maintenance costs is required (see Section 4.2.4). Based on average figures gathered in the EU countries (see Table 9) where most data is available, we were able to estimate default shares of operation and maintenance expenditures in total O&M expenditures (45% and 55%, respectively). These default shares were applied to countries where the data was not available.

Table 9 Relative shares of operation and maintenance expenditures in total O&M expenditures (averaged over all roads)

| Country | Operation | Maintenance |
|--------------------------|-----------|-------------|
| Austria (only motorways) | 33% | 67% |
| Germany | 48% | 52% |
| The Netherlands | 47% | 53% |
| Poland | 29% | 71% |
| UK | 39% | 61% |
| Selected default values | 45% | 55% |

4.2.4 Allocation of infrastructure costs

The total infrastructure costs are allocated to the various vehicle types, based on the equivalency factor method (see for example (CE Delft; VU, 2014); (ProgTRans/IWW, 2007); (ITS, 2000). This method defines certain proportionality factors (cost drivers) for each vehicle type and cost category which express the responsibility or the causation of the vehicles for the level of total costs (Fraunhofer-ISI; CE Delft, 2008). This approach is illustrated in Figure 34.



Based on a literature review, the following proportionality factors are defined (for more detailed information, see Annex G):

 Enhancement costs are assumed to be 90% capacity dependent, as enhancement of roads is applied once their capacities are too low.
 These capacity dependent costs are allocated based on Passenger Car Equivalents (PCE) kilometres²⁷. The remaining 10% of the enhancement



²⁷ Indicator measuring the impact that a single vehicle has on traffic variables (e.g. speed, density) compared to a single passenger car.

costs are assumed to be weight dependent (e.g. the type and cost of pavement materials used depends on the assumed number of heavy duty vehicles using the road) and are allocated based on axle load kilometres (4th power rule²⁸). Enhancement costs are considered 100% fixed.

- Renewal costs are assumed to be partly capacity (40%) and partly weight dependent (60%) as well. Again, the capacity dependent costs are allocated based on PCE kilometres, and the weight dependent costs based on axle load kilometres. Furthermore, based on detailed data for the Netherlands (CE Delft and VU, 2014), it is assumed that 60% of the renewal costs are variable and 40% are fixed.
- Maintenance costs; based on detailed data for the Netherlands it is assumed that 30% of the maintenance costs are variable and 70% are fixed. The variable maintenance costs (e.g. road pavement damages) are fully weight dependent and therefore allocated based on axle load kilometres. The fixed maintenance costs (e.g. road signals, maintenance of road sides, etc.) are allocated based on the approach presented by ProgTrans/IWW (2007) for Germany (and also applied by CE Delft (2008) and CE Delft and VU (2014) for the Netherlands): 50% of the costs are allocated based on PCE-kms, 35% based on vehicle kilometres and 15% are allocated to HGVs.
- *Operational costs* are considered fully fixed and are allocated based on vehicle kilometres (30%) and PCE-kilometres (70%).

Table 10 summarises the proportionality factors used to allocate the various infrastructure cost categories.

Table 10 Summary proportionality factors

| Cost category | Proportionality factor |
|----------------------------|--|
| Enhancement costs | PCE kilometres (90%) |
| | 4th power axle load kilometres (10%) |
| Renewal costs | PCE kilometres (40%) |
| | 4th power axle load kilometres (60%) |
| Variable maintenance costs | 4th power axle load kilometres (100%) |
| Fixed maintenance costs | PCE kilometres (50%) |
| | Vehicle kilometres (35%) |
| | Allocated to HGVs (15%) |
| Operation costs | Vehicle kilometres (30%) |
| | PCE kilometres (70%) |

To allocate the infrastructure costs based on the proportionality factors defined, assumptions have to be made on some vehicle kilometres. The main assumptions are summarised in Table 11.

²⁸ Based on extensive tests it has been found that road damages are proportional to the 3rd to 4th power of the vehicle's axle load (Doll, 2005). For that reason, weight dependent costs are allocated based on 4th power axle load kilometres. The average axle load of a vehicle depends on its mass, the number of axles and the axle dimension (single, tandem or tridem axles). More specifically, the 4th power axle load is equal to: $\{K_i \times (\frac{A}{10})\}^4$, where A is the actual axle load, i the number of axle groups and K a correction factor for the axle configuration (K = 1 for a single axle, 0.6 for a tandem axle and 0.45 for a tridem axle).

Table 11 Summary of EU average vehicle characteristics

| Vehicle type | Passenger Car | Average mass | Average | 4 th Power |
|---------------|------------------|--------------|-----------------|------------------------|
| | Equivalent (PCE) | (ton) | number of axles | axle load |
| Passenger car | 1 | 1.3 | 2 | 3.4 x 10 ⁻⁵ |
| Motorcycle | 0.5 | 0.4 | 2 | 3.2 x 10 ⁻⁷ |
| Bus | 2 | 15 | 2 | 0.6 |
| Van | 1.2 | 2 | 2 | 2.0 x 10 ⁻⁴ |
| HGV | 3 | 15 | 2.6 | 0.1 |

Note: Country specific figures have been applied for average mass of passenger cars and HGVs, and for the number of axles of HGVs (figures shown in this table are EU averages). For the other vehicle characteristics EU average figures have been used. No data on the average number of axles of buses is available. Based on Fraunhofer-ISI and CE Delft (2008), we assumed that buses have two axles. The impact of this assumption on the results are assessed by applying a sensitivity analysis, in which we assume an average number of axles of 2.3. The results of this sensitivity analysis are shown in footnotes 3130 to 32, and 36.

Sources: Fraunhofer-ISI and CE Delft (2008), Eurostat; adaptation by CE Delft.

4.3 Total infrastructure costs

Based on the expenditure data discussed in Chapter 3, we have applied the methodology as discussed in Section 4.2.3 to estimate the total road infrastructure costs. Table 12 provides these costs for the EU Member States in 2013. These costs y are estimated at \notin 178 billion.²⁹ The main part of these costs (83%) are fixed, while the remaining part are considered variable.

| Member State | Total | Fixed | Variable |
|-----------------------|----------------------|----------------------|----------------------|
| | infrastructure costs | infrastructure costs | infrastructure costs |
| Austria | 6.5 | 5.3 | 1.2 |
| Belgium | 3.9 | 3.3 | 0.6 |
| Bulgaria ^b | 0.9 | 0.7 | 0.2 |
| Czech Republic | 4.3 | 3.6 | 0.7 |
| Denmark | 2.6 | 2.1 | 0.5 |
| Germany | 28.9 | 24.1 | 4.8 |
| Estonia | 0.5 | 0.4 | 0.1 |
| Finland | 2.5 | 2.1 | 0.4 |
| France | 24.4 | 20.7 | 3.7 |
| Greece | 3.1 | 2.6 | 0.5 |
| Croatia | 2.1 | 1.7 | 0.4 |
| Hungary ^b | 3.9 | 3.3 | 0.6 |
| Ireland | 2.0 | 1.7 | 0.3 |
| Italy | 22.4 | 19.0 | 3.4 |
| Latvia | 0.7 | 0.6 | 0.1 |
| Lithuania | 1.0 | 0.8 | 0.2 |
| Luxembourg | 0.35 | 0.30 | 0.05 |

Table 12 Total infrastructure costs in 2013 (billion €2013, PPP adjusted)^a

²⁹ The costs of parking places have not been considered for each country in this study, because of a lack of reliable data on expenditure data (see Section 3.2). However, from CE Delft and VU (2014) it is known that the share of the costs of (both on- and off-street) public parking places in total infrastructure costs of road transport is about 17.5%, implying that the total costs of parking places in the Netherlands in 2013 are about € 1.9 billion. Roughly 80% of these costs can be attributed to on-street parking places.

| Member State | Total | Fixed | Variable |
|-----------------|----------------------|----------------------|----------------------|
| | infrastructure costs | infrastructure costs | infrastructure costs |
| Malta | 0.05 | 0.04 | 0.01 |
| The Netherlands | 9.1 | 7.3 | 1.8 |
| Poland | 8.2 | 7.1 | 1.1 |
| Portugal | 4.6 | 3.7 | 0.7 |
| Romania | 9.1 | 7.6 | 1.5 |
| Slovakia | 1.3 | 1.1 | 0.2 |
| Slovenia | 1.0 | 0.8 | 0.2 |
| Spain | 17.3 | 14.1 | 3.2 |
| Sweden | 3.1 | 2.6 | 0.5 |
| United Kingdom | 14.9 | 12.1 | 2.8 |
| EU27 | 178.4 | 148.8 | 29.6 |

^a As no reliable data on Cypriot road infrastructure investments was available, no infrastructure costs were estimated for Cyprus.

^b The infrastructure cost for Bulgaria and Hungary are based on incomplete expenditure data (see Annex A).

About 54% of the total infrastructure costs in the EU are caused by passenger cars, as is shown in Figure 35. This relatively large contribution is explained by the large share passenger cars have in total vehicle kilometres in the EU. HGVs are responsible for about 21% of the infrastructure costs, while busses and vans cause about 15% and 9% of the total costs³⁰.

Figure 35 Allocation of total infrastructure costs to the various vehicle types



However, if we zoom in on the variable infrastructure costs, we see that HGVs and busses are responsible for the main part of these costs (see Figure 36). This can be explained by the fact that these costs are mainly caused by these



³⁰ If we assume that busses have 2.3 axles on average instead of 2 (see Section 4.2.4), the share of busses in total infrastructure costs decreases slightly to 13%, while the share of HGVs slightly increases to 21%.

vehicles' high axle loads³¹. The fixed infrastructure costs, on the other hand, are for a large share dependent on capacity (PCE kilometres) and traffic volumes (vehicle kilometres) and hence the major part of these costs are caused by passenger cars.



Figure 36 Allocation of total fixed and variable infrastructure costs to the various vehicle types

An overview of the estimated total infrastructure costs per kilometre road network is given in Figure 37. For most of the countries, these costs lie between € 15,000 and 30,000 per kilometre road network. For some countries (Estonia, Finland, Latvia, Lithuania and Sweden), the costs are considerably lower, which can be explained by the large share of unpaved and low density roads, for which both investment and O&M costs are very low. The low infrastructure costs in Bulgaria are (at least partly) explained by the incomplete expenditure data used as input for the infrastructure cost estimations (see Annex A).

There are also some countries (Austria, Germany, Croatia, The Netherlands, Portugal and Romania) for which the infrastructure costs per kilometre road network are significantly higher than € 30,000. For Croatia and Portugal, this may be explained by the large-scale and relatively expensive investment programmes applied over the last twenty years (see Section 3.3.2). Additionally, the replacement of war-damaged road infrastructure may contribute to the high infrastructure costs in Croatia, particularly since this infrastructure was not yet fully depreciated. For Romania, the rather large inefficiencies in the road construction and maintenance sector is considered the main cause of the relatively high infrastructure costs (again see Section 3.3.2). A relatively complex road network (large number of bridges and tunnels), significant winter maintenance and relatively high quality roads are the main reasons for the relatively high infrastructure costs in Austria. In Germany and the Netherlands, high traffic density combined with high quality levels, explain relatively high infrastructure costs. Furthermore, the high population density in the Netherlands often requires expensive solutions for enhancing or improving the road network.

³¹ On average busses have higher axle loads than HGVs and therefore a larger share of the variable infrastructure costs are allocated to busses. The higher average axle loads of busses compared to HGVs is mainly because we assumed that busses only have two axles, while HGVs may have 3, 4 or even 5 axles. In case we assume that busses have 2.3 axles on average (see Section 4.2.4), the share of busses in the variable infrastructure costs decreases to 36%, while the share of HGVs increase to 39%.

Figure 37 Infrastructure costs per kilometre road network length



4.4 Average infrastructure costs

4.4.1 Passenger transport

Figure 38 presents the average infrastructure costs (in \in per 1,000 passenger kilometres) for the three passenger modes considered in this study. In all countries, the average infrastructure costs for busses are estimated as the highest, followed by passenger cars and motorcycles³². The relatively high infrastructure costs for busses is due to the large share of variable (weight dependent) infrastructure costs caused by these vehicles³³.

³² If we assume an average number of axles of 2.3 (instead of 2) for busses (see Section 4.2.4), the average infrastructure cost for busses decrease by about 15-20%. Even in that case the average infrastructure costs for busses are estimated to be the highest in most European countries.

³³ The vehicle category 'bus' comprise both (public) busses and coaches. CE Delft and VU (2014) shows that because of their higher average mass (and hence axle loads), coaches have higher average infrastructure costs (in €/vehicle kilometres) than busses in the Netherlands. In terms of €/passenger kilometres, however, average infrastructure costs are higher for busses in the Netherlands, because of the lower occupancy rates of these vehicles.





In general, the highest average infrastructure costs for passenger modes are estimated for countries with high infrastructure costs per kilometre road network as well (i.e. Austria, Croatia, Netherlands, Portugal and Romania). The high average infrastructure costs in Hungary are probably explained by the relatively low traffic density on their road network³⁴. The high traffic density on the road network in Germany and the UK is an important reason for the lower average infrastructure costs in these countries.

4.4.2 Freight transport

Figure 39 shows the estimated average infrastructure costs (in \notin per 1,000 tonne kilometre) for HGVs. Again, average infrastructure costs are high in countries with high costs per kilometre road network (i.e. Austria, Croatia, the Netherlands and Romania). The average costs in Hungary and the Czech Republic are relatively high due to the low traffic density on the Hungarian and Czech network and the low estimated average loads of trucks in these countries.



³⁴ This implies that the total infrastructure costs are allocated to a relatively limited number of passenger kilometers, resulting in higher levels of average costs.

Figure 39 Average infrastructure costs HGVs in 2013



4.4.3 Vans

As vans are often used for services related transport (e.g. by plumbers), the average load of vans relatively low. Expressing the average infrastructure costs in \notin per 1,000 tonne kilometres would therefore result in very high and meaningless values. Therefore, we decided to express these costs in \notin per 1,000 vehicle kilometres.

The average infrastructure costs for vans are highest in Romania, Croatia, Hungary and Austria (see Figure 40), which is in line with the results for freight and passenger transport.



62

Figure 40 Estimation of average infrastructure costs vans in 2013



4.4.4 Passenger cars vs. HGVs

Figure $3\overline{7}$ compares average infrastructure costs of passenger cars and HGVs (both in $\epsilon/1,000$ vkm). In general, the average infrastructure costs for HGVs are considerably higher than for passenger cars (factor 4 to 8). This is explained by the relatively high axle loads of HGVs and hence the large share of weight dependent costs allocated to them.







5 Comparison of revenue and costs

5.1 Introduction

This chapter seeks to compare how infrastructure costs caused by road transport compare to the tax/charge revenue they bring in. Firstly, we discuss which perspective should be used to make this comparison (Section 5.2). Secondly, we make the comparison between infrastructure costs and tax/charge revenue by presenting and discussing infrastructure cost coverage ratios, i.e. the share of infrastructure costs covered by tax/charge revenue. We do this both for the road transport sector as a whole (Section 5.2) as for the various vehicle types separately (Section 5.3).

Research question 4

How do the total tax revenue compare to the total infrastructure costs in 2013 in the EU Member States?

This question consists of two sub questions:

- 1. From which perspective can tax revenue and infrastructure costs be compared?
- 2. What are the infrastructure cost coverage ratios in 2013 in the various EU countries?

In this chapter we combine the estimated tax/charge revenue figures estimated in Chapter 2 with the infrastructure cost figures estimated in Chapter 4, implying that we also combine the uncertainties in both types of figures. This causes relatively large uncertainties in the presented infrastructure cost coverage ratios. Particularly the ratios for separate vehicle categories (Section 5.4) are relatively uncertain. But in the end, all ratios have the right order of magnitude and general conclusions are believed to be reliable.

5.2 Different perspectives to compare tax revenue and infrastructure costs

As discussed by Fraunhofer-ISI & CE Delft (2008), there are three different perspectives according to which road tax revenue and infrastructure costs can be compared:

- Government budget perspective: this perspective provides the net contribution of the transport sector to the total state budget. Therefore, the total (non-earmarked and earmarked fixed and variable) revenue is compared with total (fixed and variable) costs. This perspective provides insight in whether or not the transport sector is a net contributor to the public sector (as are many other economic sectors).
- User pay perspective: this perspective shows whether transport consumers as a group pay for their use of transport infrastructure. In this perspective total (variable and fixed) earmarked revenue is compared to total (fixed and variable) costs. Non-earmarked revenues may be meant to cover external costs or to contribute to the overall funding of the state. Although from a theoretical perspective this approach clearly distinguishes between the contribution the road transport user makes to financing public services in their function as common tax payer (non-earmarked revenue) and the



compensation road users pay for the use of transport infrastructure (earmarked revenue), in reality this distinction is often not made. In several EU Member States with relatively high tax rates (e.g. the Netherlands, Austria) none of the transport tax revenue is specially earmarked for covering road infrastructure. Applying the user pay perspective for these countries can, therefore, result in confusing results. For that reason, we will not apply this perspective in this study.

Efficiency perspective: are changes in the external and infrastructure costs caused by road users covered by changes in taxes and charges?
 This requires that variable (earmarked and non-earmarked) revenue is compared to variable external costs. This perspective is not focussed on the way the revenue of the taxes and charges are used, but on the way user behaviour can be affected by the taxes/charges. As this perspective is not in line with the objective of this study (to compare total tax revenue and infrastructure costs of road transport), we will not apply it.

As explained above, we will apply the government budget perspective in this study to compare the total tax revenue and infrastructure costs of road transport.

5.3 Infrastructure cost coverage ratio road transport

From the government budget perspective all revenue from specific road transport taxes/charges are compared to total infrastructure costs. As explained in Section 2.2, only the VAT charged on fuel excise duties and registration taxes is considered a transport tax. Therefore, only that share of the VAT revenue is included in the calculation of the infrastructure cost coverage ratios³⁵.

Figure 42 shows the infrastructure cost coverage ratios for the road transport sector as a whole. In most countries the ratio is larger than 100%, indicating that the tax/charge revenue are higher than infrastructure costs. Exceptions are Croatia, Hungary and Romania. In these three countries infrastructure costs are relatively high, while at the same time the revenue from taxes and charges is moderate. The highest coverage ratios are estimated for Bulgaria, Malta and Luxembourg. The underestimation of the infrastructure costs (partly) explains the high coverage ratio estimated for Bulgaria. For Malta and Luxembourg relatively low infrastructure costs and high tax/charge revenue is the explanation for the high coverage ratio.



65

³⁵ Revenues from insurance taxes and company car taxation are not included as well, as they are not considered specific transport taxes (see Section 2.2). Parking charge revenue is not considered as well, because of poor data availability.

Figure 42 Estimated infrastructure cost coverage ratios road transport in 2013



5.4 Infrastructure cost coverage ratios per vehicle type

At EU level, the revenue from taxes and charges is significantly higher than the infrastructure costs for passenger cars, motorcycles, and vans (see Figure 43). For busses, the costs far exceed the revenue, which results from by the significant wear and tear caused by these vehicles³⁶. For HGVs, revenue and costs are in the same range (particularly given the uncertainty in the figures estimated). Compared to passenger cars HGVs pay relatively less taxes/charges while they are responsible for a much larger share of the infrastructure costs.



³⁶ As explained in Section 4.2.4, the average number of axles for busses in the EU28 is uncertain. As this indicator affects the infrastructure cost estimates, we have carried out a sensitivity analysis by assuming that busses have 2.3 instead of 2 axles on average. In that case the infrastructure cost coverage ratio of busses increases slightly (from 29% to 33%), while the ratio for HGVs slightly decreases (from 112% to 102%).





As is shown in Figure 44 the tax/charge revenue from passenger cars is significantly higher than the infrastructure costs in almost all EU countries. An exceptions is Romania, while the ratio in Croatia is just above 100%. Notice that most of the countries with high coverage ratios (Malta, Luxembourg, Greece and Denmark) have relatively high tax revenue per car (and passenger kilometre) as well. An exception is the UK, which' relatively high coverage ratio is mainly due to the relatively low infrastructure costs per passenger car. The relatively high Bulgarian infrastructure cost coverage ratio for passenger cars (and all other vehicle types) is again (partly) explained by the underestimation of infrastructure costs.







For motorcycles almost all countries have infrastructure cost coverage ratios exceeding 100%, with the exception of Estonia and Romania. The very large infrastructure cost coverage ratio for Denmark is caused by the high registration taxes levied on Danish motorcycles. Also in other countries with high coverage ratios (like Austria, Belgium and Luxembourg) relatively high tax levels are applied on motorcycles.



Figure 45 Infrastructure cost coverage ratios for motorcycles in 2013

The infrastructure costs of busses are not covered by revenue from taxes and charges. This is mainly due to the relatively high infrastructure costs caused by these vehicles.

Figure 46 Infrastructure cost coverage ratios for busses in 2013





Vans are less heavily taxed than passenger cars, resulting in lower infrastructure cost coverage ratios than for passenger cars (see Figure 47). There are several countries for which infrastructure cost coverage ratios below or near 100% are found (Austria, Czech Republic, Croatia, Hungary, Lithuania, Romania). Most of the countries with relatively high cost coverage ratios apply relatively high tax/charge rates for vans (e.g. Denmark, Finland, Greece and Ireland).





As is shown in Figure 48 there are many countries where the infrastructure costs of HGVs are not covered by the revenue from taxes and charges.

For thirteen countries, we find cost coverage ratios below or near 100%. In Hungary, the Netherlands and Romania we estimate infrastructure cost ratios of about 50%. Particularly for the Netherlands, this ratio is significantly lower than for other modes, which can be explained by the relatively low tax levels for HGVs in the Netherlands (compared to other modes). Another conclusion that can be drawn from Figure 48 is that in almost all countries that have implemented a (general or HGV-specific) distance-based road charging scheme the infrastructure costs are estimated to be covered by tax/charge revenue (the only exceptions are Austria, Croatia and Portugal, but these are countries with relatively high infrastructure cost levels).



69

Figure 48 Infrastructure cost coverage ratios for HGVs in 2013





6 Conclusions

6.1 Tax/charge revenue from road transport is higher than infrastructure costs

At EU level, the revenue of road transport taxes and charges exceeded the infrastructure costs in 2013. In only a few countries (Croatia, Hungary and Romania) are the infrastructure costs higher than tax/charge revenue.

Motorcycles and passenger cars have the highest infrastructure cost coverage ratios. Tax/charge revenue of passenger cars is estimated to exceed their infrastructure costs by about 110%, while the revenue coming from motorcycles is about 165% higher than their infrastructure costs. For the heavy duty vehicles (busses and HGVs), infrastructure cost coverage ratios are significantly lower. Busses have coverage ratios that are far below 100% in all EU countries. A large share of the infrastructure cost is attributed to these vehicles, while at the same time average tax/charge levels for busses are relatively low. The infrastructure costs and tax/charge revenue for HGVs are in the same range at the EU level. For several individual countries, however, infrastructure costs significantly exceed tax/charge revenue. In Hungary, the Netherlands and Romania we estimate infrastructure cost coverage ratios of only about 50%, meaning that they are not covering the impact that these vehicles have on the road infrastructure. Particularly for the Netherlands, this ratio is significantly lower than for other modes, which can be explained by the relatively low tax levels for HGVs in the Netherlands.

6.2 Significant contribution road transport to total tax revenue

Road transport significantly contributes to the tax revenue collected in European countries. On average, 5-10% of total tax revenue in EU countries comes from road transport taxes (registration taxes (incl. VAT), ownership taxes, fuel excise duties (incl. VAT)), i.e. \in 256 billion. As also the revenue from infrastructure charges (which technically are not tax revenue) is taken into account (about \in 30 billion), the total revenue from specific road transport related taxes and charges in 2013 in the EU is about \notin 286 billion. Additionally, about \notin 79 billion of revenue from VAT on vehicle and fuel purchases is collected in the EU.

The majority of this revenue is from fuel excise duties (almost half of the revenue). The contributions of the various tax types differ significantly between countries, though. For example, in Denmark vehicle taxes are responsible for almost half of the total road transport tax/charge revenue, while in France more than 20% of the revenue is coming from infrastructure charges.

6.3 Costs of EU road infrastructure are estimated at € 178 billion in 2013

The total road infrastructure costs in the EU are estimated at \in 178 billion in 2013. Slightly more than half of the costs are caused by passenger cars (about 54%), even though these vehicles are contributing the majority of the
kilometres on European roads. Also the heavy vehicle categories (HGVs and busses) contribute significantly to the infrastructure costs (about 21% and 15%, respectively).

The average infrastructure costs (in \notin /vkm) are significantly higher for HGVs compared to passenger cars. Because of the higher axle loads of HGVs, the wear and tear of these vehicles per kilometre driven is much higher than for passenger cars. As busses have relatively high axle loads as well, the average infrastructure costs for these vehicles are also relatively high.

6.4 Road infrastructure expenditures have been severely affected by economic crisis

In the period 1995-2008/2010 infrastructure spending in the EU shows an increasing trend, particularly in Central and Eastern European countries. In the latter countries, the rising need for a good quality infrastructure to support economic development combined with the access to large-scale EU funds, have resulted in high levels of investments and O&M expenditures in the first decade of this century. The share of GDP spent on road infrastructure also increased significantly in these countries, up to 2.6% in 2009. The growth in road infrastructure spending in Western European countries, on the other hand, was modest in the period 1995-2008 compared to Central and Eastern Europe. The share of these expenditures in GDP shows even a decrease over the years in these countries (from about 1.1% in 1995 to about 0.7% in 2013).

As a result of the economic crisis 2008-2013, road infrastructure spending decreased significantly in many European countries since 2007/2008. At the EU level, total annual infrastructure expenditures in 2013 were about \in 105 billion, which is well below the long-term average level of \in 130 billion.³⁷ Particularly countries severely affected by the crisis (Southern and Eastern European countries) show a sharp decrease in both investments and O&M expenditures. For example, investment levels in Portugal dropped by about 83% between 2008 and 2013.

6.5 Uncertainties

The quality and availability of data on (particularly) infrastructure expenditures varies significantly between EU countries. Furthermore, no coherent framework exists for accounting infrastructure expenditures, hindering the comparison of these data between countries. For that reason, comparisons between countries should be made very carefully. The data for individual countries and groups of countries is relatively consistent over time, however, such that the identification of trends in infrastructure spending is more reliable.

In general, figures on total revenue and costs are more reliable than figures per mode of transport, as the allocation of total figures to modes does create some additional uncertainties. Furthermore, total revenue and cost figures are more reliable than average ones (expressed in \notin /vkm, \notin /tkm or \notin /pkm), as the latter have to deal with relatively large uncertainties in traffic performance data.



³⁷ Based on total annual infrastructure expenditures in the period 1995-2013.

The results of this study do contain quite some uncertainties, which have to be kept in mind when interpreting them. Nevertheless, this study provides at least the right order of magnitudes for the various results and hence the general conclusions drawn in the previous sections are considered to be reliable.



7 References

ACEA, 2013. *ACEA Tax Guide 2013*, Brussels: European Automobile Manufacturers' Association (ACEA).

BSF, 2003. Schweitzerische Strassenrechnung Revision 2000, Neuchâtel: Bundesamt für Statistik.

CE Delft ; INFRAS ; Fraunhofer-ISI, 2011. *External costs of transport in Europe* : *Update Study for 2008*, Delft: CE Delft.

CE Delft; TML; TNO; TRT, 2012. An inventory of measures for internalizing external costs in transport, Delft: CE Delft.

CE Delft; VU, 2014. Externe en infrastructuurkosten van verkeer : Een overzicht voor Nederland in 2010, Delft: CE Delft.

CE Delft, 2008. *Infrastructuurkosten van het vrachtverkeer over de weg*, Delft: CE Delft.

CEDR, 2010. *Report on BEXPRAC*, Pais: Conference of European Directors of Roads (CEDR).

Department of Transport, Tourism and Sport, 2014. *Investing in our transport future - A strategic framework for investments in land transport*, Dublin: Department of Transport, Tourism and Sport.

Doll, C., 2005. Allokation gemeinsamer Kosten der Straßeninfrastruktur. Anwendbarkeit der Konzepte kooperativer Spiele in der Wegekostenrechnung (Diss.), Baden-Baden : Nomos.

DVS, 2001. *Basisonderhoudsniveau 2001- Hoofdwegennet*, Delft: Dienst Verkeer en Scheepvaart (DVS).

DVS, 2007. Basisonderhoudsniveau 2007 - Hoofdwegennet, Delft: Dienst Verkeer en Scheepvaart (DVS).

Ecorys ; CE Delft, 2006. Infrastructure expenditures and costs. Practical guidelines to calculate total infrastructure costs for five modes of transport, Rotterdam ; Delft: Ecorys ; CE Delft.

EEA, 2002. TERM 2002 19 EU : Investment in transport infrastructure per capita and by mode, Copenhagen: European Environment Agency.

Fraunhofer-ISI ; CE Delft, 2008. Internalisation Measures and Policies for all external cost of Transport (IMPACT) - Deliverable 2: Road infrastructure cost and revenue in Europe, Karlsruhe: Fraunhofer ISI.

HLG, 1999. Calculating transport infrastructure costs. Final report of the expert advisors to the high level group on infrastructure charging, Berlin/Zürich: High Level Group, working group 1 (HLG).

ICCT, 2014. *European Vehicle Market Statistics*, Berlin: The International Council on Clean Transportation.

IMF, 2015. IMF Country Report No. 15/80: Romania - Selected Issues, Washington D.C.: International Monitary Fund (IMF).

ITF, 2013a. Spending on transport infrastructure 1995-2011 : Trends, policies, data, Paris: International Transport Forum (ITF).



ITF, 2013b. Understanding the value of transport infrastructure : Guidelines for macro-level measurement of spending and assets, Paris: OECD, International Transport Forum (ITF).

ITS , 2001. Surface Transport Costs and Charges Great Britain 1998: final Report , London: ITS UK.

ITS, 2000. The accounts approach : Unite (Unification of accounts and marginal costs for transport efficiency) Deliverable 2, Leeds: University of Leeds.

NTC, 2005. *Third Heavy Vehicle Road Pricing Determination* : *Technical Report*, Melbourne: National Transport Commission (NTC).

Ove Arup & Partners, 2010. JASPERS CFPM Study : Country Reprt: Romania, West Midlands: Ove Partners & Partners Ltd.

Pereira, A. M. & Pereira, R. M., 2015. *Is all infrastructure investment created equal? : The case of Portugal*, Williamsburg: College of William and Mary, Department of Economics.

ProgTRans/IWW, 2007. Aktualisierung der Wegekostenrechnung für die Bundesfernstraßen in Deutschland : Endbericht, Basel/ Karlsruhe: ProgTrans AG ; IWW.

Steer Davies Gleave, 2014. EU Road surfaces : Economic and safety impact of the lack of regular road maintenance, study, Brussels: European Parliament.

WBG, 2003. Croatia : Country Economic Memorandum: A strategy for growth through European Integration, Washington D.C.: World Bank Group (WBG).

WBG, 2004. *Croatia and The World Bank* : *Briefing book*, Washington D.C.: World Bank Group (WBG).

WBG, 2011. Monitoring road works contracts and unit costs for enhanced governance in Europe and Central Asia, Washington D.C.: World Bank Group (WBG).

WBG, 2015. World Bank Group: Croatia Partnership : Country Progam Snapshot, Washington D.C.: World Bank Group (WBG).

WEF, 2014. *The global Competitveness report 2013/2014*, Geneva: World Economic Forum (WEC).



Annex A Data sources and data reliability

A.1 Introduction

In this Annex we describe the data sources used as well as the methodologies applied to estimate any missing data. Additionally, we briefly discuss the main uncertainties in the input data used.

In the remainder of this Annex we first briefly describe the approach followed to identify the level of uncertainty in the various input data used (Section A.2). Next, the data sources and data reliability with respect to taxes and charges (Section A.3), road infrastructure investments (Section A.4), road infrastructure O&M expenditures (Annex A.5), vehicle, passenger and tonne kilometres (Section A.6), road network length (Section A.7), and other input data (Section A.8) are discussed. Finally, we present some recommendations to increase the availability and quality of the input data used in this study (Section A.9).

A.2 Approach to data reliability assessment

In order to assess the reliability of the various input data in a systematic and coherent way, we have developed a qualitative assessment method identifying six categories of uncertainty, ranging from little uncertainty (dark green) to much uncertainty (red) (see Table 13). Based on an expert judgement of CE Delft, all input data used in this study has been assigned to one of these six categories. In this way a good overview of the uncertainties associated to the input data can be given.

Table 13 Reliability assessment methodology

| Uncertainty categories | Explanation of uncertainty |
|------------------------|--|
| | All data is coming from reliable sources. |
| | All data is coming from reliable sources, but other sources present data that do not fully match (and differences can only be partly explained). Most of the data is coming from reliable sources; only few missing data which has to be estimated based on reliable estimation approach. |
| | Data is partly coming from reliable sources and partly estimated based on reliable estimation approach. Various data sources present different figures (and differences can only be partly explained). |
| | Data is partly from reliable sources and partly estimated. Estimation approach used has significant uncertainties. Various data sources present different figures (and differences cannot be explained). |
| | Large data gaps (only small part of the data is available from reliable sources. Main part of the data has to be estimated. All data has optimated (o.g., baced on a peighbouring country) |



A.3 Taxes and charges

The total revenue from registration taxes, ownership taxes, tolls and vignettes are based on data from the following sources (see also Table 14):

- National data: it includes data from national statistical agencies, Ministries of Finance and/or Transport, road authorities, road charging scheme operators, and car dealer associations;
- Eurostat;
- OECD environmental tax database;
- ASECAP country reports: 2013 versions of these reports were consulted;
- (CE Delft; TML; TNO; TRT, 2012), An inventory of measures for internalising the external costs in transport;
- ACEA Tax Guide 2013.

Table 14 Sources used for the various countries to determine total tax/charge revenue

| Data source | Registration tax | Ownership tax | Tollsª | Vignettes |
|-----------------|------------------|-----------------|-----------------|-----------------|
| National data | AT, BE, DK, FI, | AT, BE, CZ, DK, | AT, DE, FR, PO, | AT, DK, FR, HU, |
| | GR, HR, HU, IT, | DE, FI, GR, HR, | RO, SE, UK | LT, LU, NL, RO, |
| | LV, MT, NL, PO, | HU, IT, LV, LT, | | SE |
| | SI, ES | MT, NL, PO, SK, | | |
| | | SI, ES, SE, UK | | |
| Eurostat | FR, IE, PT, RO, | | | |
| | SK | | | |
| OECD | | BE, FR, IE, LU | | BE, CZ |
| Environmental | | | | |
| tax database | | | | |
| ASECAP country | | | CZ, DK, GR, HR, | SK, SI |
| reports | | | IE, IT, NL, PO, | |
| | | | PT, SK, SI, ES | |
| CE Delft et al. | | BG, EE, RO | | BG |
| (2012) | | | | |
| ACEA Tax Guide | | РТ | | |
| 2013 | | | | |

^a For a few countries we have included the revenue from local toll schemes as well. These are:

- Denmark/Sweden: Store Baelt bridge and Oresund bridge;
- Netherlands: Westerscheldetunnel;
- Romania: Several toll bridges;
- Sweden: Urban road pricing schemes in Stockholm and Gothenburg;

UK: M6 and London congestion charging.

Only for a few countries data was available on the allocation of total revenue from these taxes and charges to vehicle types. For the other countries this allocation has been made by CE Delft, based on the methodologies described in Section 2.2.

The revenue from fuel excise duties and VAT is estimated using a bottom-up approach:

Fuel excise duties: total transport fuel sales (distinguishing between petrol, diesel and other fuels) in the various EU countries (from Eurostat) are allocated to the various modes. This allocation has been based on the share the various modes have in the country's total energy consumption of road transport. Next, total revenue is estimated by multiplying the fuel sales per mode of transport by the excise duty rates applied in 2013 (European Commission, 2013). Where relevant, corrections have been



made for excise duty refund schemes.

- VAT on fuel; the VAT per litre was estimated based on average (country specific) commodity prices (from the T&E fuel tax database) + excise duties (from the EU excise duty tables) and VAT rates (from the EU excise duty tables). Next, total revenue was estimated by using the data on total fuel sales per mode of transport in the various Member States (see above). For busses, vans and HGVs it was assumed that VAT on fuel can be recovered.
- VAT on new vehicles; the average VAT per newly sold vehicle was estimated based on average sales prices (from ICCT and ACEA) and VAT rates in the various Member States (ACEA, 2013). By multiplying these figures with the newly sold vehicles in 2013 (from EEA, ACEA) the total VAT revenue are estimated. For busses, vans and HGVs it was assumed that VAT on vehicle purchases can be recovered.

Where possible the results of these bottom-up approaches have been crosschecked with national data.

In the assessment of the reliability of the data on tax/charge revenue a distinction has been made between the data on total revenue (Table 15) and the allocation of these total revenue to the various modes (Table 16).

Table 15 shows that the data on total revenue from registration taxes, ownership taxes and tolls and vignettes is rather reliable. For France no data was available on the revenue of the malus for fuel-inefficient vehicles in the ownership tax, which results in some uncertainty. The uncertainty on revenue from fuel excise duties and VAT on fuel is larger, as these data is estimated using a bottom-up approach. However, crosschecks with national data and data from the EU excise duty tables show that the results are reliable for many countries (indicated by light green). For some other countries such crosschecks did show some (limited) deviations between data sources, resulting in some larger uncertainties (indicated by yellow). Finally, only for a few countries the bottom-up estimates of VAT revenue on vehicle sales have been crosschecked with national data (indicated by light green). For the other countries no data was available to do such crosschecks, resulting in a significant amount of uncertainty.

| Country | Registration tax | Ownership tax | Tolls and vignettes | Fuel excise duty | VAT on fuel | VAT on new vehicles | Overall reliability |
|---------|---------------------|------------------|------------------------|------------------------|-------------------|---------------------------|------------------------|
| AT | | | | duty | Tuet | venicies | |
| BE | | | | | | | |
| BG | | | | | | | |
| CZ | | | | | | | |
| DK | | | | | | | |
| DE | | | | | | | |
| EE | | | | | | | |
| FI | | | | | | | |
| FR | | | | | | | |
| GR | | | | | | | |
| HR | | | | | | | |
| HU | | | | | | | |
| IE | | | | | | | |
| IT | | | | | | | |
| LV | | | | | | | |

 Table 15
 Assessment of reliability of total revenue from taxes and charges



| Country | Registration tax | Ownership tax | Tolls and vignettes | Fuel excise duty | VAT on fuel | VAT on new vehicles | Overall reliability |
|---------|---------------------|------------------|------------------------|------------------------|-------------------|---------------------------|------------------------|
| LT | | | | | | | |
| LU | | | | | | | |
| MT | | | | | | | |
| NL | | | | | | | |
| PO | | | | | | | |
| PT | | | | | | | |
| RO | | | | | | | |
| SK | | | | | | | |
| SI | | | | | | | |
| ES | | | | | | | |
| SE | | | | | | | |
| UK | | | | | | | |

The allocation of total tax/charge revenue to the various modes is more uncertain than the estimation of total tax/charge revenue data, as is shown by Table 16. Only for a few countries (AT, BE, DK, DE, LU, MT and NL) reliable data was available on this allocation for all (or most) taxes/charges. For the other countries estimations have been made, resulting in varying levels of uncertainty.

| Table 16 | Assessment of reliability | of allocation of | f tax/charge revenue t | to modes |
|----------|---------------------------|------------------|------------------------|----------|
|----------|---------------------------|------------------|------------------------|----------|

| Country | Registration tax | Ownership tax | Tolls and vignettes | Fuel excise | VAT on | VAT on new | Overall reliability |
|---------|---------------------|------------------|------------------------|----------------|-----------|---------------|------------------------|
| | | | | duty | fuel | vehicles | |
| AT | | | | | | | |
| BE | | | | | | | |
| BG | | | | | | | |
| CZ | | | | | | | |
| DK | | | | | | | |
| DE | | | | | | | |
| EE | | | | | | | |
| FI | | | | | | | |
| FR | | | | | | | |
| GR | | | | | | | |
| HR | | | | | | | |
| HU | | | | | | | |
| IE | | | | | | | |
| IT | | | | | | | |
| LV | | | | | | | |
| LT | | | | | | | |
| LU | | | | | | | |
| MT | | | | | | | |
| NL | | | | | | | |
| PO | | | | | | | |
| PT | | | | | | | |
| RO | | | | | | | |
| SK | | | | | | | |
| SI | | | | | | | |
| ES | | | | | | | |
| SE | | | | | | | |
| UK | | | | | | | |



A.4 Investments

For the case study countries data on investments in road infrastructure was (partly) based on national data (AT, CZ, DK, DE, FR, IT, NL, PO, ES, SE, UK). Where necessary (and possible) this data was supplemented with data from the International Transport Forum. This source was also used to gather investment data for the other countries. All missing data was estimated by CE Delft, using different types of estimation methods.

As is shown in Table 17, six groups of countries can be distinguished based on the reliability of the investment data available for them:

- For four countries (almost) complete sets of reliable investment data were available (AT, DE, NL and UK).
- For eight countries (BE, DK, FI, FR, IE, IT, LU and SE) data on total investments was available, but a further breakdown (e.g. to enhancement and renewal expenditures) was missing. These data was estimated based on the data for the first group of four countries.
- For another eight countries (CZ, GR, HR, PT, PO, SK, SI and ES) also some data on total investments was missing (e.g. data for urban roads or data for some years). These data was estimated based on (linear) extrapolation or based on cost figures derived from comparable countries. For example, to estimate the investments in urban roads in Czech Republic, the investments per kilometre urban road for Poland were used as a proxy for the Czech Republic. By multiplying this proxy with the total number of kilometres of urban roads in Czech Republic the total investments in Czech urban roads were estimated (taking differences in price level between both countries into account).
- For three countries (HU, LV, LT) a significant part of the data on total investments was missing. These data was estimated based on growth trends in investments in comparable countries. Additionally, investment levels per kilometre urban road network in comparable countries were used to estimate missing data on urban roads investments.
- For Bulgaria and Malta only a small part of the data was available and all other data had to be estimated based on growth trends in comparable countries (for Bulgaria: Poland and Romania; for Malta: Spain).
- Finally, for Estonia no data on investments was available; the data has been estimated based on cost figures from Latvia (expressed in €/km road).

| Country | Reliability | Country | Reliability | Country | Reliability | Country | Reliability |
|---------|-------------|---------|-------------|---------|-------------|---------|-------------|
| | data | | data | | data | | data |
| AT | | FI | | LV | | RO | |
| BE | | FR | | LT | | SK | |
| BG | | GR | | LU | | SI | |
| CZ | | HR | | MT | | ES | |
| DK | | HU | | NL | | SE | |
| DE | | IE | | PO | | UK | |
| EE | | IT | | PT | | | |

Table 17 Assessment of reliability of data on investments



A.5 Operational and maintenance expenditures

The same data sources as for investments have been used to gather data on O&M expenditures. For the case study countries (AT, CZ, DK, DE, IT, NL, PO, ES, SE and UK) national data was used, where necessary supplemented with data from the International Transport Forum. This source was used as main source for O&M expenditures in all other countries as well. All missing data was estimated by CE Delft, using different types of estimation methods.

As is shown in Table 18, six groups of countries can be distinguished based on the reliability of the O&M expenditure data available for them:

- For four countries (almost) complete sets of reliable data on O&M expenditures were available (AT, DE, NL and UK).
- For five other countries (DK, FR, PO, ES and SE) data on total O&M expenditures were available, but without a distinction between operational and maintenance expenditures. This distinction has been made based on the data available from the first group countries.
- For seven other countries (BE, CZ, FI, LV, LT, LU, PT, and SI) an (almost) complete set of data on maintenance expenditures was available, but data on operational expenditures was missing. The latter data have been estimated based on the share operational expenditures have in total O&M expenditures in DE, NL and the UK (for Western European countries) and in Poland (for Central and Eastern European countries). For HU, data on O&M expenditures and operation expenditures was available, but data on O&M expenditures on urban roads was missing.

These have been estimated by using Polish expenditure figures per kilometre urban road network (corrected for differences in price level).

- For two countries (HR and SK) not only data on operational expenditures was missing, but also data on O&M expenditures on urban roads. Based on the approaches described above, the missing data for these countries was estimated. For IT, data on O&M expenditures after 2010 was missing; data for 2010 and before was available from ITF statistics. Some data on O&M expenditures in Italy are also presented in the National accounts of the infrastructure and Transport Sector (CNIT), but according to the expert conducting the case study for Italy, these data was incomplete. For that reason data from ITF was preferred.
- This data was estimated by using growth rates from Spain. Finally, for Ireland data on total maintenance expenditures was available, but seems to contain a significant trend break (i.e. in the period 2001-2009 the O&M expenditure figures were far below the long term average figures). We have contacted the International Transport Forum (source of the Irish data), but they couldn't explain these findings. For that reason, the Irish maintenance expenditure data is considered to be relatively uncertain (particularly for the period 2001-2009).
- For three countries (BG, MT and RO) only a small part of the data was available and all other data had to be estimated based on growth trends and unit values (€/km road) from comparable countries (for Bulgaria and Romania: Poland; for Malta: Spain).
- Finally, for two countries (EE and GR) (almost) no data on O&M expenditures was available. For Estonia, the data has been estimated based on cost figures from Latvia (expressed in €/km road). For Greece, data on O&M expenditures in 1995 were extrapolated based on Spanish growth rates.



| Country | Reliability | Country | Reliability | Country | Reliability | Country | Reliability |
|---------|-------------|---------|-------------|---------|-------------|---------|-------------|
| | data | | data | | data | | data |
| AT | | FI | | LV | | RO | |
| BE | | FR | | LT | | SK | |
| BG | | GR | | LU | | SI | |
| CZ | | HR | | мт | | ES | |
| DK | | HU | | NL | | SE | |
| DE | | IE | | PO | | UK | |
| EE | | IT | | PT | | | |

A.6 Vehicle, passenger and tonne kilometres

In this study we make use of data on vehicle, passenger and tonne kilometres on national territories, i.e. all kilometres made by national and foreign vehicles on the roads within a specific country. However, in most statistical databases the vehicle, passenger and tonne kilometres for vehicles registered in a country are defined. Data with a national territory scope is only scarcely available and as a consequence the uncertainty in these data is rather large.

For vehicle kilometres the following data sources were used (see also Table 19):

- National statistics, mainly from national statistical agencies;
- Eurostat data; fragmented data is available on vehicle kilometres for some EU countries;
- Data from CE Delft; INFRAS; Fraunhofer-ISI, (2011), External costs of transport in Europe; in this study a complete set of vehicle kilometre data is available, but only for 2008;
- TREMOVE; an EU wide transport model providing a complete estimated set of vehicle kilometre data for all modes and all EU countries.

Passenger and tonne kilometres are estimated by multiplying the vehicle kilometres with an average occupancy rate and average load respectively (both differentiated to mode and country). By applying this bottom-up approach a consistent set of vehicle, passenger and tonne kilometres is estimated. The average occupancy rates were preferably based on national data or data from Eurostat. If these data was not available (which was often the case for motorcycles and busses), figures from TREMOVE were used. The same approach has been applied for tonne kilometres.

| Data source | Car | мс | Bus | Van | HGV |
|-------------|---------------|----------------|----------------|---------------|--------------|
| National | AT, BE, DK, | BE, DK, DE, | BE, DK, DE, | BE, DK, DE, | BE, DK, DE, |
| data | DE, FI, FR, | FR, IE, NL, | FI, FR, IE, | FI, FR, IE, | FI, FR, IE, |
| | IE, NL, PO, | PO, ES*, UK | NL, PO, ES*, | NL, PO, ES*, | NL, PO, ES*, |
| | ES*, UK | | UK | UK | UK |
| Eurostat | CZ*, EE*, | AT, CZ*, HU*, | AT, BG, CZ*, | AT, EE*, LV*, | AT, LV*, SE* |
| | HU*, LV*, | LV*, LT*, | EE*, HU*, | LT*, MT*, SE* | |
| | LT*, MT*, | MT*, RO*, SI*, | LT*, MT*, | | |
| | RO*, SI*, SE* | SE* | RO*, SK*, SI*, | | |
| | | | SE* | | |

Table 19 Sources used for the various countries to estimate vehicle kilometers



| Data source | Car | мс | Bus | Van | HGV |
|-------------|----------------|----------------|----------------|----------------|----------------|
| CE Delft et | BG*, GR*, | BG*, EE*, | GR*, HR*, | BG*, CZ*, | BG*, CZ*, |
| al. (2011) | HR*, IT*, LU*, | GR*, HR*, | IT*, LV*, LU*, | GR*, HU*, | EE*, GR*, |
| | PT*, SK* | IT*, LU*, PT*, | PT* | IT*, LU*, PT*, | HU*, IT*, LT*, |
| | | SK* | | RO*, SK*, SI* | LU*, PT*, |
| | | | | | RO*, SK*, SI* |
| TREMOVE | | FI | | HR | HR, MT |

* For these countries the data source provides only figures for years before 2013 or for part of the network. Therefore, additional estimations have been carried out by CE Delft to estimate the vehicle kilometres in 2013 for the entire network (e.g. by applying growth rates from TREMOVE).

Table 20 shows the results of the reliability assessment of the vehicle, passenger and tonne kilometre data. These results show the relatively large uncertainty in these data, reflecting the limited amount of data available in official statistics. The following groups of countries can be distinguished based on the reliability of the data available for them:

- For nine countries (AT, BE, DK, DE, FI, FR, NL, PO and UK) relatively reliable data on vehicle, passenger and tonne kilometre data is available.
- For one country (IE) reliable data on vehicle kilometres is available, but data on passenger and tonne kilometres is less reliable (occupancy rates for busses and motorcycles and average loads for HGVs were estimated, only occupancy rate figures for bus have been crosschecked).
- For three countries (LV, LT, SE) data on vehicle kilometres have been available from reliable sources (mainly Eurostat) for relatively recent years, but not for 2013. Therefore these data have been extrapolated by CE Delft, resulting in a higher level of uncertainty. The average occupancy rate and load used to estimate passenger and tonne kilometres have been crosschecked.
- For twelve countries (BG, CZ, EE, GR, HU, IT, LU, MT, PT, RO, SK and SI) only relatively old data on vehicle kilometres have been available (mainly from CE Delft et al, 2011), which have been extrapolated by CE Delft. This results in significant levels of uncertainty. Average occupancy rates and loads used to estimate passenger and tonne kilometres have only be roughly crosschecked.
- For two countries (HR, ES) a significant part of the data was missing.
 For HR data on vehicle kilometres of vans and HGVs was missing and hence had to be based on TREMOVE data. For Spain, data on vehicle kilometres on urban roads was missing; it was estimated by CE Delft based on data from Fraunhofer-ISI and CE Delft (2008).

| Country | vkm | pkm | tkm | Country | vkm | pkm | tkm |
|---------|-----|-----|-----|---------|-----|-----|-----|
| AT | | | | LV | | | |
| BE | | | | LT | | | |
| BG | | | | LU | | | |
| CZ | | | | MT | | | |
| DK | | | | NL | | | |
| DE | | | | PO | | | |
| EE | | | | PT | | | |
| FI | | | | RO | | | |
| FR | | | | SK | | | |
| GR | | | | SI | | | |
| HR | | | | ES | | | |

Table 20 Assessment of reliability of data on vehicle, passenger and tonne kilometres



| Country | vkm | pkm | tkm | Country | vkm | pkm | tkm |
|---------|-----|-----|-----|---------|-----|-----|-----|
| HU | | | | SE | | | |
| IE | | | | UK | | | |
| IT | | | | | | | |

A.7 Road network length

In order to compose an accurate set of data on road network length in the various countries, we studied two sources: Eurostat and the 'EU Transport in figures - Statistical pocketbook'. Deviations between these two sources have been studied and explained (for most deviations reasonable explanations have been found, e.g. urban roads or unpaved roads were not covered by one of the sources), and a best estimate of the road network length was made. Where possible these estimates were crosschecked with data from national statistical agencies.

As is shown in Table 21, for most countries complete and reliable data on road network length was available. For some countries some of the data was missing (e.g. on urban roads) and had to be estimated by combining both sources or by using other sources. For example, for Spain data on urban roads was missing in the data from Eurostat and the EU pocketbook and has been estimated based on Fraunhofer-ISI and CE Delft (2008).

Table 21 Assessment of reliability of data on road network length

| Country | Reliability | Country | Reliability | Country | Reliability | Country | Reliability |
|---------|-------------|---------|-------------|---------|-------------|---------|-------------|
| | uala | | uala | | uala | | uala |
| AT | | FI | | LV | | RO | |
| BE | | FR | | LT | | SK | |
| BG | | GR | | LU | | SI | |
| CZ | | HR | | MT | | ES | |
| DK | | HU | | NL | | SE | |
| DE | | IE | | PO | | UK | |
| EE | | IT | | PT | | | |

A.8 Other data

An overview of the remaining input data and the sources from which they are derived is given in Table 22.

Table 22 Overview of data sources

| Data | Source |
|--|--|
| Exchange rates | Eurostat |
| GDP and GDP/capita | Eurostat |
| Price Indexes | Eurostat, WBG |
| PPP figures | Eurostat |
| Discount rate | Fraunhofer-ISI and CE Delft (2008) |
| Vehicle fleet | Eurostat |
| Vehicle characteristics (PCE, axle load, mass) | Eurostat, Fraunhofer-ISI and CE Delft (2008) |



A.9 Recommendations to improve data availability and quality

Decreasing the uncertainties in the input data is key to improve the reliability of the results of assessments on road infrastructure costs/expenditures and tax/charge revenue. Particularly the following issues require additional research:

- The development and implementation of a coherent accounting framework for infrastructure expenditures in all European countries would increase the comparability of the expenditure data between countries.
- At the same time big efforts are needed to reduce the number of data gaps with respect to infrastructure spending in the EU. Particularly data on expenditures on urban roads is missing in many countries. Furthermore, detailed data on the breakdown of investments (to enhancement and renewal expenditures) is currently only available for a few countries. Finally, there is a lack of reliable data on operation expenditures on all road types in a significant number of EU countries.
- In many countries it is unknown which share the various vehicle types have in total tax/charge revenue. Improving data on this issue would reduce uncertainties significantly.
- Data on VAT revenue on vehicle sales is not available for most countries.
 As this data is difficult to estimate, improving data on this issue would significantly improve the reliability of the results.
- It is important to improve the data on traffic performance (vehicle kilometres, passenger kilometres and tonne kilometres) on national territories. These data is currently missing in statistical databases (e.g. Eurostat) for many countries and the data that are provided are fragmented and of low quality.



Annex B Profiles case study countries

B.1 Introduction

In this annex, we present the eleven case study countries as well as some of the main results of the assessments we made for these countries. For each of the countries the development in infrastructure spending over the period 1995-2013 is briefly discussed. Graphical representations of these developments can be found in Annex D.

B.2 Austria

| Country analys | is: Austria | |
|-------------------------|--|---|
| National | HERRY | |
| expert | | |
| Investments | | |
| Available | Description of data | Data sources |
| data on | - Total investments on all roads for | – Statistik Austria |
| expenditures | the period 2000-2013 Total investments on motorways for the period 1979-2013: For the period 1997-2013: breakdown of investments to enhancement and renewal expenditures Total investments on other main roads (Landesstraßen B) and in other roads (Landesstraßen and Gemeindestraßen) for the periods 1979-2000 and 2002-2006 For the period 2002-2006: breakdown of investments to enhancement and renewal expenditures | 1979-1997: BMWA 1998-2013: ASFINAG 1979-2000: BMWA 2002-2006: Administrations of the 9 Federal Countries |
| | other main roads | |
| Missing | Description of data | Estimation approach |
| data on expenditures | Total investments on all roads for the period 1979-1999 Breakdown of total investments on all roads to enhancement and renewal expenditures | Based on available data for motorways, other main roads and other roads For motorways: for the period 1979-1996 breakdown based on average shares enhancement and renewal expenditures in the period 1998-2013. For main other roads and other roads: breakdown based on average shares enhancement and renewal expenditures have in |
| | | total investments in main other roads in the period 2002-2006 |



| Country analys | is: Austria | | | | | |
|-----------------------------|--|-----------------------|--------------------------------------|--------------------|--|--|
| Key results | Total investment | Share of total | Total | Total | | |
| (in mln € ₂₀₁₃ , | expenditures in | investment | investment | investment | | |
| PPP adjusted) | 2013 | expenditures in | costs in 2013 | costs per km | | |
| | | GDP in 2013 | | road in 2013 | | |
| | 1,648 | 0.6% | 4,377 | 0.035 | | |
| Development | The investments in A | ustrian roads show a | a significant increas | e (in real terms) | | |
| over time in | between 1999 and 20 | 002 (about 60%), whi | ich can (at least) pa | rtly explained by | | |
| the period | increased investment | s in the main road r | network by ASFINAG | . Between 1997 | | |
| 1995-2013 | and 2008 the investments done by ASFINAG more than quadrupled | | | | | |
| | (Steer Davies Gleave | , 2014). In the perio | d 2002-2009 investr | nents levels in | | |
| | Austria were relative | ly stable, but a seri | ous drop took place | in 2010. Due to | | |
| | the economic crisis in | nvestment budgets v | were cut (e.g. ASFIN | IAG diminished its | | |
| | investments by appro | pximately 25% in 201 | 0 and 2011). | | | |
| Operating and | maintenance costs/ex | penditures | - | | | |
| Available | Description of data | | Data sources | | | |
| data on | Total O&M exper | ditures on all | Statistik Austr | ia | | |
| expenditures | roads for the per | iod 2002-2013 | | | | |
| | Iotal O&M exper | iditures on | - !998-2013: AS | FINAG | | |
| | motorways for th | ie period 1995- | 1995-1998: BN | IWA | | |
| | ZUI3 | a pariad ap | | | | |
| | For the entire | e perioù an | | | | |
| | maintenance | expenditures is | | | | |
| | available | experial cares is | | | | |
| | Total O&M exper | ditures on main | – 1995-2000: B/ | AWM | | |
| | roads (Landesstra | aßen B) for the | 2002-2006: A | dministrations of | | |
| | period 1995-2000 |) and 2000-2006 | the 9 Federal | Countries | | |
| | – Total O&M exper | ditures on other | – Statistik Austr | ia | | |
| | roads ((Landessti | raßen and | | | | |
| | Gemeindestraßer | n) for the period | | | | |
| | 2000-2006 | | | | | |
| Missing | Description of data | | Estimation approc | ıch | | |
| data on | Total O&M exper | nditures on all | Based on O&M | expenditures on | | |
| expenditures | roads for the per | iod 1995-2001 | motorways, of | ther main roads | | |
| | | | and other roads | | | |
| | Total O&M exper | nditures on main | Based on linea | ar interpolation | | |
| | roads in 2001 | | | | | |
| | Iotal O&M exper | ditures on other | Based on the s | same trend in | | |
| | roads for the per | 100 1995-2000 | expenditures | as for other main | | |
| | Broakdown of to | | For motorway | s based on | | |
| | - Dieakuowii oi toi | operational and | - Tor motor way available data | for other roads | | |
| | maintenance exp | enditures | based on aver | age shares in | | |
| | manifemanee exp | | Germany. The | Netherlands and | | |
| | | | the UK | nether tands and | | |
| Key results (in | Total O&M | O&M | Share of total | Share of O&M | | |
| mln € ₂₀₁₃ , PPP | expenditures/costs | expenditures per | O&M | expenditures in | | |
| adjusted) | in 2013 | km road in 2013 | expenditures in | total infra | | |
| | | | GDP in 2013 | expenditures in | | |
| | 2013 | | | | | |
| | 2,113 | 0.017 | 0.8% | 56% | | |
| Development | The O&M expenditure | es the main Austrian | road network have | shown a | | |
| over time in | gradually increasing | trend in the first de | cade of this century | , although | | |
| the period | expenditure levels ha | ave been significant | y reduced in 2010 a | nd 2011 due to | | |
| 1995-2013 | the economic crisis. On the other hand, the O&M expenditures on regional | | | | | |

| Country analysis: Austria | | | | | | |
|-----------------------------|---|----------------------|--|--------------------|--|--|
| | and local roads have | decreased significar | ntly over the last 20 |) years, such that | | |
| | the overall O&M exp | enditures show a dee | creasing trend since | e 1998. | | |
| Revenue from t | axes and charges in 2 | 013 | | | | |
| Available data | Description of data | | Data sources | Data sources | | |
| | Total revenue regi | istration tax | Austrian Minis | try of Finance | | |
| | Total revenue ownership tax | | Austrian Ministry of Finance | | | |
| | Total revenue tolls (MAUT) | | – ASFINAG | | | |
| | Total revenue vigr | nettes | – ASFINAG | | | |
| | Total revenue fue | l excise duty | Bottom-up analysis CE Delft | | | |
| | Total revenue VAT | on fuel and | Bottom-up and | alysis CE Delft | | |
| | vehicle sales | | | | | |
| Key results (in | Total revenue | Total revenue | Total revenue | Total VAT | | |
| bln € ₂₀₁₃ , PPP | vehicle taxes | infrastructure | fuel excise duty | revenue | | |
| adjusted) | | charges | | | | |
| | 2.0 | 1.5 | 3.4 | 2.7 | | |

B.3 Czech Republic

| Country analys | Country analysis: Czech Republic | | | | | |
|-----------------|---------------------------------------|-----------------------|------------------------------------|--------------------|--|--|
| National | Herbert Seelmann (Br | rno University of Te | chnology) | | | |
| expert | | | | | | |
| Investments | | | | | | |
| Available | Description of data | | Data sources | | | |
| data on | Total investment | s on all non-urban | Transport Yea | arbook Czech | | |
| expenditures | roads for the per | iod 1990-2013 | Republic, Min | istry of Transport | | |
| | Total investment | s on motorways | Directorate o | f Roads and | | |
| | for the period 19 | 90-2013 | Motorways (R | SD) | | |
| Missing | Description of data | | Estimation approc | ach | | |
| data on | Total investments | s on all roads for | For the period | 1990-2013 based | | |
| expenditures | the period 1979-2 | 2013 | on actual/esti | mated | | |
| | | | investments o | n motorways, | | |
| | other roads and urban road | | | nd urban roads. | | |
| | | | For the period | I 1979-1989 | | |
| | | | estimated bas | ed on the same | | |
| | | | trend as for in | ivestments in | | |
| | - | | Poland. | | | |
| | Total investments | s on all urban | - Based on unit investments (in | | | |
| | roads for the peri | 100 1990-2013 | €/KM urban road) from Poland. | | | |
| | Brockdown of tot | al invoctments on | - Dased on share | and renewal | | |
| | - Dieakuowii oi tot | ac investments on | expenditures in total | | | |
| | renewal expendit | | investments in Poland | | | |
| Key results (in | Total investment | Share of total | Total | Total | | |
| mln €2013. PPP | expenditures in | investment | investment | investment | | |
| adjusted) | 2013 | expenditures in | costs in 2013 | costs per km | | |
| . , | | , GDP in 2013 | | , road in 2013 | | |
| | 1,278 | 0.6% | 3,317 | 0.025 | | |
| Development | Accessing the Europe | an Union in 2004 ha | s boosted the road | infrastructure | | |
| over time in | investments in the Cz | ech Republic: inves | tments almost doub | oled between | | |
| the period | 2003 and 2005, which | n is mainly explained | d by the large contr | ibution of EU | | |
| 1995-2013 | funding. Since 2011 ir | nvestments have fal | len, which is at leas | st partly | | |
| | explained by the ecor | nomic crisis. | | | | |
| Operating and | maintenance costs/exp | penditures | | | | |
| Available | Description of data | | Data sources | | | |

| Country analysi | s: Czech Republic | | | |
|-----------------------------|--|---|---|---|
| data on expenditures | Total maintenan- for the period 19 Total O&M exper motorways for th 2003-2013 | ce expenditures 195-2013 nditures on ne period | Transport Year Czech Republic Directorate of Motorways (RSI | book : Roads and D) |
| Missing | Description of data | | Estimation approc | ach |
| data on expenditures | Total O&M expenditures on all roads for the period 1995-2013 Total operational expenditures on all roads for the period 1995-2013 | | Based on total and total oper expenditures of the period 199 Based on ratio expenditures of period 1995-21 | l maintenance ration on all roads for 95-2013 o maintenance operation in Poland in the 013 |
| Key results (in | Total O&M | O&M | Share of total | Share of O&M |
| mln € ₂₀₁₃ , PPP | expenditures/costs | expenditures per | O&M | expenditures in |
| adjusted) | in 2013 | km road in 2013 | expenditures in GDP in 2013 | total infra expenditures in 2012 |
| | 000 | 0.008 | 0.7% | 2013 |
| Development | O&M expenditures or | Czech roads show t | he same pattern ov | ver time as |
| over time in | investments: a sharp | increase around 200 |)4 when the Czech I | Republic joined |
| the period | the EU (and got acce | ss to EU funding) an | d a drop around 201 | 10 (probably |
| 1995-2013 | explained by the eco | nomic crisis). | · | |
| Revenue from t | axes and charges in 2 | 013 | | |
| Available data | Description of data | | Data sources | |
| | Total revenue owr | ership tax | Czech Ministry of Finance | |
| | Total revenue toll | s (MAUT) | – ASECAP | |
| | Total revenue vigr | nettes | OECD Environmental tax | |
| | | | database | |
| | Total revenue fuel | excise duty | - Bottom-up analysis CE Delft | |
| | Total revenue VAT vehicle sales | on fuel and | Bottom-up and | alysis CE Delft |
| Key results (in | Total revenue | Total revenue | Total revenue | Total VAT |
| bln € ₂₀₁₃ , PPP | vehicle taxes | infrastructure | fuel excise duty | revenue |
| adjusted) | | charges | | |
| | 0.3 | 0.7 | 4.3 | 2.3 |

B.4 Denmark

| Country analys | Country analysis: Denmark | | | | |
|----------------|--|--|--|--|--|
| National | Nilsson Production | | | | |
| expert | | | | | |
| Investments | | | | | |
| Available | Description of data | Data sources | | | |
| data on | Total investments on all roads for | Danish Road Directorate | | | |
| expenditures | the period 1979-2013 | | | | |
| | Total investments on urban roads | | | | |
| | for the period 1979-2013 | | | | |
| | Total investments on non-urban | | | | |
| | roads for the period 1979-2013 | | | | |
| Missing | Description of data | Estimation approach | | | |
| data on | Breakdown of total investments on | Breakdown based on the average | | | |



| Country analysi | is: Denmark | | | | |
|-----------------------------|---|-----------------------|--|----------------------|--|
| expenditures | all roads to enhance | cement and | shares enhancement and renewal | | |
| | renewal expenditu | res | expenditures have in Austria and | | |
| | | | the Netherlands | | |
| Key results (in | Total investment | Share of total | Total | Total investment | |
| mln € ₂₀₁₃ , PPP | expenditures in | investment | investment | costs per km road | |
| adjusted) | 2013 | expenditures in | costs in 2013 | in 2013 | |
| | | GDP in 2013 | | | |
| | 746 | 0.4% | 1,937 0.026 | | |
| Development | Investment levels on | Danish roads are rat | ther stable over the | years (both in | |
| over time in | absolute terms and as share of GDP), with the exception of the period 2004- | | | | |
| the period | 2008; in these years investment levels were considerably higher (up to 60%) | | | | |
| 1995-2013 | than the long-term a | verage. | | | |
| Operating and I | maintenance costs/exp | penditures | | | |
| Available | Description of data | | Data sources | | |
| data on | Total O&M expend | itures on all roads | – Danish Road Di | rectorate | |
| expenditures | for the period 199 | 5-2013 | | | |
| | Total O&M expend | itures on urban | | | |
| | roads for the perio | d 1995-2013 | | | |
| | Total O&M expend | itures on | | | |
| | non-urban roads for the period | | | | |
| | 1995-2013 | | | | |
| Missing | Description of data | | Estimation approc | ach | |
| data on | Breakdown of tota | I O&M | Based on average shares in | | |
| expenditures | expenditures on al | l roads to | Germany, the Netherlands and | | |
| | operation and maintenance | | the UK | | |
| | expenditures | | | | |
| Kev results (in | Total O&M | O&M | Share of total | Share of O&M | |
| mln € ₂₀₁₃ , PPP | expenditures/costs | expenditures per | O&M | expenditures in | |
| adjusted) | in 2013 | km road in 2013 | expenditures in | r total infra | |
| | | | GDP in 2013 | , expenditures in | |
| | | | | , 2013 | |
| | 656 | 0.009 | 0.3% | 47% | |
| Development | O&M expenditures gr | adually decreased ir | n Denmark (both in | absolute terms and | |
| over time in | as share of GDP). Bet | ween 1995 and 2013 | B the total O&M exp | enditures on | |
| the period | Danish roads was red | uced by almost 50%. | However, O&M exp | penditure levels per | |
| 1995-2013 | kilometre road netwo | ork are still among t | he highest in Europe | e and are | |
| | comparable by expen | diture levels in Ger | many, the Netherla | nds and the UK. | |
| Revenue from t | axes and charges in 2 | 013 | | | |
| Available data | Description of data | | Data sources | | |
| | – Total revenue regi | stration tax | – Danish Ministry of Finance | | |
| | Total revenue own | ership tax | – Danish Ministry of Finance | | |
| | Total revenue tolls | s (MAUT) | – ASECAP | | |
| | Total revenue vign | ettes | Statistics Denr | mark | |
| | Total revenue fuel | excise duty | Bottom-up and | alysis CE Delft | |
| | Total revenue VAT | on fuel and | Bottom-up and | alysis CE Delft | |
| | vehicle sales | | | | |
| Key results (in | Total revenue | Total revenue | Total revenue | Total VAT | |
| bln € ₂₀₁₃ , PPP | vehicle taxes | infrastructure | fuel excise duty | revenue | |
| adjusted) | | charges | | | |
| | 3.4 | 0.4 | 1.6 | 1.5 | |



B.5 Germany

| Country analysi | is: Germany | | | |
|-----------------------------|---|----------------------|------------------------------------|-------------------|
| National | CE Delft | | | |
| expert | | | | |
| Investments | | | | |
| Available | Description of data | | Data sources | |
| data on | Total investment | s on all roads for | Statistisches | Bundesamt |
| expenditures | the period 1979-2 | 2011 | | |
| | Total investment | s on motorways | | |
| | for the period 19 | 79-2011 | | |
| | Total investment | s on urban roads | | |
| | for the period 19 | 79-2011 | | |
| | Total investment | s on other roads | | |
| | for the period 19 | 70-2011 | | |
| Missing | Description of data | | Estimation approc | ach |
| data on | Total investments | s on all roads for | Estimated bas | ed on growth |
| expenditures | the period 2012-2 | 013 | rates in invest | ment data for |
| | | | Germany from | n International |
| | | | Transport For | um |
| | Breakdown of tot | al investments on | Breakdown ba | ised on the |
| | all roads to enhar | ncement and | average share | s enhancement |
| | renewal expendit | ures | and renewal e | expenditures have |
| | | | in Austria and | the Netherlands |
| Key results (in | Total investment | Share of total | Total | Total |
| mln € ₂₀₁₃ , PPP | expenditures in | investment | investment | investment |
| adjusted) | 2013 | expenditures in | costs in 2013 | costs per km |
| | | GDP in 2013 | | road in 2013 |
| | 9,340 | 0.5% | 22,002 | 0.034 |
| Development | Road infrastructure ir | ivestment levels ha | ve been rather stab | le in the period |
| over time in | 1996-2013. In real ter | ms, 2013 investmer | nts are just below t | he investment |
| the period | levels of 1996. Howeve | er, according to Ste | er Davies and Gleav | /e (2014) a |
| 1995-2013 | significant (long-term |) Dacklog in road in | ivestments exist for | roads managed |
| Operating and | by local authorities. | ondituros | | |
| | Description of data | | Data sources | |
| data on | Total OGM ovpon | dituros on all | Statistischer F | Rundocamt |
| expenditures | - Total Oam expen | and 1005 2011 | - Statistisches E | bundesami |
| experiatures | Total OSM expon | dituros on | | |
| | - Total Oam expen | a poriod | | |
| | 1005-2011 | e period | | |
| | Total O&M expen | ditures on urban | | |
| | roads for the peri | and 1995-2011 | | |
| | Total O&M expen | ditures on other | | |
| | roads for the peri | ind 1995-2013 | | |
| | rouds for the peri | 00 1775 2015 | | |
| | For all roads a break | lown to operation | | |
| | and maintenance exp | enditures is | | |
| | available for the enti | re period | | |
| Missing | Description of data | | Estimation approx | ach |
| data on | Total O&M expen | ditures on all | Estimated base | sed on growth |
| expenditures | roads for the peri | od 2012-2013 | rates in O&M | expenditure data |
| | · · · · | | for Germany | from |
| | | | International | Transport Forum |
| | Breakdown of tot | al O&M | Same shares o | f operation and |
| | expenditures on a | all roads to | maintenance e | expenditures in |

| Country analys | is: Germany | | | | |
|-----------------------------|--|------------------------|-----------------------------------|-------------------|--|
| | operation and m | aintenance | total O&M exp | enditures as in | |
| | expenditures for | the period 2012- | 2011 are assumed | | |
| | 2013 | | | | |
| Key results (in | Total O&M | O&M | Share of total | Share of O&M | |
| mln € ₂₀₁₃ , PPP | expenditures/costs | expenditures per | O&M | expenditures in | |
| adjusted) | in 2013 | km road in 2013 | expenditures in | total infra | |
| | | | GDP in 2013 | expenditures in | |
| | | | | 2013 | |
| | 6,879 | 0.011 | 0.3% | 42% | |
| Development | Expenditures on maintenance and operation of the German road network | | | | |
| over time in | increased notably sir | nce 2005, particularl | y for national roads | . Since that year | |
| the period | the national policy f | or the road network | is one of 'maintena | nce before | |
| 1995-2013 | extension', which ha | s led to a significant | increase in O&M ex | xpenditures. This | |
| | increase continued a | fter the start of the | economic crisis, ma | ainly due to two | |
| | economic stimulus p | ackages implemente | d in these years to | mitigate the | |
| | impacts of the econo | omic downturn in Ge | rmany. | | |
| Revenue from | taxes and charges in 2 | 2013 | | | |
| Available data | Description of data | | Data sources | | |
| | Total revenue own | nership tax | German Minist | ry of Finance | |
| | Total revenue toll | s (MAUT) | Bundesamt für | r Güterverkehr | |
| | Total revenue fue | l excise duty | Bottom-up and | alysis CE Delft | |
| | Total revenue VAT | on fuel and | Bottom-up and | alysis CE Delft | |
| | vehicle sales | | | | |
| Key results (in | Total revenue | Total revenue | Total revenue | Total VAT | |
| bln € ₂₀₁₃ , PPP | vehicle taxes | infrastructure | fuel excise duty | revenue | |
| adjusted) | | charges | | | |
| | 8.1 | 4.2 | 32.4 | 23.4 | |

B.6 France

| Country analysi | Country analysis: France | | | | | | |
|-----------------|--|--|--|--|--|--|--|
| National | SETEC | | | | | | |
| expert | | | | | | | |
| Investments | | | | | | | |
| Available | Description of data | Data sources | | | | | |
| data on | Total investments on all roads for | Ministry of Ecology, Sustainable | | | | | |
| expenditures | the period 2003-2013 | Development and Energy | | | | | |
| | | | | | | | |
| | Total investments on all roads for | International Transport Forum | | | | | |
| | the periods 1970-1984 and | | | | | | |
| | 1987-2002 | | | | | | |
| | Total investments on tolled | 2007-2013: ASFA annual report | | | | | |
| | motorways for the period 1985- | 1985-2013: Ministry of Ecology, | | | | | |
| | 2013 | Sustainable Development and | | | | | |
| | Breakdown to enhancement and | Energy | | | | | |
| | renewal expenditures for the | | | | | | |
| | period 2001-2013 | | | | | | |
| | Total investments on national | Ministry of Ecology, Sustainable | | | | | |
| | roads (excl. tolled motorways) for | Development and Energy | | | | | |
| | the period 1985-2013 | | | | | | |
| | Total investments on urban roads | Ministry of Ecology, Sustainable | | | | | |
| | for the period 2003-2013 | Development and Energy | | | | | |



| Country analys | is: France | | | | |
|-----------------------------|---|--|--|-----------------------|--|
| Missing | Description of data | | Estimation approach | | |
| data on | Total investment | s on all roads for | Estimated based on linear | | |
| expenditures | the period 1985- | 1986 | interpolation | | |
| | Breakdown of tot | tal investments on | Based on the | average shares | |
| | all roads to enha | ncement and | enhancement and renewal | | |
| | renewal expendi | tures | expenditures have in Austria and | | |
| | | | the Netherlan | ds | |
| Key results (in | Total investment | Share of total | Total | Total investment | |
| mln € ₂₀₁₃ , PPP | expenditures in | investment | investment | costs per km road | |
| adjusted) | , 2013 | expenditures in | costs in 2013 | , in 2013 | |
| | | , GDP in 2013 | | | |
| | 9,451 | 0.6% | 17,854 | 0.017 | |
| Development | In absolute terms, in | vestment levels in F | rance have been ra | ther stable over the | |
| over time in | period 1995-2013. Ho | wever, a gradually | decreasing trend is | shown for | |
| the period | investments as share | of GDP: between 19 | 995 and 2013 the sh | are of road | |
| 1995-2013 | infrastructure invest | ments in GDP has de | creased from almos | st 0.9 to 0.6%. | |
| Operating and | maintenance costs/ex | penditures | | | |
| Available | Description of data | | Data sources | | |
| data on | Total O&M exper | ditures on all | Ministry of Eco | logy, Sustainable | |
| expenditures | roads for the per | riod 2006-2013 | Development | and Energy | |
| | - Total O&M exper | ditures on tolled | – Ministry of Eco | logy. Sustainable | |
| | motorways for th | e period 2005- | Development | and Energy + ASFA + | |
| | 2013 | | APRR + SANEF | | |
| | Breakdown t | o operation and | | | |
| | maintenance | e expenditures for | | | |
| | the period 2010-2013 | | | | |
| | - Total O&M expenditures on - Ministry of Ecology | | logy. Sustainable | | |
| | national roads (e | s (excl. tolled Development and Energy | | and Energy | |
| | motorways for th | ne period | | | |
| | 2011-2013 | | | | |
| | Breakdown t | o operation and | | | |
| | maintenance | e expenditures for | | | |
| | the period 2 | 011-2013 | | | |
| | Total maintenand | ce expenditures | Ministry of Ecology, Sustainable | | |
| | on national road | s (excl. tolled | Development and Energy | | |
| | motorways) for t | he period | Development | | |
| | 1999-2004 | | | | |
| | Total O&M experi | ditures on urban | Ministry of Eco | logy. Sustainable | |
| | roads for the per | iod 2003-2013 | Development | and Energy | |
| Missing | Description of data | | Estimation approx | ach | |
| data on | Total O&M exper | ditures on all | - Estimated based on growth rate of | | |
| expenditures | roads for the per | iod 1995-2005 | O&M expenditures in Belgium | | |
| · | Breakdown of tot | tal O&M | Based on average shares in | | |
| | expenditures on | all roads to | Germany, The | e Netherlands and the | |
| | operation and maintenance | | | | |
| | expenditures for | the entire period | | | |
| Key results (in | Total O&M | O&M | Share of total | Share of O&M | |
| mln € ₂₀₁₃ , PPP | expenditures/costs | expenditures per | , O&M | expenditures in | |
| adjusted) | , in 2013 | km road in 2013 | expenditures in | , total infra | |
| , | | | , GDP in 2013 | expenditures in | |
| | | | | 2013 | |
| | 6,559 | 0.006 | 0.3% | 41% | |



| Country analysis: France | | | | | | |
|-----------------------------|--|----------------------|---|-----------------------|--|--|
| Development | O&M expenditure levels have been rather stable over the period 1995-2013. In | | | | | |
| over time in | contrast to many oth | er countries, the Fr | ench national gover | nment increased | | |
| the period | O&M expenditures af | ter the start of the | economic crisis to b | oost economic | | |
| 1995-2013 | growth. | | | | | |
| Revenue from | from taxes and charges in 2013 | | | | | |
| Available data | Description of data | | Data sources | | | |
| | Total revenue registration tax | | – Eurostat | | | |
| | Total revenue ownership tax | | - OECD Environmental tax database | | | |
| | Total revenue tolls | | - URF (2015), Faits et Chiffres 2014 | | | |
| | Total revenue vignettes | | – URF (2015), Fa | aits et Chiffres 2014 | | |
| | - Total revenue fuel excise duty | | Bottom-up analysis CE Delft | | | |
| | Total revenue VAT | on fuel and | Bottom-up ana | alysis CE Delft | | |
| | vehicle sales | | | | | |
| Key results (in | Total revenue | Total revenue | Total revenue | Total VAT revenue | | |
| bln € ₂₀₁₃ , PPP | vehicle taxes | infrastructure | fuel excise duty | | | |
| adjusted) | | charges | | | | |
| | 2.0 | 9.8 | 20.1 | 13.1 | | |

B.7 Italy

| Country analys | is: Italy | | | | |
|-----------------------------|---------------------------------------|---|---|---------------------|--|
| National | TRT | | | | |
| expert | | | | | |
| Investments | | | | | |
| Available | Description of data | | Data sources | | |
| data on | Total investments | s on all roads for | – CNIT | | |
| expenditures | the period 2004-2 | 2013 | | | |
| | Total investments | s on all roads for | International | Transport Forum | |
| | the periods 1970- | 1984 and 1987- | | | |
| | 2003 | | | | |
| | Total investments | s on motorways | – Parliamentar | y hearing of the | |
| | for the period 2008-2013 | | Minister of Ti | ransport in January | |
| | | | 2014 | | |
| Missing | Description of data | | Estimation approach | | |
| data on | Total investments | s on all roads for | Estimated based on linear | | |
| expenditures | the period 1985-1986 | | interpolation | | |
| | Breakdown of tota | al investments on | Based on the average shares | | |
| | all roads to enhar | ncement and | enhancement and renewal | | |
| | renewal expendit | ures | expenditures have in Austria and | | |
| | | | the Netherlan | ds | |
| Key results (in | Total investment | Share of total | Total | Total investment | |
| mln € ₂₀₁₃ , PPP | expenditures in | investment | investment | costs per km road | |
| adjusted) | 2013 | expenditures in | costs in 2013 | in 2013 | |
| | | GDP in 2013 | | | |
| | 3,215 | 0.2% | 16,980 | 0.024 | |
| Development | Road infrastructure in | vestments in Italy I | have significantly d | ecreased since the | |
| over time in | start of the economic | start of the economic crisis: a reduction of about 80% between 2007 and 2013. | | | |
| the period | This decrease in inves | stments affects bot | h national, regional | and local roads. | |
| 1995-2013 | | | | | |



| Uperating and maintenance costs/expenditures | | | | | |
|--|--|--|--|--|--|
| Available Description of data Data sources | | | | | |
| data on – Total O&M expenditures on all – International Transport | International Transport Forum | | | | |
| expenditures roads for the period 1995-2010 | | | | | |
| Total O&M expenditures on Parliamentary hearing or | - Parliamentary hearing of Minister of | | | | |
| motorways for the period 2008- Transport in January 20 | Transport in January 2014 | | | | |
| 2013 | | | | | |
| Missing Description of data Estimation approach | | | | | |
| data on - Total O&M expenditures on all - Estimated based on gro | wth rate of | | | | |
| expenditures roads for the period 2011-2013 O&M expenditures in Sp | pain | | | | |
| Breakdown of total O&M Based on average share | es in | | | | |
| expenditures on all roads to Germany, The Netherla | ands and the | | | | |
| operation and maintenance UK | | | | | |
| expenditures for the entire period | | | | | |
| Key results (in Total O&M O&M Share of total Sh | hare of O&M | | | | |
| mln \in_{2013} , PPP expenditures/costs expenditures per O&M exp | enditures in | | | | |
| adjusted) in 2013 km road in 2013 expenditures in | total infra | | | | |
| GDP in 2013 exp | enditures in | | | | |
| | 2013 | | | | |
| 5,432 0.008 0.4% | 63% | | | | |
| Development The expenditures on maintaining and operating Italian roads have dec | reased | | | | |
| over time in since 2007/2008. Maintenance budgets of both the national road oper | since 2007/2008. Maintenance budgets of both the national road operator (ANAS) | | | | |
| the period and of provinces/municipalities were drastically cut due to the econo | mic crisis. | | | | |
| 1995-2013 This also resulted in a shift from major long-term maintenance project | ts to minor | | | | |
| (and hence less expensive) short-term projects (for national roads). Ir | n the longer | | | | |
| term this shift may lead to a backlog of road maintenance on Italian r | oads. | | | | |
| Revenue from taxes and charges in 2013 | | | | | |
| Available data Description of data Data sources | | | | | |
| Total revenue registration tax ACI (2015), Stima del generativa | ettito delle | | | | |
| principali imposte e tas | se gravanti | | | | |
| su stra | | | | | |
| - Total revenue ownership tax - Same source as for tota | l revenue | | | | |
| registration tax | | | | | |
| - Total revenue tolls - ASECAP | - ASECAP | | | | |
| - Total revenue fuel excise duty - Bottom-up analysis CE L | - Bottom-up analysis CE Delft | | | | |
| - Total revenue var on fuet and - ACT (2015), Stillid det ge | – ACI (2015), Stima del gettito delle | | | | |
| venicle sales principali impose e las | se gravanti lucic | | | | |
| Su Stra, DOLLOIII-UP and | (3)2 | | | | |
| Key results (in Total revenue Total revenue Total revenue Total | VAT revenue | | | | |
| hin from PPP vehicle taxes infrastructure fuel excise duty | ALIEVENUE | | | | |
| adjusted) | | | | | |
| 7.0 6.6 23.3 | 14 7 | | | | |



B.8 The Netherlands

| Country analysi | is: The Netherlands | | | |
|---|---|--|--|--|
| National | CE Delft | | | |
| expert | | | | |
| Investments | | | | |
| Available | Description of data | | Data sources | |
| data on expenditures | Total investments on all roads for the period 1979-2013 Breakdown to enhancement and renewal expenditures for entire | | Calculated expenditure of roads | by summing up the res on all different types |
| | Total investments on motorways for the period 1979-2013 Breakdown to enhancement and renewal expenditures for entire period Total investments on urban roads for the periods 1979-2001 and 2005-2013 | | 2001-2013 Transport 1985-2000 1979-1984 Efficiënte 2005-2013 1985-2001 1979-1984 Efficiënte | : Dutch Ministry of - Annual accounts : Statistics Netherlands : CE Delft (1999), prijzen voor het verkeer : Statistics Netherlands : CE Delft (1999), prijzen voor het verkeer |
| | Total investments on other roads For the periods 1979-2001 and 2005-2013 | | ces as for urban roads | |
| Missing | Description of data | | Estimation app | roach |
| data on expenditures | Total investments on urban non-urban roads for the per 2002-2004 Breakdown of total investments | | Estimated based on linear interpolation Based on DVS (2001) and DVS (2007) | |
| | enhancement and expenditures | renewal | (2007) | |
| Key results (in mln €2013, PPP adjusted) | Total investment expenditures in 2013 | Share of total investment expenditures in GDP in 2013 | Tota investmer costs in 201 | al Total investment at costs per km road 3 in 2013 |
| | 5,124 | 0.9% | 7,58 | 5 0.055 |
| Development over time in the period 1995-2013 Operating and | Investment levels in the Netherlands have shown a gradually increasing trend between 1995 and 2004 and were remain at a relatively high level in the years afterwards. The high level of investments in recent years may be explained by several large-scale investment projects aimed at widening significant parts of the Dutch motorway network. | | | |
| Available | Description of data | | Data sources | |
| data on expenditures | Total O&M expenditures on all roads for the period 1995-2013 Breakdown to operation and maintenance expenditures for entire period Total O&M expenditures on motorways for the period 195-2013 Breakdown to operation and maintenance expenditures for entire period | | Calculated I expenditure of roads | by summing up the es on all different types |
| | | | 2001-2013: Transport - 1995-2000: Statistics Mathematics Mathematics | Dutch Ministry of Annual accounts Statistics Netherlands |
| | roads for the perio | ods 1995-2001 | - Statistics Ne | eurertanus |

| Country analysi | is: The Netherlands | | | |
|-----------------------------|--|--------------------------------------|---|---------------------------------|
| | and 2005-2013 – Total O&M expen urban roads for t 2013 | nditures on non- the period 1995- | Statistics Neth | erlands |
| Missing | Description of data | | Estimation approx | ach |
| data on expenditures | Total O&M expenditures on urban and non-urban roads for the period 2002-2004 Breakdown of total O&M expenditures on urban and non-urban roads to operation and maintenance expenditures | | Estimated base extrapolation Based on DVS (2007) | sed on linear (2001) and DVS |
| Key results (in | Total O&M | O&M | Share of total | Share of O&M |
| mln € ₂₀₁₃ , PPP | expenditures/costs | expenditures per | O&M | expenditures in |
| adjusted) | in 2013 | km road in 2013 | expenditures in | total infra |
| | | | GDP in 2013 | expenditures in |
| | | | | 2013 |
| | 1,448 | 0.010 | 0.2% | 22% |
| Development | Expenditures on maintaining and operating the Dutch road network show an | | | |
| over time in | increasing trend bety | ween 1995-2004. In t | the period 2005-201 | 2 O&M expenditures |
| the period | slightly decreased (b | oth in absolute term | is and as share of G | DP), followed by a |
| 1995-2013 | significant drop in 20 |)13. | | |
| Revenue from t | taxes and charges in 2 | 2013 | _ | |
| Available data | Description of data | | Data sources | |
| | Total registration | tax | Dutch Ministry of Finance | |
| | Total revenue owr | nership tax | Dutch Ministry of Finance | |
| | Total revenue toll | S | – ASECAP | |
| | Total revenue vigr | nettes | RAI/Bovag (2014), Mobiliteit in Cijfers | |
| | Total revenue fue | l excise duty | - Dutch Ministry of Finance | |
| | Total revenue VAT on fuel and | | – RAI/Bovag (2014), Mobiliteit in | |
| | vehicle sales | | Cijfers | |
| Key results (in | Total revenue | Total revenue | Total revenue | Total VAT revenue |
| bln € ₂₀₁₃ , PPP | vehicle taxes | infrastructure | fuel excise duty | |
| adjusted) | | charges | | |
| | 5.6 | 0.1 | 6.9 | 2.1 |

B.9 Poland

| Country analysi | is: Poland | |
|-------------------------|---|--|
| National | Agnieszka Markowska | |
| expert | | |
| Investments | | |
| Available | Description of data | Data sources |
| data on expenditures | Total investments on all roads for the period 1987-2013 Breakdown to enhancement and renewal expenditures for period 2008-2013 Total investments on motorways for the period 2008-2013 Breakdown to enhancement and renewal expenditures for entire period | Total investments: International Transport Forum (crosschecked with data GDDKiA) Breakdown: GDDKiA GDDKiA |

| Country analys | is: Poland | | | |
|--|---|---|--|---|
| | Total investment for the period 20 Breakdown to e | s on urban roads 08-2009 enhancement and | – GDDKiA | |
| | renewal expenditures for entire period Total investments on other roads for the period 2008-2009 Breakdown to enhancement and | | – GDDKiA | |
| | period | | | |
| Missing | Description of data | | Estimation approx | ach |
| data on | Total investments | s on all roads for | Estimated bas | ed on extrapolation |
| expenditures | the period 1979-1 | 986 | | |
| | Breakdown of tot all roads to enhau renewal expendit period 1979-2007 | al investments on ncement and ures for the | Average share and renewal e investments fe are assumed f 1979-2007 | s of enhancement expenditures in total or period 2008-2013 or the period |
| Key results (in | Total investment | Share of total | Total | Total investment |
| mln € ₂₀₁₃ , PPP adjusted) | expenditures in 2013 | investment expenditures in GDP in 2013 | investment costs in 2013 | costs per km road in 2013 |
| | 4,068 | 0.8% | 7,472 | 0.026 |
| Development over time in the period 1995-2013 | shown a significant in EU and the World Bar significantly increasir funding combined wit Polish national budge | crease, in part as re- crease, in part as re- k. This support has g the national mote the negative cons t, explains the shar | u), investments on esult of substantial led to a large-scale prway network. A re sequences of the ec p decrease in invest | Polish roads have co-financing from the programme to eduction in EU onomic crisis on the tment levels in the |
| Operating and | maintenance costs/ext | enditures | | |
| Available | Description of data | | Data sources | |
| data on expenditures | Total O&M expenditures on all roads for the period 1995-2013 Breakdown to operation and maintenance expenditures for the period 2008-2013 Total O&M expenditures on motorways for the period 2008- 2013 | | Total expendit Transport Foru Breakdown: G GDDKiA | ures: International ım DDKiA |
| | Breakdown to operation and maintenance expenditures Total O&M expenditures on urban roads for the period 2008-2009 Breakdown to operation and maintenance expenditures Total O&M expenditures on other roads for the period 2008-2009 Breakdown to operation and maintenance expenditures | | – GDDKiA – GDDKiA | |
| Missing | Description of data | | Estimation approx | ach |
| data on expenditures | Breakdown total expenditures on a operation and ma | O&M all roads to nintenance | Average share maintenance period 2008-2 | es of operation and expenditures in the 2013 are applied for |
| | expenditures for | the period 1995- | the period 1995-2007 as well | |

| Country analysi | Country analysis: Poland | | | | |
|---|---|--|--|---|--|
| | 2007 | | | | |
| Key results (in mln € ₂₀₁₃ , PPP adjusted) | Total O&M expenditures/costs in 2013 | O&M expenditures per km road in 2013 | Share of total O&M expenditures in GDP in 2013 | Share of O&M expenditures in total infra expenditures in 2013 | |
| | 761 | 0.003 | 0.2% | 16% | |
| Development over time in the period 1995-2013 | O&M expenditures on Polish roads significantly increased after the accession of Poland to the European Union in 2004. Supported by EU funding overall road improvements were realised through road maintenance and rehabilitation projects. In most recent years (2012-2013) expenditures levels have significantly fallen, partly explained by cuts in national budgets (due to the economic crisis) and roduced EU funding | | | | |
| Revenue from t | axes and charges in 2 | 2013 | | | |
| Available data | Description of data | | Data sources | | |
| | Total registration tax Total revenue ownership tax Total revenue tolls Total revenue fuel excise duty Total revenue VAT on fuel and vehicle sales | | Polish Ministry Polish Ministry ASECAP; GDDH Bottom-up and Bottom-up and | y of Finance y of Finance KiA alysis CE Delft alysis CE Delft | |
| Key results (in | Total revenue | Total revenue | Total revenue | Total VAT revenue | |
| bln € ₂₀₁₃ , PPP adjusted) | vehicle taxes | infrastructure charges | fuel excise duty | | |
| | 1.0 | 0.9 | 11.2 | 6.8 | |

B.10 Spain

| Country analysi | is: Spain | | | |
|-------------------------|---|---|--|--|
| National | University of Madrid | | | |
| expert | | | | |
| Investments | | | | |
| Available | Description of data | Data sources | | |
| data on expenditures | Total investments on all roads for the periods 1979-1983 and 1988- 2013 Total investments on tolled motorways for the period 1988- 2013 Total investments on national roads for the period 1988-2013 Total investments on departmental and regional roads Total investments on local roads for the period 1988-2013 | – Ministerio de Fomento | | |
| Missing | Description of data | Estimation approach | | |
| data on expenditures | Total investments on all roads for the period 1984-1987 Breakdown of total investments on all roads to enhancement and renewal expenditures for the entire period | Estimated based on linear interpolation Based on average shares enhancement and renewal expenditures have in Austria and the Netherlands | | |



| Country analysi | is: Spain | | | |
|-----------------------------|---|-----------------------|--|------------------------|
| Kev results (in | Total investment | Share of total | Total | Total investment |
| mln €2012 PPP | expenditures in | investment | investment | costs per km road |
| adjusted) | 2013 | expenditures in | costs in 2013 | in 2013 |
| uujuseed) | 2010 | GDP in 2013 | 2015 11 2015 | |
| | 3,951 | 0.3% | 14,332 | 0.022 |
| Development | Over the last 20 year | s Spain have experie | enced a road constru | uction boom; since |
| over time in | 1991 investment levels have been significantly higher than before, resulting in a | | | |
| the period | growth of the Spanisl | h motorway network | by almost 250% (fr | om 4,496 km in 1993 |
| 1995-2013 | to 11,676 km in 2013). The gradually increasing trend in road investments | | | |
| | stopped in 2009; due | to the economic cr | isis, national investr | ment budgets were |
| | significantly cut (in 2 | 012 this budget was | 98% lower than in 2 | 2008), but also severe |
| | cuts have been made | e to local budgets. T | his reduction may p | artly reflect a return |
| | to long-term spendin | g patterns after a co | onstruction boom of | two decades. |
| Operating and | maintenance costs/ex | penditures | | |
| Available | Description of data | | Data sources | |
| data on | Total O&M exper | nditures on tolled | Ministerio de F | omento |
| expenditures | motorways for th | ne period 1995- | | |
| | 2013 | | | |
| | Total O&M exper | nditures on | Ministerio de F | omento |
| | national roads fo | or the period 1995- | | |
| | 2013 | | | |
| | Total O&M expenditures on | | Ministerio de F | omento |
| | departmental an | d regional roads | | |
| | for the period 19 | 995-2013 | | |
| | Total O&M exper | nditures on local | Ministerio de Hacienda y | |
| | roads for the per | riod 2002-2013 | Administraciones Publicas | |
| | | | | |
| | | | | |
| Missing | Description of data | | Estimation approx | ach |
| data on | - Total O&M exper | ditures on all | Based on (est | imated) total O&M |
| expenditures | roads for the period 1995-2013 expenditures on different road | | | on different road |
| | | | types | |
| | Total O&M exper | nditures on local | Same growth | rate in O&M |
| | roads for the per | riod 1995-2001 | expenditures | as for departmental |
| | | | roads is assun | ned |
| | Breakdown total | O&M | Based on aver | rage shares in |
| | expenditures on | all roads to | Germany, the | Netherlands and the |
| | operation and ma | aintenance | UK | |
| | expenditures for | the period | | |
| | 1995-2007 | | | |
| Key results (in | Total O&M | O&M | Share of total | Share of O&M |
| mln € ₂₀₁₃ , PPP | expenditures/costs | expenditures per | O&M | expenditures in |
| adjusted) | in 2013 | km road in 2013 | expenditures in | total infra |
| | | | GDP in 2013 | expenditures in |
| | | | | 2013 |
| | 2,967 | 0.004 | 0.3% | 43% |
| Development | O&M expenditures or | n Spanish roads show | a sharp increasing | trend over the period |
| over time in | 1995-2009. However, | , the review of the r | oad infrastructure s | pending policies |
| the period | started in 2010 has s | ignificantly affected | the maintenance b | udgets, both at |
| 1995-2013 | national and local sc | ale, leasing to a red | uction in O&M expe | nditure levels. |
| | The effect of a fallin | g budget allocation | tor road network op | perations is |
| | particularly evident for pavement rehabilitation. | | | |



| Country analysis: Spain | | | | | | | |
|-----------------------------|--|---------------|--|-------------------|--|--|--|
| Revenue from t | axes and charges in 2 | | | | | | |
| Available data | Description of data | | Data sources | | | | |
| | Total registration | tax | – Agencia Tributaria | | | | |
| | Total revenue owr | ership tax | – ANFAC | | | | |
| | Total revenue toll | S | – ASECAP | | | | |
| | Total revenue fuel | excise duty | Bottom-up analysis CE Delft | | | | |
| | Total revenue VAT | on fuel and | ANFAC; Bottom-up analysis CE Delft | | | | |
| | vehicle sales | | | | | | |
| Key results (in | Total revenue | Total revenue | Total revenue | Total VAT revenue | | | |
| bln € ₂₀₁₃ , PPP | vehicle taxes infrastructure | | fuel excise duty | | | | |
| adjusted) | | charges | | | | | |
| | 3.4 | 1.7 | 13.0 | 9.2 | | | |

B.11 Sweden

| Country analysi | s: Sweden | | | | | | | |
|--|--|-----------------------------|--|----------------------|--|--|--|--|
| National | Nilsson Production | | | | | | | |
| expert | | | | | | | | |
| Investments | | | | | | | | |
| Available | Description of data | | Data sources | | | | | |
| data on | Total investments | s on all roads for | 1998- 2013: Statistics Sweden | | | | | |
| expenditures | the periods 1979- | 1984 and 1987- | 1987 - 1997: International | | | | | |
| | 2013 | | Transport Fo | rum | | | | |
| | | | 1979 - 1984: International | | | | | |
| | | | Transport Forum | | | | | |
| | Total investments for the period 199 | s on urban roads 98-2013 | – 1998-2013: Statistics Sweden | | | | | |
| | Total investments | s on non-urban | – 2002-2013: S | tatistics Sweden | | | | |
| | roads for the peri | od 1979-2013 | 1979 - 2001: | Statistical Yearbook | | | | |
| | | | Sweden | | | | | |
| Missing | Description of data | | Estimation appro | ach | | | | |
| data on | Total investments | on all roads for | Based on linear interpolation | | | | | |
| expenditures | the period 1985-1 | 1986 | | | | | | |
| | Breakdown of tota | al investments | Breakdown based on the average | | | | | |
| | on all roads to en | hancement and | shares enhancement and renewal | | | | | |
| | renewal expendit | ures | expenditures have in Austria and | | | | | |
| | | | the Netherlar | nds | | | | |
| Key results (in | Total investment | Share of total | Total | Total investment | | | | |
| mln € ₂₀₁₃ , PPP | expenditures in | investment | investment | costs per km road | | | | |
| adjusted) | 2013 | expenditures | costs in 2013 | in 2013 | | | | |
| | | in GDP in 2013 | | | | | | |
| | 1,441 | 0.6% | 1,694 | 0.008 | | | | |
| Development | Since the start of this | century, investme | ents in the Swedish | road network show a | | | | |
| over time in | constant increasing tr | end. This trend ev | en continues durin | g the economic | | | | |
| the period | crisis. | | | | | | | |
| 1995-2013 | | | | | | | | |
| Operating and maintenance costs/expenditures | | | | | | | | |
| Available | Description of data | | Data sources | | | | | |
| data on | Total O&M expend | ditures on all | Statistics Swe | eden | | | | |
| expenditures | roads for the peri | od 2001-2013 | | | | | | |
| | Total maintenance | e expenditures | International Transport Forum | | | | | |
| | on all roads for th | e period 1995- | (ITF) | | | | | |
| | 2013 | | | | | | | |
| | Total O&M expend | ditures on urban | Statistics Sweden | | | | | |



| Country analysis: Sweden | | | | | | | |
|-----------------------------|---|--|---|--|-------------------|--|--|
| | roads for the p – Total O&M exp urban roads fo 2013 | period 2001-2013 penditures on non r the period 1995 | 2001-2013: Statistics Sweden 1995-2000: Statistical yearbook Sweden | | | | |
| Missing data | Description of date | 2 | | Estimation approach | | | |
| on expenditures | Total O&M expenditures on all roads for the period 1995-2000 | | | Based on total operation and total maintenance expenditures for the period 1995-2000 | | | |
| | Total operation all roads for th | n expenditures o ne period 2001-20 | Difference between total O&M expenditures from Statistics Sweden and total maintenance expenditures from ITF | | | | |
| | Total operation all roads for the | n expenditures o ne period 1995-20 | n)00 | Based on same growth rate as for total maintenance expenditures | | | |
| Key results (in | Total O&M | 0 | £М | Share of total | Share of O&M | | |
| mln € ₂₀₁₃ , PPP | expenditures/cos | expenditures p | ber | O&M | expenditures in | | |
| adjusted) | ts in 2013 | km road in 20 | 013 | expenditures in | total infra | | |
| | | | | GDP in 2013 | expenditures in | | |
| | 4 777 | 0.0 | 207 | 0.2% | 2013 | | |
| Dovelopment | 1,377 0.006 | | | U.3% | 49% | | |
| over time in | increasing trend over the last 15 years. In contract to pointhouring countries | | | | | | |
| the period | like Denmark and Finland, the OBM expenditures as part of CDP did not | | | | | | |
| 1995-2013 | significantly fall over the last 15 years, but remained rather constant | | | | | | |
| Revenue from t | axes and charges in | 2013 | | but i cilitanica i atin | | | |
| Available data | Description of date | 2 | Da | Data sources | | | |
| | - Total revenue ov | wnership tax | - | Swedish Ministry of | of Finance | | |
| | Total revenue tolls | | - | - Swedish Ministry of Finance | | | |
| | Total revenue vignettes | | - | Swedish Ministry of Finance | | | |
| | Total revenue fuel excise duty Total revenue VAT on fuel and | | - | Bottom-up analysis CE Delft | | | |
| | | | - | Bottom-up analysis CE Delft | | | |
| | vehicle sales | | | | | | |
| Key results (in | Total revenue | Total revenue | T | otal revenue fuel | Total VAT revenue | | |
| bln € ₂₀₁₃ , PPP | vehicle taxes | infrastructure | | excise duty | | | |
| adjusted) | | charges | | | | | |
| | 1.0 | 0.2 | 3.9 | 2.7 | | | |



B.12 UK

| Country analysis: UK | | | | | | | | |
|--|---|---------------------|--|--------------------------------|--|--|--|--|
| National | National CE Delft | | | | | | | |
| expert | | | | | | | | |
| Investments | | | | | | | | |
| Available | Description of data | | Data sources | | | | | |
| data on | Total investmen | ts on all roads for | – Department | for Transport | | | | |
| expenditures | the periods 1979 | -1984 and 1987- | | | | | | |
| | 2013 | | | | | | | |
| | Total investmen | ts on motorways | | | | | | |
| | + trunk roads for | the period 1987- | | | | | | |
| | 2013 | | | | | | | |
| Missing | Description of data | | Estimation approach | | | | | |
| data on | Total investment | s on all roads for | Based on linear interpolation | | | | | |
| expenditures | the period 1985- | 1986 | | | | | | |
| | Breakdown of to | tal investments | Breakdown based on the average | | | | | |
| | on all roads to e | nhancement and | shares enhan | shares enhancement and renewal | | | | |
| | renewal expendi | tures | expenditures have in Austria and | | | | | |
| Kov rogulta (in | Total investment | Shara of total | | Total investment | | | | |
| min face DDD | avpondituros in | invostment | investment | costs par km road | | | | |
| adjusted) | 2013 experiorcures | expenditures | costs in 2013 | in 2013 | | | | |
| aujusteu) | 2013 | in GDP in 2013 | | 11 2015 | | | | |
| | 6 400 | 0.3% | 11 425 | 0.027 | | | | |
| Development | Annual investments | on the UK road net | work have signification | ntly fall between | | | | |
| over time in | 1991 and 1997 and remained rather stable at this low level over the years | | | | | | | |
| the period | afterwards. The relatively low investment levels in the UK have resulted in one | | | | | | | |
| 1995-2013 | of the most dense European truck road network, particularly since no | | | | | | | |
| | considerable road expansion has taken place over the past 10 years. | | | | | | | |
| Operating and maintenance costs/expenditures | | | | | | | | |
| Available | Description of data Data sources | | | | | | | |
| data on | Total O&M expension | nditures on all | Department f | or Transport | | | | |
| expenditures | roads for the pe | riod 1995-2013 | | | | | | |
| | Breakdown to | operation and | | | | | | |
| | maintenance e | xpenditures for | | | | | | |
| | the entire peri | od | | | | | | |
| | Total O&M expension | iditures on | | | | | | |
| | motorways + tru | nk roads for the | | | | | | |
| | Breakdown to | operation and | | | | | | |
| | • Dieakdowii to | xpenditures for | | | | | | |
| | the entire peri | od | | | | | | |
| Missing | Description of data | | Estimation approach | | | | | |
| data on | , | | rr | | | | | |
| expenditures | | | | | | | | |
| Key results (in | Total O&M | O&M | Share of total | Share of O&M | | | | |
| mln € ₂₀₁₃ , PPP | expenditures/cos | expenditures per | O&M | expenditures in | | | | |
| adjusted) | ts in 2013 | km road in 2013 | expenditures in | total infra | | | | |
| | | | GDP in 2013 | expenditures in | | | | |
| | 2 472 | 0.008 | 0.2% | 2013 | | | | |
| Development | Road operation and u | naintenance expen | ditures in the LIK v | aried slightly over | | | | |
| over time in | the period 1995-2010 |). Since 2010 maint | enance budgets ha | ve fallen by almost | | | | |
| the period | 40%. Local budgets (| making up 70 to 80 | % of all road mainte | enance expenditures | | | | |
| 1995-2013 | in the UK) are under pressure from growing spending trends in social care and | | | | | | | |
| | environmental services. | | | | | | | |

| Country analysis: UK | | | | | | | |
|--|---|-----------------|---|-------------------|--|--|--|
| Revenue from taxes and charges in 2013 | | | | | | | |
| Available data | Description of dat | а | Data sources | | | | |
| | Total revenue o | wnership tax | Department for Transport Statistics | | | | |
| | | | (only figures for G | ireat Britain | | | |
| | Total revenue to | olls | available) | | | | |
| | Total revenue fi | uel excise duty | ASECAP; Transport for London | | | | |
| | Total revenue VAT on fuel and | | Bottom-up analysis CE Delft | | | | |
| | vehicle sales | | ONS Family spending report 2014; | | | | |
| | | | Bottom-up analysis CE Delft | | | | |
| Key results (in | Total revenue | Total revenue | Total revenue fuel | Total VAT revenue | | | |
| bln € ₂₀₁₃ , PPP | vehicle taxes infrastructure | | excise duty | | | | |
| adjusted) | charges | | | | | | |
| | 4.7 | 0.3 | 28.8 | 12.2 | | | |



Annex C Tax revenue not adjusted for PPP

In Table 2 (Section 2.4) the total road tax revenue in the EU27 Member States in 2013 are presented (figures for Cyprus can be found in the textbox below that table). These figures are adjusted for PPP. The unadjusted figures can be found in Table 23.

| Member State | Registration tax | Ownership tax | Tolls and vignettes | Fuel excise duty | VAT on registration taxes | VAT on fuel excise duty | VAT on vehicle purchase | VAT on fuel | Total (incl. VAT on | Total (incl. all VAT) |
|-----------------|---------------------|------------------|---------------------|---------------------|---------------------------------|----------------------------|-------------------------------|-------------|------------------------|--------------------------|
| Austria | 0.5 | 1.8 | 1.7 | 3.8 | 0.1 | 0.6 | 1.5 | 0.9 | 8.5 | 10.8 |
| Belgium | 0.5 | 1.7 | 0.1 | 4.2 | 0.1 | 0.6 | 1.3 | 0.9 | 7.2 | 9.4 |
| Bulgaria | - | 0.1 | 0.1 | 0.9 | - | 0.1 | 0.1 | 0.3 | 1.2 | 1.6 |
| Cyprus | 0.0 | 0.1 | 0.0 | 0.3 | 0.0 | 0.0 | 0.0 | 0.1 | 0.5 | 0.6 |
| Czech Republic | - | 0.2 | 0.5 | 2.9 | - | 0.4 | 0.6 | 0.6 | 4.0 | 5.1 |
| Denmark | 2.7 | 1.9 | 0.6 | 2.1 | 0.4 | 0.4 | 0.7 | 0.5 | 8.0 | 9.3 |
| Germany | - | 8.5 | 4.4 | 34.0 | - | 5.1 | 13.1 | 6.4 | 52.0 | 71.5 |
| Estonia | - | 0.0 | - | 0.3 | - | 0.0 | 0.1 | 0.1 | 0.4 | 0.6 |
| Finland | 0.9 | 0.9 | - | 2.5 | 0.2 | 0.4 | 0.3 | 0.3 | 4.9 | 5.5 |
| France | 2.0 | 0.2 | 11.1 | 22.8 | 0.4 | 3.0 | 7.1 | 4.3 | 39.5 | 50.9 |
| Greece | 0.1 | 1.2 | 0.5 | 3.4 | 0.0 | 0.5 | 0.1 | 0.6 | 5.8 | 6.5 |
| Croatia | 0.0 | 0.0 | 0.3 | 0.8 | 0.0 | 0.1 | 0.1 | 0.3 | 1.3 | 1.7 |
| Hungary | 0.1 | 0.1 | 0.2 | 1.6 | 0.0 | 0.3 | 0.3 | 0.5 | 2.2 | 3.0 |
| Ireland | 0.5 | 1.1 | 0.2 | 2.3 | 0.1 | 0.3 | 0.3 | 0.4 | 4.5 | 5.2 |
| Italy | 1.3 | 5.7 | 6.7 | 23.5 | 0.3 | 4.0 | 5.6 | 4.6 | 41.5 | 51.6 |
| Latvia | 0.0 | 0.1 | - | 0.4 | 0.0 | 0.0 | 0.1 | 0.1 | 0.5 | 0.6 |
| Lithuania | - | 0.1 | 0.0 | 0.6 | - | 0.1 | 0.1 | 0.2 | 0.8 | 1.0 |
| Luxembourg | - | 0.1 | 0.0 | 0.9 | - | 0.1 | 0.2 | 0.2 | 1.1 | 1.4 |
| Malta | 0.0 | 0.0 | - | 0.1 | 0.0 | 0.0 | 0.0 | 0.0 | 0.2 | 0.2 |
| Netherlands | 1.2 | 5.0 | 0.2 | 7.6 | 0.3 | 0.8 | 0.5 | 0.8 | 15.0 | 16.2 |
| Poland | 0.3 | 0.2 | 0.5 | 6.4 | 0.1 | 0.9 | 1.2 | 1.7 | 8.5 | 11.4 |
| Portugal | 0.4 | 0.5 | 0.8 | 2.6 | 0.1 | 0.4 | 0.5 | 0.7 | 4.8 | 5.9 |
| Romania | 0.2 | 0.2 | 0.2 | 2.0 | 0.0 | 0.2 | 0.2 | 0.4 | 2.9 | 3.5 |
| Slovakia | 0.0 | 0.1 | 0.2 | 1.1 | 0.0 | 0.1 | 0.3 | 0.1 | 1.6 | 2.0 |
| Slovenia | 0.0 | 0.1 | 0.3 | 1.0 | 0.0 | 0.1 | 0.2 | 0.2 | 1.6 | 2.0 |
| Spain | 0.3 | 2.8 | 1.6 | 11.7 | 0.1 | 2.0 | 2.3 | 3.8 | 18.5 | 24.7 |
| Sweden | - | 1.3 | 0.3 | 5.2 | - | 0.9 | 1.7 | 1.0 | 7.7 | 10.4 |
| UK | - | 5.1 | 0.4 | 31.3 | - | 4.3 | 4.9 | 4.1 | 41.1 | 50.1 |
| EU27 | 0.5 | 1.8 | 1.7 | 3.8 | 0.1 | 0.6 | 1.5 | 0.9 | 8.5 | 10.8 |

Table 23 Total revenue from road taxes/charges in the EU in 2013 (billion ε_{2013})



Annex D Average tax/charge revenue

This annex presents the average revenue from taxes and charges expressed in € per 1,000 vehicle kilometres for passenger cars (Figure 49), motorcycles (Figure 50), busses (Figure 51) and HGVs (Figure 52). In Section 2.5 the average revenue for these vehicle types expressed in € per 1,000 passenger kilometres or € per 1,000 tonne kilometres are presented. Any differences in the 'ranking' of countries in both types of graphs are caused by differences in average occupancy rates (passenger cars, motorcycles and busses) or loads (HGVs) between countries.



Average revenue from taxes and charges for passenger cars in 2013 Figure 49



Figure 50 Average revenue from taxes and charges for motorcycles in 2013



Figure 51 Average revenue from taxes and charges for busses in 2013




Figure 52 Average revenue from taxes and charges for HGVs in 2013





Annex E Infrastructure expenditures not adjusted for PPP

The PPP adjusted investments, O&M expenditures and total road infrastructure expenditures are presented in Table 5 (Section 3.3), Table 6 (Section 3.4) and Table 7 (Section 3.5), respectively. The figures unadjusted for PPP can be found in Table 24, Table 25 and Table 26.

| Member State | Long-term (1995- | Investments in 2013 | Ratio 2013 | |
|-----------------|--------------------|---------------------|--------------------|--|
| | 2013) average | | investments/long | |
| | annual investments | | term average | |
| | | | annual investments | |
| Austria | 2,365 | 1,970 | 83% | |
| Belgium | 1,536 | 1,432 | 93% | |
| Bulgaria | 273 | 380 | 139% | |
| Czech Republic | 1,597 | 897 | 569 | |
| Denmark | 1,163 | 1,047 | 90% | |
| Germany | 13,516 | 12,379 | 92% | |
| Estonia | 128 | 120 | 94% | |
| Finland | 909 | 1,115 | 123% | |
| France | 13,805 | 11,961 | 87% | |
| Greece | 1,591 | 921 | 58 % | |
| Croatia | 718 | 450 | 63% | |
| Hungary | 692 | 401 | 58% | |
| Ireland | 1,017 | 561 | 55% | |
| Italy | 7,157 | 2,663 | 37% | |
| Latvia | 161 | 150 | 93% | |
| Lithuania | 274 | 253 | 92 % | |
| Luxembourg | 199 | 220 | 110% | |
| Malta | 24 | 38 | 159% | |
| The Netherlands | 5,340 | 5,400 | 101% | |
| Poland | 3,180 | 2,931 | 92 % | |
| Portugal | 1,609 | 213 | 13% | |
| Romania | 2,343 | 3,061 | 131% | |
| Slovakia | 493 | 360 | 73% | |
| Slovenia | 444 | 104 | 23% | |
| Spain | 6,889 | 2,926 | 42% | |
| Sweden | 1,943 | 2,469 | 127% | |
| United Kingdom | 6,075 | 6,166 | 102% | |
| EU27 | 75,439 | 60,588 | 78% | |

Table 24 Total investments in road infrastructure (mln €2013)

Note: As no reliable data on Cypriot road infrastructure investments was available, Cyprus is not shown in this table.



| Member State | Long-term (1995- 2013) average annual O&M expenditures | O&M expenditures in 2013 | Ratio 2013 O&M expenditures/long term average annual O&M expenditures |
|-----------------|---|-----------------------------|---|
| Austria | 3,357 | 2,527 | 75% |
| Belgium | 1,008 | 1,020 | 101% |
| Bulgaria | 131 | 153 | 117% |
| Czech Republic | 744 | 695 | 93% |
| Denmark | 1,245 | 920 | 74% |
| Germany | 7,206 | 9,118 | 127% |
| Estonia | 100 | 109 | 109% |
| Finland | 1,531 | 1,064 | 69 % |
| France | 7,986 | 8,300 | 104% |
| Greece | 321 | 243 | 76% |
| Croatia | 312 | 319 | 102% |
| Hungary | 897 | 800 | 89 % |
| Ireland | 202 | 235 | 116% |
| Italy | 9,514 | 4,499 | 47% |
| Latvia | 113 | 143 | 127% |
| Lithuania | 207 | 182 | 88% |
| Luxembourg | 67 | 75 | 111% |
| Malta | 6 | 4 | 60% |
| The Netherlands | 1,808 | 1,525 | 84% |
| Poland | 605 | 548 | 9 1% |
| Romania | 422 | 319 | 76% |
| Portugal | 1,088 | 1,059 | 97 % |
| Slovakia | 225 | 291 | 129% |
| Slovenia | 184 | 176 | 9 5% |
| Spain | 2,922 | 2,197 | 75% |
| Sweden | 1,799 | 2,360 | 131% |
| United Kingdom | 4,890 | 3,346 | 68% |
| EU27 | 48,889 | 42,226 | 83% |

Table 25 Total O&M expenditures on road infrastructure (mln €2013)

Note: As no reliable data on Cypriot road infrastructure investments was available, Cyprus is not shown in this table.

Table 26 Total expenditures on road infrastructure (mln €2013)

| Member State | Long-term (1995-2013) average annual expenditures | Expenditures in 2013 | Ratio 2013 expenditures/long term average annual expenditures |
|----------------|---|-------------------------|--|
| Austria | 5,722 | 4,497 | 79 % |
| Belgium | 2,544 | 2,452 | 96% |
| Bulgaria | 403 | 532 | 132% |
| Czech Republic | 2,341 | 1,592 | 68% |
| Denmark | 2,407 | 1,967 | 82% |
| Germany | 20,721 | 21,497 | 104% |
| Estonia | 228 | 229 | 101% |
| Finland | 2,440 | 2,179 | 89 % |
| France | 21,791 | 20,261 | 93% |
| Greece | 1,912 | 1,164 | 61% |

| Member State | Long-term (1995-2013) average annual expenditures | Expenditures in 2013 | Ratio 2013 expenditures/long term average annual expenditures |
|-----------------|---|-------------------------|--|
| Croatia | 1,030 | 769 | 75% |
| Hungary | 1,589 | 1,201 | 76% |
| Ireland | 1,219 | 796 | 65% |
| Italy | 16,670 | 7,162 | 43% |
| Latvia | 274 | 293 | 107% |
| Lithuania | 481 | 435 | 90% |
| Luxembourg | 267 | 295 | 111% |
| Malta | 30 | 42 | 138% |
| The Netherlands | 7,148 | 6,925 | 97 % |
| Poland | 3,784 | 3,479 | 92% |
| Portugal | 2,031 | 532 | 26% |
| Romania | 3,431 | 4,120 | 120% |
| Slovakia | 718 | 651 | 91% |
| Slovenia | 628 | 280 | 45% |
| Spain | 9,811 | 5,123 | 52% |
| Sweden | 3,743 | 4,829 | 1 29 % |
| United Kingdom | 10,964 | 9,513 | 87% |
| EU27 | 124,329 | 102,814 | 80% |

Note: As no reliable data on Cypriot road infrastructure investments was available, Cyprus is not shown in this table.



Annex F Development of expenditures in the EU Member States

F.1 Introduction

In this annex we present the development of infrastructure expenditures in the period 1995-2013 in the individual countries. The development of total expenditures are presented in Section F.2 (investment indices) and Section F.3 (O&M expenditure indices), while the development of the share of these expenditures in GDP are shown in Sections F.4 and F.5, respectively.

To present the results in a clear way, we grouped them for six groups of countries:

- Scandinavian countries: Denmark, Finland, Sweden;
- North Western European countries: Austria, Belgium, Germany, France, Luxembourg, the Netherlands, UK;
- PIIGS: Portugal, Ireland, Italy, Greece, Spain;
- Baltic countries: Estonia, Latvia, Lithuania;
- Other countries that accessed the EU in 2004: Czech Republic, Hungary, Malta, Poland, Slovakia, Slovenia;
- Countries that recently accessed the EU: Bulgaria, Croatia and Romania.

The graphs presented in this Annex are meant to assess the developments over time of road expenditures in the individual EU countries. These graphs do not have the objective of comparing/benchmarking the various countries³⁸.

The data sources used to estimate the developments of road infrastructure expenditures over time as well as the main uncertainties associated to these data are discussed in Annex A.



³⁸ As discussed in Section 1.4 countries may differ in accounting principles applied and in the quality and availability of the data, which may hamper the comparison of expenditure data between countries.

F.2 Investment indices

The investment indices for the various EU countries are presented in Figure 53 to Figure 58. All indices are based on annual figures expressed in euros price level 2013 and PPP adjusted.



Figure 53 Investment indices for road infrastructure in the period 1995-2013 in Scandinavian countries

Figure 54 Investment indices for road infrastructure in the period 1995-2013 in North Western European countries





Figure 55 Investment indices for road infrastructure in the period 1995-2013 in PIIGS



Figure 56

Investment indices for road infrastructure in the period 1995-2013 in Baltic countries



Note: As the investments in Estonia are estimated based on the investments per kilometre road in Latvia, the development over time is almost the same in both countries (see Annex A.4).



Figure 57 Investment indices for road infrastructure in the period 1995-2013 in the other countries that accessed the EU in 2004



Figure 58 Investment indices for road infrastructure in the period 1995-2013 in the countries that recently accessed the EU



Note: As the investments over the period 1995-2005 in Bulgaria and Romania are estimated using the same growth trend, the development in these expenditures over that period shows the same pattern for both countries (see Annex A.4).



F.3 O&M expenditure indices

The O&M expenditure indices for the various EU countries are presented in Figure 59 to Figure 64. All indices are based on annual figures expressed in Euros price level 2013 and PPP adjusted.











Figure 61 O&M expenditure indices for road infrastructure in the period 1995-2013 in PIIGS



Note: As the same growth rate is used to estimate the O&M expenditures over the period 1995-2013 in Greece and Spain, the development in these expenditures over that period shows the same pattern for both countries (see Annex A.5).

Figure 62 O&M expenditure indices for road infrastructure in the period 1995-2013 in Baltic countries



Note: As the same growth rate is used to estimate the O&M expenditures over the period 1995-2001 in Estonia, Latvia and Lithuania, the development in these expenditures over that period shows the same pattern for the three countries (see Annex A.5)



Figure 63 O&M expenditure indices for road infrastructure in the period 1995-2013 in the other countries that accessed the EU in 2004



Figure 64 O&M expenditure indices for road infrastructure in the period 1995-2013 in the countries that recently accessed the EU



Note: As the O&M expenditures over the period 2006-2013 in Bulgaria and Romania are estimated using the same growth trend, the development in these expenditures over that period shows the same pattern for both countries (see Annex A.5). For Bulgaria it was not possible to provide significantly reliable O&M expenditure figures for the period 1995-2002 and hence the figures were kept constant.



F.4 Investments as share of GDP

The development of investments as share of GDP in the various EU countries is presented in Figure 65 to Figure 70.



Figure 65 Investments as share of GDP in the period 1995-2013 in Scandinavian countries

Figure 66 Investments as share of GDP in the period 1995-2013 in North Western European countries





Figure 67 Investments as share of GDP in the period 1995-2013 in PIIGS



Figure 68

Investments as share of GDP in the period 1995-2013 in Baltic countries









Figure 70 Investments as share of GDP in the period 1995-2013 in countries that recently accessed the EU





F.5 O&M expenditures as share of GDP

The development of O&M expenditures as share of GDP in the various EU countries is presented in Figure 71 to Figure 76.



Figure 71 O&M expenditures as share of GDP in the period 1995-2013 in Scandinavian countries

Figure 72 O&M expenditures as share of GDP in the period 1995-2013 in North Western European countries





Figure 73 O&M expenditures as share of GDP in the period 1995-2013 in PIIGS



O&M expenditures as share of GDP in the period 1995-2013 in Baltic countries Figure 74





Figure 75 O&M expenditures as share of GDP in the period 1995-2013 in other countries that accessed the EU in 2004



Figure 76 O&M expenditures as share of GDP in the period 1995-2013 in countries that recently accessed the EU





Annex G Allocation of infrastructure costs

In CE Delft and VU (2014), a literature review on the allocation of infrastructure costs of road transport was carried out. Five international studies estimating the infrastructure costs of road transport for a specific country (i.e. Switzerland, The Netherlands, Great Britain, Germany and Australia) and two studies providing a (qualitative) meta-analysis on this issue (i.e. (Fraunhofer-ISI; CE Delft, 2008) and (HLG, 1999) were reviewed. The main results of the review of these studies are summarised in Table 27.

Based on this literature review we conclude the following:

- Almost all studies distinguish between capacity and weight related enhancement costs. In general, about 90% of the enhancement costs are considered capacity related and 10% weight related (an exception is NTC (2005) which assumes that just 45% of the enhancement costs in Australia are weight related). In this study we use the same shares of capacity and weight related enhancement costs. As CE Delft (2008) and ITS (2001) we allocate the capacity related enhancement costs based on Passenger Car Equivalent (PCE) kilometres and the weight related enhancement costs based on axle load kilometres (4th power rule).
- Different approaches are used to allocate the *renewal costs* of road infrastructure. However, all studies distinguish between capacity related and weight related renewal costs. Based on the results from the various studies and detailed data for the Netherlands (not shown in Table 27, but presented in CE Delft (2008)) we assume that ca. 60% of the renewal costs are weight related and ca. 40% are capacity related. As for enhancement costs, the capacity related costs are allocated based on PCE-kilometres and the weight related costs on axle load kilometres (4th power rule).
- With respect to the allocation of maintenance costs, most studies distinguish between fixed and variable costs. Since the variable maintenance costs are - at least partly weight dependent - we allocate these costs based on axle load kilometres. With respect to the fixed maintenance costs almost all studies agree that these costs depend both on weight and capacity factors. However, different kind of approaches are used. In this study we apply the German approach (which was also used by CE Delft (2008) and CE Delft and VU (2014) for the Netherlands), implying that 50% of the costs are allocated based on PCE-km, 35% based on vehicle kilometres and 15% are allocated to heavy goods vehicles.
- Both vehicle kilometres and PCE-kilometres (or a combination of both) are used as cost drivers to allocate the *operational costs* of road transport. In this study we apply the German approach (30% of the operational costs are allocated based on vehicle-kilometres, 70% based on PCE-kilometres).



| Source | Country | Enhancement costs | Renewal costs | Maintenance costs | Operational costs |
|---|--------------------|--|---|--|--|
| BSF (2003) | Switzerland | 95% capacity related: 80% allocated based on PCE-km, 20% allocated based on vkm 5% weight related: fully allocated to HDVs (> 3.5t) based on standard axle load kilometres | 55% capacity related: 80% allocated based on PCE-km, 20% allocated based on vkm 45% weight related: fully allocated to HDVs (>3.5t) based on axle load vehicle kilometres | 100% allocated based on vkm | 100% allocated based on vkm |
| CE Delft (2008) | The Netherlands | 89% capacity related: allocated based on PCE-km 11% weight related: allocated based on standard axle load kilometres | Fixed costs: 35% allocated based on vkm, 50% on PCE-km and 15% are fully allocated to HGVs (>12t) Variable costs: mainly allocated based on standard axle load km | Fixed costs: 35% allocated based on vkm, 50% on PCE-km and 15% are fully allocated to HGVs (>12t) Variable costs: mainly allocated based on standard axle load km | Ca. 44% are allocated based on vkm, 14% based on standard axle load km and 41% based on PCE-km |
| ITS (2001) | UK | 85% capacity related: allocated based on PCE-km 15% weight related: allocated based on standard axle load km | Detailed approach Renewal costs are mainly allocated based on standard axle load kilometres | Detailed approach Variable costs are mainly allocated based on standard axle km Fixed costs are mainly allocated based on PCE-km and/or average gross vehicle weight km | Operational costs are mainly allocated based on PCE-km (a minor part is allocated based on average gross tonne km) |
| ProgTrans (2007) | Germany | Very detailed approach Main part of cost elements is allocated based on PCE-km Smaller part of costs is allocated fully to specific vehicle categories (mainly HGVs) | Very detailed approach According to IMPACT D2 about 50% are allocated based on PCE-km, 22% on vkm and 28% on standard axle load km | Fixed costs: 35% allocated based on vkm, 50% on PCE-km and 15% are fully allocated to HGVs (>12t) Variable costs: allocated based on PCE-km, axle load km and vkm. A part of the costs is allocated to specific vehicle categories | 30% allocated based on vkm 70% allocated based on PCE-km |
| NTC (2005) | Australia | 45% weight related: allocated based on standard axle load 55% not attributable: allocated based on vkm | 45% weight related: allocated based on standard axle load 55% not attributable: allocated based on vkm | Detailed approach Allocated based on axle load km, PCE-km and vkm | 100% allocated based on vkm |
| Fraunhofer- ISI and CE Delft (2008) | EU | 100% allocated based on PCE-km | 80% allocated based on PCE-km 10% allocated based on vkm 10% allocated based on standard axle load km | 100% based on standard axle load km | 100% allocated based on vkm |
| High level group (1999) | EU | | Costs should be allocated based on axle weight, gross vehicle weight and/or PCE-km | Cost should be allocated based on axle weights | Cost should be allocated based on vkm or PCE-km |

Table 27 Main results of literature review on allocation approaches applied in infrastructure costs studies

